

City of Hamilton

AUDIT, FINANCE AND ADMINISTRATION COMMITTEE REVISED

Meeting #: 21-002

Date: February 4, 2021

Time: 9:30 a.m.

Location: Due to the COVID-19 and the Closure of City

Hall

All electronic meetings can be viewed at:

City's Website:

https://www.hamilton.ca/council-committee/council-committee-meetings/meetings-and-agendas

City's YouTube Channel:

https://www.youtube.com/user/InsideCityofHa

milton or Cable 14

Angela McRae, Legislative Coordinator (905) 546-2424 ext. 5987

1. APPROVAL OF AGENDA

(Added Items, if applicable, will be noted with *)

- 2. DECLARATIONS OF INTEREST
- 3. APPROVAL OF MINUTES OF PREVIOUS MEETING
 - 3.1. January 14, 2021

4. COMMUNICATIONS

4.1. Correspondence from Cameron Kroetsch, respecting suggested changes to consolidated Procedural By-law 18-270

Recommendation: Be received and referred to consideration of Item 9.4 - Governance Review Sub-Committee Report 21-001.

*4.2. Correspondence from Joey Coleman, respecting a proposed bylaw banning letters critical of Council behaviour from being submitted as public correspondence

Recommendation: Be received and referred to consideration of Item 9.4 - Governance Review Sub-Committee Report 21-001.

- 5. DELEGATION REQUESTS
- 6. CONSENT ITEMS
- 7. PUBLIC HEARINGS / DELEGATIONS
 - 7.1. Sanaullah Chaudhry, respecting a water bill totaling \$2,193.34 (for October 18 Dec. 18th through Alectra-Bill) (Approved January 14, 2021)
 - 7.1.a. Staff Supporting Documentation Sanaullah Chaudhry
 - 7.2. Robert Ridley, on behalf of Olympia Gerl, respecting a Water Billing Charging Anomaly (Approved January 14, 2021)
 - 7.2.a. Staff Supporting Documentation Robert Ridley on behalf of Olympia Gerl
- 8. STAFF PRESENTATIONS
- 9. DISCUSSION ITEMS
 - 9.1. Citizen Committee Report Hamilton Status of Women Advisory Committee Donation of Remaining 2020 Budget Allocation
 - 9.2. 2020 City Enrichment Fund Update (GRA21001) (City Wide)
 - 9.3. Citizen Committee Report Lesbian, Gay, Bisexual, Transgender and Queer (LGBTQ) Advisory Committee All Advisory Committee Event
 - 9.4. Governance Review Sub-Committee Report 21-001 January 25, 2021
- 10. MOTIONS
- 11. NOTICES OF MOTION
- 12. GENERAL INFORMATION / OTHER BUSINESS
- 13. PRIVATE AND CONFIDENTIAL

13.1. Commercial Relationship between the City of Hamilton and 1389797 Ontario Inc. o/a Medical Centre Optical (FCS11020(a) / LS11001(a)) (City Wide)

Pursuant to Section 8.1, Sub-section (f) of the City's Procedural By-law 18-270, as amended, and Section 239(2), Sub-section (f) of the *Ontario Municipal Act, 2001*, as amended, as the subject matter pertains to advice that is subject to solicitor-client privilege, including communications necessary for that purpose.

14. ADJOURNMENT



AUDIT, FINANCE AND ADMINISTRATION COMMITTEE MINUTES 21-001

9:30 a.m. January 14, 2021 Council Chambers Hamilton City Hall

Present: Councillors B. Clark (Chair), C. Collins, L. Ferguson, B. Johnson, M.

Pearson, A. VanderBeek, and M. Wilson

Absent: Councillor J. Partridge – Personal

Also Present: Councillor J.P. Danko

THE FOLLOWING ITEMS WERE REFERRED TO COUNCIL FOR CONSIDERATION:

Development Charge (DC) Demolition Credit Extension Request, 651-677
 Upper James Street (Mountain Plaza Mall) (FCS17008(a)) (City Wide) (Item 9.1)

(Pearson/Collins)

- (a) That the City Development Charges (DC) demolition credits for the Mountain Plaza Mall (651-677 Upper James Street), be extended for an additional two years and funded by \$106,579.73 from the Water Reserve, \$212,263.84 from the Wastewater Reserve, and \$521,255.50 from the Unallocated Capital Reserve; with the following conditions:
 - (i) That the site plan be submitted within 9 months; and,
 - (ii) That no further extensions will be granted to this property;
- (b) That the subject matter respecting the "Delegation from Kevin Rachman, SmartCentres REIT, regarding an Extension to a Development Charges Credit", be identified as complete and removed from the Audit, Finance and Administration Committee Outstanding Business List.

Result: Main Motion As Amended CARRIED by a vote of 4 to 3, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

NO - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

NO - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

NO - Ward 1 Councillor Maureen Wilson

2. 2020 City of Hamilton External Audit Plan (FCS21002) (City Wide) (Item 9.2)

(Pearson/Collins)

That Appendix "A" to Report FCS21002, respecting the 2020 City of Hamilton External Audit Plan prepared by KPMG, be received.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

3. City of Hamilton / Ministry of Transportation Municipal Transit Enhanced Cleaning Transfer Payment Agreement (FCS21005) (City Wide) (Item 9.3)

(Johnson/Pearson)

- (a) That the Mayor and the City Clerk be authorized and directed to enter into a Transfer Payment Agreement attached as Appendix "A" to Report FCS21005 between the City of Hamilton and the Province of Ontario related to the funding commitment made by the Province of Ontario to the municipality regarding Municipal Transit Enhanced Cleaning funding in a form satisfactory to the City Solicitor;
- (b) That, upon being passed, a certified copy of the authorizing By-law and resolution, together with the signed Transfer Payment Agreement for the Municipal Transit Enhanced Cleaning be forwarded to the Ministry of Transportation.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

4. Protection of Privacy Policy (FCS21003) (City Wide) (Added Item 9.4)

(Pearson/Wilson)

- (a) That Report FCS21003, respecting the Protection of Privacy Policy, be received;
- (b) That the Protection of Privacy Policy, attached as Appendix "A" to Report FSC21003, be approved; and,
- (c) That the request for 1 FTE to support the administration of the privacy policy, be referred to 2021 budget deliberations for consideration.

Result: Motion CARRIED by a vote of 6 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

NOT PRESENT - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

5. Waiving of the 'False Alarm' Fee for Famee Furlane (Added Item 11.1)

(Johnson/Pearson)

WHEREAS, the Famee Furlane of Hamilton is the social and cultural centre for Italians from the FRIULI, the northeastern most region of Italy and a non-profit community-based organization with membership comprised mostly of people of FRIULAN origin;

WHEREAS, Famee Furlane of Hamilton's revenues have been significantly impacted by the Coronavirus pandemic;

WHEREAS, the organization is comprised of volunteers;

WHEREAS, on November 30, 2020 Famee Furlane of Hamilton had a new volunteer testing their fire alarm and he was unaware that the Fire Department needed to be contacted first: and

WHEREAS, Famee Furlane of Hamilton was therefore issued a "false alarm" fee as per the City of Hamilton's User Fee bylaw;

THEREFORE. BE IT RESOLVED:

That staff be directed to waive the "false alarm" fee of \$753.50 for Famee Furlane's false alarm on November 30, 2020, as compassionate relief, and that staff communicate to Famee Furlane of Hamilton that this is a one-time waiving of the 2021 fee only.

Result: Motion CARRIED by a vote of 6 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

NOT PRESENT - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

FOR INFORMATION:

(a) CHANGES TO THE AGENDA (Item 1)

The Committee Clerk advised of the following changes to the agenda:

5. DELEGATION REQUESTS

5.3 Robert Ridley, on behalf of Olympia Gerl, respecting a Water Billing Charging Anomaly (For a future meeting)

9. DISCUSSION ITEMS

9.4 Protection of Privacy Policy (FCS21003) (City Wide)

11. NOTICES OF MOTION

11.1 Waiving of the 'False Alarm' Fee for Famee Furlane

(Pearson/Johnson)

That the agenda for the January 14, 2021 Audit, Finance and Administration Committee meeting be approved, as amended.

Result: Motion CARRIED by a vote of 6 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

NOT PRESENT - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(b) DECLARATIONS OF INTEREST (Item 2)

There were no declarations of interest.

(c) APPROVAL OF MINUTES OF PREVIOUS MEETING (Item 3)

(i) December 10, 2020 (Item 3.1)

(Ferguson/Pearson)

That the Minutes of the December 10, 2020 meeting of the Audit, Finance and Administration Committee be approved, as presented.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(d) DELEGATION REQUESTS (Item 5)

(i) Sanaullah Chaudhry, respecting a water bill totaling \$2,193.34 (for October 18 - Dec. 18th through Alectra-Bill) (For a future meeting) (Item 5.1)

(Wilson/Ferguson)

That the Delegation Request from Sanaullah Chaudhry, respecting a water bill totaling \$2,193.34 (for October 18 - Dec. 18th through Alectra-Bill), be approved for a future meeting.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(ii) Kevin Rachman, SmartCentres REIT, respecting staff Report FCS17008(a) (Item 9.1 on today's agenda) on a request made, on behalf of SmartCentres (For today's meeting) (Item 5.2)

(Johnson/Collins)

That the Delegation Request from Kevin Rachman, SmartCentres REIT, respecting Item 9.1 - Development Charge (DC) Demolition Credit Extension Request, 651-677 Upper James Street (Mountain Plaza Mall) (FCS17008(a)), be approved for a today's meeting.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(iii) Robert Ridley, on behalf of Olympia Gerl respecting a Water Billing Charging Anomaly (For a future meeting) (Added Item 5.3)

(Wilson/Ferguson)

That the Delegation Request from Robert Ridley, on behalf of Olympia Gerl respecting a Water Billing Charging Anomaly, be approved for a future meeting.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(e) PUBLIC HEARINGS / WRITTEN DELEGATIONS / VIRTUAL DELEGATIONS (Item 7)

(i) Kevin Rachman, SmartCentres REIT, respecting staff Report FCS17008(a) (Item 9.1 on today's agenda) on a request made, on behalf of SmartCentres (Added Item 7.1)

Kevin Rachman, SmartCentres REIT, addressed the Committee respecting Item 9.1 - Development Charge (DC) Demolition Credit Extension Request, 651-677 Upper James Street (Mountain Plaza Mall) (FCS17008(a)).

(Johnson/Pearson)

That the delegation from Kevin Rachman, SmartCentres REIT, respecting Item 9.1 - Development Charge (DC) Demolition Credit Extension Request, 651-677 Upper James Street (Mountain Plaza Mall) (FCS17008(a)), be received.

Result: Motion CARRIED by a vote of 7 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(f) DISCUSSION ITEMS (Item 9)

(i) Development Charge (DC) Demolition Credit Extension Request, 651-677 Upper James Street (Mountain Plaza Mall) (FCS17008(a)) (City Wide) (Item 9.1)

(Pearson/Collins)

That sub-section (a) be amended as follows:

- (a) That the City Development Charges (DC) demolition credits for the Mountain Plaza Mall (651-677 Upper James Street) for an additional five years, to January 12, 2025, be denied; be extended for an additional two years and funded by \$106,579.73 from the Water Reserve, \$212,263.84 from the Wastewater Reserve, and \$521,255.50 from the Unallocated Capital Reserve; with the following conditions:
 - (i) That the site plan be submitted within 9 months; and,
 - (ii) That no further extensions will be granted to this property;

Result: Amendment CARRIED by a vote of 4 to 3, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

NO - Ward 13 Councillor Arlene VanderBeek

YES - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

NO - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

NO - Ward 1 Councillor Maureen Wilson

For further disposition of this matter, refer to Item 1.

(g) NOTICES OF MOTION (Item 11)

(i) Waiving of the 'False Alarm' Fee for Famee Furlane (Added item 11.1)

(Johnson/Pearson)

That the Rules of Order be waived to allow for the introduction of a Motion respecting Waiving of the 'False Alarm' Fee for Famee Furlane.

Result: Motion CARRIED by a vote of 6 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

NOT PRESENT - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

For further disposition of this matter, refer to Item 5.

(h) PRIVATE AND CONFIDENTIAL (Item 13)

(i) December 10, 2020 – Closed Minutes (Item 13.1)

(Collins/VanderBeek)

- (a) That the Closed Session Minutes of the December 10, 2020 Audit, Finance and Administration Committee meeting, be approved as presented; and,
- (b) That the Closed Session Minutes of the December 10, 2020 Audit, Finance and Administration Committee meeting, remain confidential

Result: Motion CARRIED by a vote of 6 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

NOT PRESENT - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

YES - Ward 9 Councillor Brad Clark

YES - Ward 5 Councillor Chad Collins

YES - Ward 1 Councillor Maureen Wilson

(i) ADJOURNMENT (Item 14)

(Pearson/Johnson)

That, there being no further business, the Audit, Finance and Administration Committee, be adjourned at 11:24 a.m.

Result: Motion CARRIED by a vote of 6 to 0, as follows:

NOT PRESENT - Ward 15 Councillor Judi Partridge

YES - Ward 13 Councillor Arlene VanderBeek

NOT PRESENT - Ward 12 Councillor Lloyd Ferguson

YES - Ward 11 Councillor Brenda Johnson

YES - Ward 10 Councillor Maria Pearson

Audit, Finance and Administration Minutes 21-001

Page 13 of 261 January 14, 2021 Page 9 of 9

YES - Ward 9 Councillor Brad Clark YES - Ward 5 Councillor Chad Collins YES - Ward 1 Councillor Maureen Wilson

Respectfully submitted,

Councillor Clark, Chair Audit, Finance and Administration Committee

Angela McRae Legislative Coordinator Office of the City Clerk

Suggested changes to the City's Procedural By-law

Audit, Finance and Administration Committee

February 4, 2021

Cameron Kroetsch



The suggestions outlined in my 20 page submission are the culmination of work I have been doing with the City of Hamilton to suggest improvements to its bylaws since November 2017.



Why does this matter?

These comments are being made in the spirit of continued collaboration and a genuine desire to see both short and long term improvements.

2

The entire bylaw should be reworded using the principles of **Clear Writing** Standards and using language that is accessible to as many Hamiltonians as possible.



Why does this matter?

The easier the document is to read, the easier it will be to interpret and use by **everyone**, including City Council, staff, and the general public.

3

The bylaw should be reorganized, shortened, and similar things should be grouped together. In general, the bylaw should spell out anything that's unclear, missing, or requires another source.



Why does this matter?

Members of Council will be able to rely on the bylaw as a complete document and it will be easier to navigate and find specific wording.



Members of City Council should give themselves more time to review documents in between meetings and the meeting schedule should be changed as soon as possible to prioritize this.



Why does this matter?

This will give members of Council more time to review reports, documents, and correspondence. It will also give the public more time to engage.

Reorganizing, shortening, and increasing clarity

Has the bylaw gotten longer?

45 pgs 2010

52 pgs 13k wds 14k wds 2014

67 pgs **20k wds** 2021

+50% increase in # of pgs / wds

Was the bylaw made clearer?



-1 full letter grade in readability

Readability of current bylaw

- Flesch-Kincaid / Gunning Fog 12.0
 - Good scores are less than 10.0
- Flesch Reading Ease 40.1
 - Good scores are over 60.0
- Sentences with more than 30 syllables 42%
 - Good scores are less than 6%

Readability of current bylaw

- Sentences with more than 20 syllables 56%
 - Good scores are less than 12%
- Words with more than 12 letters 0%
 - Good scores are less than 3%
- Words with more than 4 syllables 2%
 - Good scores are less than 3%



Where is your time spent?

City Council
General Issues
Board of Health

All members of Council

AF & A - 8/16

E & CS - 6/16

Planning - 8/16

Public Works - 11/16

38% - 69%

Where is your time spent?

3 mtgs

6.25%

1 member

5 mtgs

81.25%

13 members

6 mtgs

18.75%

3 members

81.25% of Council already attends 5

What am I suggesting?

City Council

General Issues

Board of Health

All members of Council

AF & A - 16/16

E & CS - 16/16

Planning - 16/16

Public Works - 16/16

All members of Council

What am I suggesting?

6 mtgs

100.00%

~20% of Council already attends 6

What are the benefits?

All members of council are at all meetings

More time to review things in between

No repetition

At least 1 week

What are the benefits? All members of council are at all meetings

+ Inclusion + Awareness + Representation To the members of the City of Hamilton's Audit, Finance and Administration Committee for consideration at their February 4, 2021 meeting

Suggested changes to consolidated Procedural By-law 18-270 Cameron Kroetsch

Preamble

The comments shared in this document are the result of work I have been doing since late 2017 with both City of Hamilton staff and the Governance Review Sub-Committee. Some of these comments were shared with then Acting City Clerk Janet Pilon in advance of the Council term changes made in 2018 (before City Council approved my suggestion to go to yearly reviews of the bylaw).

General comments

- The bylaw should be reorganized with an emphasis on grouping things together by priority. What terms are universal? What is necessary? Many of the subtitles are confusing (i.e. "General", "Application"). It would be useful, for instance, to have universal sections on "Motions", one on "Points of Order", etc. This would help to streamline the bylaw and likely reduce the total number of pages.
- The bylaw should spell out basic procedures very clearly and directly from Bourinot's Rules of Order (BRO) rather than suggesting that the absence of any of those rules should cause the reader to have to consult BRO. The bylaw should be a complete document.
- An orientation and training outline should be made part of the bylaw to ensure that there is mandatory compliance with training and that any training is sufficient enough for councillors to properly execute their duties.
- More direct references to BRO (where something has been paraphrased) and the *Municipal Act* (MA), when it is being relied on, should be made throughout the bylaw so that it's easier for councillors to find the relevant legislation when they need to.
- In general, and where legally permissible, this bylaw should be written in more
 accessible language and with the acknowledgement that it is going to be used by
 non-experts including the general public. Those non-experts need a document that is
 impeccably clear, using agreed upon Clear Writing Standards.

- A process for reporting mechanisms and report tracking should be included here. It should be obvious to anyone who reads this document how the City's processes work with respect to reports, deadlines, and follow up.
- A completely revised schedule of meetings for Council should be considered such that
 everyone, including Councillors, has more time to review documents. This would be a
 considerable overhaul of the current system that could be accomplished most easily
 either in December or during the summer months of any calendar year (see comments
 in section 3.2).
- Strict deadlines for staff submissions to Council should be included with a provision for "emergency only" items and a defined set of parameters for what an emergency is. If staff do not meet those deadlines then the item should be forced onto the next agenda, giving everyone more time. If staff needs more time then everyone needs more time.
- It would be good to set a time for delegations and for specific time slots for delegates during all meetings. It is far too often the case that delegates wait around for hours to delegate. If they are assigned a time slot at the beginning of a meeting, along with allotted times for other initial activities, this will aid in encouraging more participation from the public.

1.1 Definitions

While these may be definitions and not technically in the "body of the bylaw" it would be useful to include as many references to legislation as possible in the definitions section. This document may be used by the public in any number of ways and may not be used sequentially. More references are usually better than fewer and this is no exception.

- In general, the definitions are missing the kind of detail that exists in the <u>City of Ottawa's 2007 Bill 130</u>, which helps to classify the different types of "committees". After reviewing the City's website, it appears that these are the categories of "committee" that exist in the City of Hamilton. They should be defined in this bylaw and are as follows.
 - Advisory Committee (e.g. Seniors Advisory Committee)
 - Advocacy Group (e.g. Federation of Canadian Municipalities)
 - Authority (e.g. Niagara Peninsula Conservation Authority)
 - Board (e.g. Alectra Board of Directors)
 - Council (e.g. City Council)
 - Commission (e.g. Niagara Escarpment Commission)
 - Liaison Group (e.g. Development Industry Liaison Group)
 - Shareholder (e.g. Hamilton Renewable Power Inc. Shareholder)
 - Sole Voting Member (e.g. Hamilton Farmers' Market, Sole Voting Member)
 - Standing Committee (e.g. Audit, Finance and Administration Committee)

- Sub-Committee (e.g. Governance Review Sub-Committee)
- Tribunal (e.g. Hamilton Licensing Tribunal)
- Trust (e.g. Knowles Bequest Trust)
- "Approve" This seems problematic. There are only decisions in BRO, not "approvals".
 A report can be "accepted" but to vote in favour or against is not the same as "approve".
 To approve something is an action, but the only thing that these bodies do with actions is make decisions.
- "Bill" It would be useful to make the exact reference to the legislation here (i.e. where it says that a proposed bylaw is a called a bill).
- "Closed Meeting" Be specific about what Act is being referred to and what section of the Act is implied here so that it can be quickly referenced.
- "Consent Items" This is very tricky and should be removed. There is no such thing in BRO as a "consent item". There is a "consent agenda", in some general use, but even that is not used very widely. This idea supposes that there are items that will be met with the automatic approval of Council. In general, if staff are already empowered to do something then it doesn't need to go on the agenda at all (except perhaps as an item of Correspondence or as a staff Information Report). If, however, there is some Council approval required, that approval should be outlined clearly and it should be treated like any other motion (i.e. not a Consent Item). Attempting to classify motions this way, and only in the positive, is confusing to Councillors and the public and establishes "consent" as an awkward democratic practice.
- "Committee of the Whole" It would be best to remove this practice entirely. It does not appear to have a mandatory legal use and is here primarily out of "tradition". This may have made more sense when there was a different structure, especially when there were other bodies like a Board of Control for instance. While it is understood that Council considers reports from Standing Committees in this manner, it is not technically necessary for them to do so. Council can consider them without doing this. It no longer makes sense to consider that City Council must be at a certain legislative distance from Committees or that it has to do some parts of its work as the "Committee of the Whole" away from its role as a municipal Council. It is clear to all concerned, and the public, that Council ratifies certain things. Comments in section 3.2 suggest a way to get around this.
- "Deputy Mayor" If the Mayor has or derives certain powers the source of those powers should be outlined here.
 - (a) What does it mean to "assist" the Mayor? If it means, as is stated, "that if the Mayor requires the Deputy Mayor's assistance they would call on them to attend

a meeting in their absence when appointments conflict and the rest of the definition speaks to their responsibilities" then it should be stated more clearly. At this point, it's not clear that the Deputy Mayor would assist the Mayor at their request.

(b)

(i) It is not clear what powers the Deputy Mayor can exercise and while "all" is useful it's not instructive. What are the terms for "absent" from the city? Does this mean the city limits or does this mean a period of time? Is there notification that must be given? In what form? How does the Deputy Mayor know this?

(ii)

- (2) What does this mean and when does it apply?
- "Designate" This definition could use more detail around how an office holder (like the City Clerk, or the Mayor) "designates" someone with their authority. Is this done in writing? Where is this recorded? How is it known to those who are accepting the authority of the designate?
- "Emergency" It is clear, because of the use of the word "emergency" throughout parts of the bylaw, that the City should define this, generally speaking. What constitutes an emergency? Be specific. At this point, especially in relation to section 3.5, it appears to be at the whim of the Mayor. It would make more sense for a quorum of members of Council to decide on whether an emergency meeting is called since all of them must be in attendance. This seems achievable through email.
- "Ex-officio" Wording is not succinct enough and the first sentence is very confusing. This definition is trying to do too many things. This is not where the reference to the Mayor being ex officio should be made, that should be made in a section about the Mayor which should stand alone and could include information about the Mayor's powers, etc. Also, the Latin term "ex officio" does not require a hyphen. Finally, to have someone be both "ex officio" and have full voting privileges creates a general sense of confusion. If this applies just to the Mayor then say so in a separate section, if all ex officio roles have voting privileges then they are no different than any other member of a committee and this should be made clearer. Also, all of the Standing Committees should be properly listed under that definition.
- "Motion" This has become confusing. Is there a definition in BRO that is slightly more straightforward? Is it a "proposal" or is it related to a "decision"? Motions are usually just decisions. Does it make sense to decide that a motion is only something on a formal

agenda? Council can put forward motions that are not on the agenda. More clarity would help with this and it is important to define it properly as it is foundational to the way that all meetings work.

- "Notice of Motion" It is very important to specify under what circumstances a Notice of Motion is required. If all decisions of Council technically happen as a "motion", of some sort or other, then this is not clear enough. It would presume that every decision, unless the rules were suspended, must have notice provided at least 1 meeting in advance. If this is merely optional or applies to specific motions, that should be made clear (cf. 3.12 [8]).
- "Point of Order" Where does the phrase "practised conduct" come from and what does it mean? Make a reference to this so it is clear to the reader. If it does not have an origin, consider rewording this.
- "Point of Privilege" Please provide clear and specific examples that would relate to this body. There are few, and this may help some Councillors better understand this issue.
 What does the phrase "that ensure Council's or a Committee's ability to function freely" mean? Does the word "freely" apply physically, in a purely political manner, or both?
- "Receive" A suggestion for this which would be more straightforward "means to
 accept information presented as part of the agenda in the official record" (or something
 similar to this). The word "receipt" makes this confusing as it is closely associated with
 the word "receive" (i.e. they have the same root meaning).
- "Recess" Members are now taking breaks for meals. This should be changed to include more specific language around breaks at meetings, when they normally take place, time limits, etc.
- "Recorded Vote" Make a reference to the section of the bylaw where this is otherwise mentioned.
- "Question" It doesn't appear that this is correct. It would appear that the word "question" is rightly the vote "in favour" or "opposed" to a motion only in the context of "putting the question" (see BRO sections 13 and 42), and nothing else. This would mean reviewing the bylaw to correct this error. Strictly speaking, City Council does not follow the "Parliamentary Rules" section of Stanford 1995 ("Motions" under paragraph 8 versus "Motions" under paragraph 36) entirely. It would be useful to make this clearer and more understandable. In the end, and for the purposes of City Council, "putting the question" is not the same as a "question" (it seems that Stanford may have misused this word). See also section 17 on "Questions", which further complicates the matter. The matter may be most easily resolved by looking at the origins of municipal rules, set out separately from rules for Canada's Parliament, as outlined in the *Procedure of Public*

Meetings by J. G. Bourinot (1894) or his Rules of Order (1918). While it is understandable to use the Stanford edition from 1995, it doesn't likely go into great enough detail. It would likely be better to rely on the manner in which the word "question" is used in the section in Stanford that talks about "general assemblies" (rather than the section on Canada's Parliament).

- "Quorum" If there is one standard rule that applies to all City of Hamilton meetings then this would be a good place to enshrine that rule. If it's 50% + 1, for instance, then explaining that here would be helpful. What happens in the case of an odd number, like 7 where the majority is 50%? A common understanding that staff has invoked at some meetings is 3.5 = 3; +1 = 4 is quorum. But, there have also been staff who have suggested that 3.5 = 4; +1 = 5 is quorum (in the same situation). It would be good if this was written down. Also, it would be better to cite sections 3.7 and 5.4 on Quorum and/or define all of the major points here. The sections on Quorum are confusing throughout this document and could easily have one set of standards applied and outlined here so as not to confuse things. If another section is needed it should be universal and not repeated separately for individual kinds of bodies (i.e. Council, Standing Committee, etc.).
- "Selection Committee" In this definition "agencies" are listed but they are not included in other definitions like, for instance, "Meeting" (but there are others). Is this intentional?
- "Time Sensitive" This phrase is only used once in the bylaw. The definition is ambiguous. It would be better to be specific and just outline what is meant in this one instance rather than have it listed as a definition (i.e. if a matter is affected by an impending deadline date or pertains to a specific time period).

2.2

• "as far as is reasonably practicable" - Deputy Clerk Pilon's comment from the previous review was, "There may be instances when the use of the BRO conflicts with another section of the bylaw, therefore it would not be reasonably practicable to follow the BRO in all respects when the PBL is silent on a matter". This phrase should be stated more clearly. It is too vague. Is there a reference in legislation? In general, it would be better if the bylaw was a complete document and did not require readers to access other documents for where it may be "silent on a matter".

3.1 First Council Meeting

• (2) (b) Specify the Act every time it is mentioned. Those who use this document may not be reading the entire document and might only be looking at a specific clause.

• (2) (c) It seems to me that this should not be done at the first meeting of Council. Council should have the opportunity to discuss this when it is formally a sitting Council. At the time this is discussed it is a "Council elect" and there are not appropriate ways for brand new Councillors to be able to enact themselves at that moment. It would be more appropriate to do this at the first meeting of the new year. It is a serious and important task that should be administered by staff and not left for an unbalanced field of power of "more senior" and "more junior" Councillors-elect to make decisions about. This can wait until they are sworn in. A good practice would actually be for all Councillors to sit in on all meetings during their first month in office and then, from there, to make decisions about their placement in the new calendar year.

3.2 Regular Council Meeting Times

- "Regular Meeting Times" should include customary times for meetings to end. It may also be customary to approve a meeting be extended, but a norm should be set here.
- In general, there is not enough time between regular meetings of Standing Committees and City Council sufficient for all concerned to adequately review the materials. It is often the case, at least once but sometimes more than once, that meeting agendas are sparse or a meeting is cancelled. This points to a system of inefficiencies that could be corrected. There are several ways to solve this problem using a variety of scheduling arrangements. The suggestion below would seem to require the least amount of disruption to the current methods and naming conventions. The goal of any system of changes would be to increase the amount of time that staff have to prepare and submit reports and that Council has to read and receive reports.

There are presently 6 Standing Committees, including the General Issues Committee (GIC) and the Board of Health. Since GIC consists of all members of Council it seems more appropriate to simply ensure that all Councillors attend all Standing Committee meetings (see analysis of this below the charts).

Caveat - this is one possible solution and not meant to be perfect. It's meant to demonstrate that, with a few tweaks, an alternative system exists that would allow both Councillors and the general public more time to absorb, read about, and participate in decision making.

In order to illustrate one possible solution, though there are many, let's use the month of October 2020 as an example.

Apart from any special budget GIC meetings, Council and GIC met a combined number of 4 times in October 2020, or about once a week. If GIC was eliminated in favour of **all members of Council attending all Standing Committee meetings** and Council meeting once every 2 weeks, this would help to streamline things.

Issues that have normally fallen outside of the 4 traditional Standing Committees or the Board of Health would be referred to the **Administration** portion of the Audit, Finance, and Administration Committee.

Once this system was up and running for a while, it would essentially work to provide everyone involved, including the public, **at least 1 full week** to read and receive materials in advance. It would take some practice and would change the flow of everyone's duties but because of the break periods that currently exist in the summer months and in December, this could be implemented without complete disruption (as long as ample training was provided in advance).

All reports and other City materials for Council meetings would be due 12 days in advance.

An example of how this would work in a sample month -

- The month starts on Monday the 1st and has only 30 days
- Wednesday the 10th is the first Council meeting (Council 1) and Wednesday the 24th is the second Council meeting (Council 2) and carries on in a biweekly cycle
- Audit, Finance and Administration and Emergency and Community Services
 would report to Council 1; reports and other City materials would be due on a
 Friday 12 days prior (only 2 Standing Committees report to Council 1 because of
 the extra material for Audit, Finance and Administration that would be absorbed
 by the former now dissolved General Issues Committee)
- All of the Standing Committees that report to Council 1 would have to meet a minimum of 14 days before Council 1 meets.
- Board of Health; Planning; and Public Works would report to Council 2; reports and other City materials would be due on a Friday 12 days prior
- All of the Standing Committees that report to Council 2 would have to meet a minimum of 14 days before Council 2 meets

Below is an example of how this could work over a 2 month period.

MONTH 1

Mon	Tue	Wed	Thu	Fri	Sat	Sun
1	2	3	4	5	6	7
PL C2		PW C2				
8	9	10	11	12	13	14
вон с2		C1		C2 due		
15	16	17	18	19	20	21
AFA C1		ECS C1				
22	23	24	25	26	27	28
		C2		C1 due		
29	30					
PL C2						

MONTH 2

Mon	Tue	Wed	Thu	Fri	Sat	Sun
		1	2	3	4	5
		PW C2				
6	7	8	9	10	11	12
вон с2		C1		C2 due		
13	14	15	16	17	18	19
AFA C1		ECS C1				
20	21	22	23	24	25	26
		C2		C1 due		
27	28	29	30	31		
PL C2		PW C2				

Having materials due on a Friday means that if there is more time needed for an exceptional situation, the weekend is there as a buffer; as long as the Office of the City Clerk has the materials in before Monday they would have the entire next 2 days to prepare the agenda to be distributed before Wednesday morning.

This is the current impact on City Council (meetings = Standing Committees, Board of Health, City Council) -

3 meetings	6.25%	1 member*
4 meetings	0.00%	0 members
5 meetings	81.25%	13 members
6 meetings	18.75%	3 members**

^{*}Mayor Fred Eisenberger is only a member of City Council, GIC, and Board of Health **Councillors Chad Collins, Lloyd Ferguson, and Maria Pearson

This would be the future impact on City Council with the changes suggested -

6 meetings 100.00% 16 members

3.3 Location of Meetings

• After this section, as a new section 3.4, there should be detailed instructions for the procedures associated with online meetings.

3.6 Meetings of Council for Emergencies

• (5) There is an extra leading space at the beginning of this clause.

3.8 Remedy for Lack of Quorum

• Cite the exact place in the *MCIA* where this originates (i.e. include a citation at the beginning or the end of the clause).

3.9 Delegations

• This statement does not appear to be correct. It should be modified to reflect the practices of Council or to mention that the rules must be suspended by a 2/3 majority in order to allow it (i.e. 3.14 [12]).

3.10 Communication Items

- (1) There are a few issues here. The first is with the word "presented". Who is presenting them? It's not clear that this makes sense. There's also no clear relationship established between the person sending it in and the body receiving it. A paragraph explaining what a "Communication Item" is, with it also being added to the definitions, would be helpful here. If "Correspondence" is the same as a "Communication Item" then this is likely where the confusion lies. Also, when it says "consideration" here it's not clear if this kind of consideration is the subject of a "decision/motion" or some other action.
- (2) The word "of" is likely missing in "Wednesday of the week prior".
- (5) It is not clear what is being "debated" specifically here. The practice is understood, but it could be made much clearer. It also seems repetitive. It appears that the phrase "recommended disposition" is not defined and that, in the absence of understanding the process, it is not clear what is happening here. It is also not clear who makes these recommendations. Also, it says "Council communications" here. Does it mean "Communication Items"? Something including these points would be better, "The Office of the City Clerk will make recommendations with respect to motions for Council to consider for each individual Communication Item and they will be listed with that item. Communication Items can either be moved to be received individually or as a group or not at all (i.e. moved not to be received). If a member wishes to move the receipt of a Communication Item individually, they may do so by referring to where it is listed and they may speak to the reasons why they have moved to do so. A motion to refer a Communication Item to another body is also in order. Debate on any motion with respect to a Communication Item must be limited either to reasons to receive or not receive a Communication Item or to reasons with respect to the whether or not an item should be referred to another body".
- (6) It is not clear that this is necessary as the rights around Communication Items do not change based on who submits them.
- (8) This appears to be inappropriate. City Council should make these decisions. Many Councillors have stated this preference over the years and it should be considered.
- (9) This appears to be inappropriate. The notion that any conduct of a member of City
 Council must rise to the level of a complaint accompanied by a \$100 fee would literally
 act as a filter to keep criticism away from City Council and would set an unnatural
 precedent or standard for criticism.
- (10) This clause refers to itself but is likely meant to refer to clauses (8) and (9).

3.11 Order of Business

- "unless changes by Council in the course of the meeting" this should be removed. It
 seems inappropriate and inconsiderate of those in the public who, for many reasons,
 are relying on the order of the agenda in order to tune in to something or in order to
 appear as a delegate. It is not reasonable for Council to change the order without
 advance notice.
- (i) This practice should be eliminated.

3.12 Rules of Debate

- There is nothing in this section, though there should be, that addresses personal attacks, insults, or raising unsubstantiated claims. It would be a good idea to include something that states that a procedure will be followed to ensure equity and freedom from discrimination or prejudice on specific grounds (i.e. the *Ontario Human Rights Code*).
- (6) This is good but it should also include a statement that says that it is only the member's own time that is being tracked, not the length of time of the response from a delegate. This has been interpreted broadly in different ways and it can serve to interrupt the flow if a delegate takes too much time in answering a question. In essence, the member's time clock should be based on their own comments and questions and nothing else.
- (8) It has never been clear that "leave" is required as part of this bylaw. This word should be removed unless it has an important legal precedence (which should be cited). Of the items on this list, this might be a better way to separate them for accuracy.

"Procedural Motions" (or motions that direct the course of the meeting but do not themselves bear on a decision about a specific matter) are -

- a motion to amend or make an amendment to a motion
- (c) a motion to adjourn
- (d) a motion to call the question
- (f) a motion to recess
- (g) a motion to refer

- (h) a motion to defer
- (j) a motion to go into a closed meeting
- (k) a motion to suspend the rules of procedure

"Motions" (decisions that are real or have a material impact on a matter before Council and are not merely procedural in effect); all of these motions should appear on the agenda and should be submitted on the agenda either in advance or at the start of the meeting when the agenda is being approved (though this should only be in the case of an emergency)

- all motions other than those listed above including motions to "ratify" the decisions of other bodies (which are not outlined in the bylaw except at 5.11 and in the definition for "Confirming By-law")
- (I) a motion to lift A motion to lift something does not procedurally change the
 fact that a decision is being made with respect to the matter "being lifted" for
 instance, if Council is approving a report of the Planning Committee, it tends to
 do so based on individual items in the report; there is no need, then, to lift an
 item out, just to make a specific decision to ratify
- (8) (e) The Committee of the Whole is unnecessary as is this motion (see comments under section 1.1).
- (8) (j) Cite the MA.
- (8) (I) The idea of a motion to lift should be eliminated from the bylaw unless it is going to be defined and contextualized. While it may be useful in creating expediency, it is not necessary to have a motion to lift something. Motions can be made about individual things without the need to lift them from larger things. Another way of dividing these things so that it's clear is to group motions about decisions separate from "procedural motions" (though all motions are related to decisions).

3.13 Voting Procedures

• (1) This is not accurate in terms of proper procedure. The person who moves the motion has the ultimate right to withdraw that motion if they wish. Council cannot prevent that without reintroducing the motion by having another one of its members else move and then second that motion once it has been withdrawn.

- (2) This appears to be related to a Declaration of Interest or, perhaps, a section that should be rightly called Statutory Conflicts. An example should be provided.
- (5) This does not clarify where the vote will be recorded in these instances (though the previous section clarifies this).
- (7) It should be made clear under "Point of Privilege" that this is relevant to that category. It is not clear if this is accurate. If it is, this would be a prudent and useful example. It would be best to list all acceptable points of privilege so that there is no confusion about what this term means. It is also not clear that a Point of Privilege can be voted on in this manner. If someone declares a legitimate Point of Privilege then the Chair would rule on that subject to a Challenge of the Chair that would be voted upon by a simple majority.
- (9) This should be reworded entirely. It is not clear that the only interest that should be declared is "pecuniary", this is also worded in a way that makes it illegible. If this refers to the MA, please refer to the specific section. What happens when someone is simply not present? How is this a "recusal"? These things are not clear.
- (10) Are recorded votes otherwise not tabulated? Do we have no record of who voted on something in the absence of an electronic system? If so, that should be stated clearly.
- (14) This seems undemocratic. While the discussion on such a matter can happen in a
 closed meeting under certain circumstances (cite the MA here), it does not make sense
 for the vote on that matter to happen in private. It is reasonable to simply come out of a
 closed meeting and vote on matters in public without revealing those matters in their
 entirety. Clear titles about those matters are sufficient in such instances.

3.14 Motions Procedures (should this be "Motion Procedures"?)

- (1) (a) It is still not clear what the required process is. It seems to be that someone can put a motion on the agenda by announcing it in the previous meeting or simply by sending it in the day before. If there is no requirement for (i) then it should be stated. It's unclear, for instance, when a Notice of Motion should be happening. There should also be an emergency provision here to make it clear that motions for a particular meeting have to be submitted well in advance unless it's an emergency. Again, the public and other members of Council deserve to have time to consider these items well in advance.
- (b) Does this mean, "withdrawn" by the mover and seconder? What does "not proceeded with" mean? This does not appear to be an official action.

- (c) Similar to (b), there is new language here about something being "dropped". This does not appear to be an official action.
- (2) Define "emergency", if it is the case that Council decides on what is or is not an emergency, please make this point clearly.
- (3) See earlier comments about this. The motion does not come under the carriage of Council until it is carried (or if the motion fails to be carried by Council). Once it is carried it then becomes a decision of that body. Until that time, it is the mover and the seconder who have placed the item "on the table" and the mover and seconder who can withdraw it from the table. If this rule came from BRO, please specify where.
- (4) The wording in this section appears to be incorrect. There are amendments referring to questions. They are not questions, they are motions. Look at the definition of the word "Question" in section 1.1, as follows, for more information about this -

"Question" means an issue before a meeting on which a decision has to be made. **A question cannot be** debated, **amended** or voted on until it has been proposed as a motion" (for more clarification see my comments on "Questions" in the definitions section).

- (5) This is not consistent with the definition. Here it says that a "date" must be specified but in the definitions section it says that it can be "after an event". Choose one definition and repeat it in both places.
- (6) This seems arbitrary and seems to limit the privileges of speakers, which is not permissible. So called "Dilatory Motions" (see Stanford section 11, which is under the parliamentary section), seem to provide some guidance but do not go as far as the bylaw does. It might be useful to look at Stanford sections 11, 42 (d), and 42 (e) for more clarity and to dial back what has been suggested in (6) (d) and (6) (e). A motion to refer should be debatable and a motion to refer should be subject to an amendment. Motions to refer should also be seen as benign and not limiting the privileges of any member.
- (6) (f) This is not possible under the current circumstances as, on occasion, "first time speakers" come forward after some people have spoken more than once. If rules are going to be generated around the completion of a list of first time speakers then it is up to Council to announce when the list has been fulfilled so that anyone wishing to conduct business on this motion or another motion related to a first time speakers' list will know that they can do so. Otherwise, it is confusing and potentially always out of order.

- (7) (d) What is a "proceeding" in this case? When can it be deemed that Council has conducted "further proceedings"?
- (8) See my comments on (6) (f) about the first time speakers' list.
- (9) (a) This seems undemocratic. While the idea makes sense, it doesn't appear that this is prescribed anywhere in BRO. Since the motion to reconsider depends upon a 2/3 majority, why should it also be dependent on someone voting in the majority to put it forward? This institutes 2 kinds of limits on a motion to reconsider, when only 1 is prescribed in BRO.
- (10) The motion under (d) from this list does not appear above in 3.12. This motion is not in order and not permissible under BRO. It is a main and major difference between Robert's Rules of Order and BRO and should likely just be removed. There is no mechanism for members to "move to ask a question" of another member. While questions may be asked of staff, those are in the context of other motions and do not stand on their own.
- (12) This is dangerous if left in this condition. It is important to be clearer about what is meant here. Suspending which Rules of Procedure? A motion to do this should specify exactly which rules are being suspended. To suspend them, generally, suggests that nothing in this bylaw would apply and that cannot be the case.

3.16 Editorial and Other Changes

- (1) "resolution" This word appears about a dozen times in the bylaw but does not have a definition. It appears at 3.10 (6), 3.10 (7), 3.14 (7) (b), 3.16 (1), 3.17 (a), 3.17 (b), 4.1, 8.4 (a), Appendix G (b), Appendix G (g). It is important to define this term as it actually seems to mean something with respect to Parliamentary procedure. It would be better to use either "decision" or "motion" as needed/applicable or to reword things to avoid the use of the phrase "resolution".
- (1) (h) This is not the current practice. In the recent past, when an error was made with respect to a Citizen Committee Report of the LGBTQ Advisory Committee, an entire paragraph was redacted rather than having the erroneous word corrected. If this will be the practice going forward, it should be stated clearly here.

SECTION 4 - COMMITTEE OF THE WHOLE

See comments in section 1.1. This is mostly meant for the purposes of Parliament, not a
City Council. While the rules here are meant to be "less strict" they don't seem to apply
at all to City Council since the outcomes of all decisions as part of a Committee of the
Whole have the same force and effect of all other decisions. The only "slackening" of

the rules is with respect to "lifting items". Also see the definitions in Stanford to see how this relates more specifically to Parliament.

5.1 Standing Committees

- (a) Should be spelled General **Issues** Committee.
- (e) Should be spelled Emergency and Community Services Committee for consistency with https://www.hamilton.ca/council-committee/council-committee-meetings/emergency-and-community-services-committee.
- (f) Should be spelled Audit, Finance **and** Administration Committee for consistency with https://www.hamilton.ca/council-committee/council-committee-meetings/audit-finance-a nd-administration-committee.

5.2 Appointment of Standing Committee Chairs and Vice Chairs

- "Vice Chair" as in the title should be the preference over "Vice-Chair" as is used throughout this section. It's clearer and easier to read.
- (1) and (1) (a) change "2nd" to appear the same way as it does in (1) (b). All of these ordinal superscripts should be the same size as the regular text (to make it clearer to those who may be using a screen reader or other assistive device to view text).

5.3 Standing Committee Membership

• (8) As previously mentioned, there should be a section on the Mayor that clearly outlines his powers separate from these individual mentions. It is confusing when only found in specific sections. It can be reiterated here but the prevailing new paragraph on the Mayor should also be cited for a full list of their powers. Some of this is currently in an Appendix but it should be moved into the body of the document so that it is not overlooked. Roles and responsibilities seem to be important enough to be included in the bylaw itself.

5.4 Quorum

See comments under "Quorum" in section 1.1.

5.5 Standing Committee Meeting Times

See comments on 3.2.

5.7 Standing & Selection Committee Reports

• This is not sufficient time. See comments on section 3.2.

5.8 Rules of Procedure

 There should be no need for separate Rules of Procedure for Standing Committees if other changes that have been recommended are taken into account. In general, no matter where they are taken into account or not, there could easily be a general section on Rules of Procedure and then exceptions or a short list of those that apply to different bodies.

5.10 Order of Business

See comments on 3.11.

5.11 Delegations

- (1) (a) This does not make sense. It is deferring the power of accepting a delegation request to Council, rather than to the Standing Committee by waiting for the matter to come before Council in the form of a Standing Committee report. This is an unnecessary delay and one that prejudices the rights of individuals to address Standing Committees. Also, the Standing Committee should be empowered to make a decision about whether or not it wants to hear a delegation immediately or have it postponed. This should not be the purview of the Office of the City Clerk.
- (1) (b) The hyperlink to www.hamilton.ca is not active.
- (3) The word "entertain" seems inappropriate here.
- (11) Should be "add a delegation".

5.13 Communication Items

• See comments on 3.10 and comments about not including repetitive sections for City Council and Standing Committees (cf. comments on 5.8, for instance).

5.14 Reconsideration of a Matter Decided Within a Standing Committee Meeting

Same as 5.13.

5.15 Conflict of Interest Declarations

• Same as 5.13.

5.16 Special Meetings of a Standing Committee

• Same as 5.13 except there should be no additional powers granted to the Mayor with respect to Standing Committees of which he is an ex officio member. It should be the Chair (and at the purview of a majority of members).

5.17 Change to a Scheduled Committee Meeting

Same as 5.13.

7.7

- Probably useful to have a title for all such numbered sections throughout the document (e.g. 2.1, 2.2, 7.1 7.7, 8.1 8.4, 9.1).
- (b) "Closed Meetings" has capitals at the beginning of each word. This is not consistent throughout.

8.1

• City the MA and list from it in quotation marks. Cite the specific section.

8.2 and 8.3

 Both of these sections could be worked together and made part of 8.1 with a proper title.

8.4

• (a) This should also include a brief description of the matter and specifically what part of the MA is being invoked, not just to the general sentiments of a "closed meeting". As much detail as possible should be used so as not to give away any confidential or private information but to allow the public to know what is being spoken about (i.e. Red Hill Valley Parkway road conditions report, Sewage spill into Cootes Paradise, Integrity Commissioner complaint against a member of an Advisory Committee). None of the titles given as examples would reveal any confidential information and are all factual representations of the matters at hand.

10.1 and 10.2

• This is not enough time. See my comments on 3.2.

SECTION 11 - GENERAL

• The public copy should be dated and signed and the section for City Clerk should be updated to read A. Holland.

Appendix G - Roles of Council, Mayor and Committee Chair

• This should be incorporated into the body of the bylaw in the places it makes the most sense so that it's not an afterthought. Some of the information here is crucial to the understanding of the document and procedures as a whole and having it in an Appendix is confusing.

Joey Coleman Ward 2 Resident Hamilton, ON Personal Information

January 30, 2021

ATTN:

Council of the Corporation of the City of Hamilton

via email: clerk@hamilton.ca

This letter serves as my official submission to the Corporation of the City of Hamilton regarding a proposed bylaw banning letters critical of Council behaviour from being submitted as public correspondence.

In what has become an annual exercise for Hamilton City Council, Deputy City Clerk Janet Pilon, City Clerk Andrea Holland, and City Solicitor Nicole Auty have brought forth recommendations to this Council which will violate the Charter Rights of Hamiltonians to express themselves regarding matters of public interest.

In a democracy, citizens have the right to express displeasure with their elected government.

The proposal to forbid correspondence to Council is anti-democratic and violates the oath of office sworn by members of Council. (Suggested reading: Grade 10 civics curriculum)

It is quite unfortunate that it is both being proposed and being considered. In a responsibly managed municipal government, this kind of proposal would not even be contemplated, let alone made a priority during a public health emergency.

It is even more unfortunate that I must submit this correspondence with the expectation this matter will end up in Superior Court of Justice.

<u>Legislation</u>

- 1) Constitution Act, 1982, Canadian Charter of Rights and Freedoms;
- 2) Municipal Act, 2001, SO 2001, c 25;
- 3) City of Hamilton Act, 1999, SO 1999, c 14, Sch C;
- 4) Codes of Conduct Prescribed Subject Matters, O Reg 55/18.

Decision of Council Subject to the Charter

In this letter, I provide Council with basic arguments against the proposed provisions of the Procedural By-Law. One can cite the Grade 10 Ontario Curriculum for civics classes to explain how the proposed Bylaw is illegal.

The City of Hamilton is a municipality incorporated under the City of Hamilton Act, 1999. Decisions of the municipalities are bound by the Charter of Rights and Freedoms¹.

In a 1994 decision, Supreme Court Justice McLachlin (as she was then) wrote "It is important that municipalities not assume powers which have not been conferred on them, that they not violate civil liberties ... and that abuses of power are checked."²

In banning correspondence critical of Council behaviour, the City of Hamilton assumes powers to regulate speech not conferred upon it and violates civil liberties for the purpose of preventing citizens from speaking against abuses of power.

City of Hamilton v. Hamilton Distillery Co (1907) 38 S.C.R. 239 is also informative regarding the need for the Courts to intervene when municipalities seek to confer powers upon themselves which are not granted in law.

Expressive Activities, Municipal Government, and the Charter

Gammie v. Town of South Bruce Peninsula³ is informative regarding how the Charter protects the expressive activities of residents in seeking redress with their elected municipal governments.

In the decision, the Ontario Superior Court Justice Price explains the importance of public expressive activities to the ability of residents to exercise their rights in our democracy.

Put simply, municipalities cannot regulate expression because they do not like it.

¹ Godbout v. Longueuil (City), [1997] 3 SCR 844, par. 51.

² Shell Canada Products Ltd. v. Vancouver (City), [1994] 1 S.C.R. 231.

³ Gammie v. Town of South Bruce Peninsula, 2014 ONSC 6209

Hamilton City Council has expressed, in a 13-2 vote on February 25, 2015, the position that members of Council should be allowed to assault individuals without any consequence. The current proposed by-law bans residents from expressing disagreement with this decision in the form of written correspondence on Council agendas.

Without a doubt, written correspondence in Council agendas is expressive activity protected by the Charter. Unless a letter takes an unprotected form, "such as violence or threats of violence", a municipality cannot ban the expression.

While Hamilton City Council may not like it when residents express the opinion that elected officials should behave themselves in a manner befitting their office, it does not have the power to prohibit residents from expressing this view to Council.

Written correspondence to Council is at the centre of municipal political life in Hamilton. During COVID, these letters take greater importance as the opportunities to engage in political life are curtailed by necessary public health measures.

Council responds to issues when it receives many letters upon matters of importance to residents of Hamilton. Debates resulting from correspondence are lengthier and more substantive as the number of letters increase. For example, the recent debates regarding the police budget are reflective of this in practice.

Similarly, while a single letter regarding traffic safety in a particular location is ignored by Council, a series of letters regarding a particular location will usually result in debate by Council. Usually, Council acts in response to letters.

This proves the effectiveness and importance of correspondence to the expressive activities of the citizens of Hamilton.

Residents who submit letters to Hamilton City Council regarding the behaviour of Councillors do so for purposes protected by guarantee of Free Expression provided in Section 2(b) of the Charter of Rights.

Clearly in this instance it must be explained that "the purpose of the guarantee is to permit free expression to the end of promoting truth, political or social participation, and self-fulfillment".⁴

Political violence, intimidation, threats, and other improper behaviours engaged in by some members of Hamilton City Council discourage political and social participation.

Letters regarding Council behaviour serve to advance the goal of promoting political and social participation.

Citizens are discouraged from expressing themselves when met with hostility, intimidation, and disparaging remarks by elected officials.

Redressing this behaviour is critically important to our democracy.

Letters regarding this topic are important to the health of our democracy inviting the strongest possible protection under the Charter.

Council's can no more prohibit these letters, than Council's can prohibit individuals from bringing signs into Council Chambers.

As Superior Court Justice Price wrote in *Gammie*, Council can only regulate "the forms of signs that could be brought into Council Chambers to paper or cardboard without solid handles".⁵

The Charter protections for written correspondence to Council are the same as attending a Council meeting. The legal principles espoused in *Gammie* apply to written correspondence.

It is true the Municipal Act does not implicitly require correspondence to be included in public agendas. However, when read in its entirety, the only means of meeting the requirements of open meeting rules⁶ is for correspondence to be published as part of the Council agenda.

Thus, while ending all critical correspondence would remove the Charter violation in this instance, it would create a separate violation of law.

⁴ R. v. Zundel, [1992] 2 S.C.R. 731

⁵ Gammie v. Town of South Bruce Peninsula, 2014 ONSC 6209 at par. 92

⁶ Municipal Act, 2001, SO 2001, c 25, s 239

Once a government creates a platform for expression, that platform is subject to the Charter⁷.

In *Greater Vancouver*, the transportation authority forbade political ads. The ads in question in this case were encouraging people to vote.

The Supreme Court of Canada held that government cannot forbid speech for political reasons.

"The policies allow for commercial speech but prohibit all political advertising ... In sum, the policies amount to a blanket exclusion of a highly valued form of expression in a public location that serves as an important place for public discourse. They therefore do not constitute a minimal impairment of freedom of expression.8"

Council cannot choose to only accept correspondence which it likes.

"It is clear from this Court's s. 1 jurisprudence on freedom of expression that location matters, as does the audience", Supreme Court Justice Deschamps wrote in the unanimous decision.

In the matter presently in front of Council, the location of the expression is the [virtual] City Council Chamber. The audience is the City Council.

These factors further contribute to both the appropriateness of the public addressing concerns about Council behaviour by means of written correspondence, and to the strong Charter protections afforded to this expression.

"Since the Canadian Charter clearly applies to the provincial legislatures and governments, it must, in my view, also apply to entities upon which they confer governmental powers within their authority. Otherwise, provinces could (in the manner outlined earlier) simply avoid the application of the Charter by devolving powers on municipal bodies" wrote Supreme Court Justice Major in *Godbout* explaining why the Charter applies to municipalities.

⁷ Greater Vancouver Transportation Authority v. Canadian Federation of Students — British Columbia Component, 2009 SCC 31, [2009] 2 S.C.R. 295

⁸ Ibid at par. 77

⁹ Ibid at par. 78

¹⁰ Godbout v. Longueuil (City), [1997] 3 SCR 844, par. 51

It follows that Council cannot remove Charter protections from correspondence by devolving an authority to the City Clerk.

Simply put, the City Clerk cannot exercise the authority to decide when a letter is, or is not, critical of Council.

*Irwin Toy*¹¹ provides an effective reference explaining how the Divisional Court will strike down this Bylaw.

The Court will determine that correspondence regarding Council behaviour is within the scope of protections afforded by the Charter.

The Court will find that the purpose of this government action is to restrict that freedom of expression. An Oakes Test will be applied, and this bylaw will not be saved by Section 1 of the Charter.

I will briefly address *Bracken v. Fort Erie*¹² before providing an Oakes Test for Council's consideration.

The facts of *Bracken* need not be exhaustively detailed in this letter. Much like the basics of civics taught in Grade 10, the facts of this case should already be well known to members of Hamilton City Council

Writing for a unanimous Ontario Court of Appeal, Justice Miller provides some key insights relevant to the matter before Council in the proposed bylaw.

In paragraph 54:

"Freedom of expression has received broad protection in Canadian law, not only through the Charter, but also through legislation and the common law. As Rand J. noted in Saumur v. City of Quebec, 1953 CanLII 3 (SCC), [1953] 2 S.C.R. 299, at p. 329: "Strictly speaking, civil rights arise from positive law; but freedom of speech, religion and the inviolability of the person are original freedoms which are at once the necessary attributes and modes of self-expression of human beings and the primary conditions of their community life within a legal order." Section 2(b) further entrenches the limits on government action in order to safeguard the ability of persons to express themselves to others. As expressed in Irwin Toy Ltd. v. Quebec

¹¹ Irwin Toy Ltd. v. Quebec (Attorney General), [1989] 1 S.C.R. 927

¹² Bracken v. Fort Erie (Town), 2017 ONCA 668

(Attorney General), 1989 CanLII 87 (SCC), [1989] 1 S.C.R. 927, at pp. 968-969:

Freedom of expression was entrenched in our Constitution and is guaranteed ... so as to ensure that everyone can manifest their thoughts, opinions, beliefs, indeed all expressions of the heart and mind, however unpopular, distasteful or contrary to the mainstream. Such protection is, in the words of both the Canadian and Quebec Charters, "fundamental" because in a free, pluralistic and democratic society we prize a diversity of ideas and opinions for their inherent value both to the community and to the individual. Free expression was for Cardozo J. of the United States Supreme Court "the matrix, the indispensable condition of nearly every other form of freedom" (Palko v. Connecticut, 302 U.S. 319 (1937), at p. 327); for Rand J. of the Supreme Court of Canada, it was "little less vital to man's mind and spirit than breathing is to his physical existence" (Switzman v. Elbling, 1957 CanLII 2 (SCC), [1957] S.C.R. 285, at p. 306). And as the European Court stated in the Handyside case, Eur. Court H. R., decision of 29 April 1976, Series A No. 24, at p. 23, freedom of expression:

... is applicable not only to "information" or "ideas" that are favourably received or regarded as inoffensive or as a matter of indifference, but also to those that offend, shock or disturb the State or any sector of the population. Such are the demands of that pluralism, tolerance and broadmindedness without which there is no "democratic society"." (par. 25)

"There can be no question that the area in front of a Town Hall is a place where free expression not only has traditionally occurred, but can be expected to occur in a free and democratic society. The literal town square is paradigmatically the place for expression of public dissent."

Continuing, Justice Miller writes in paragraphs 57 and 58:

I acknowledge that several of the affiants attested that Mr. Bracken's speech was incomprehensible, and that the application judge made that finding. But again, the finding was unsupportable. Some affiants, up on the balcony or elsewhere on the second floor, might not have heard him distinctly. Others, who distinctly heard him saying "kill the bill" might not have had sufficient context to understand the message. That did not make his speech "incomprehensible", with the insinuation – made in various places in the Town's affidavits – that Mr. Bracken was raving. To the contrary, Mr. Brady, watching from the atrium and well-acquainted with Mr.

Bracken's grievances, heard Mr. Bracken and clearly understood what he was saying. He didn't like it.

Mr. Bracken's speech, that day, was directed towards protesting the expected adoption of a by-law that he understood to be promoting the interests of a marijuana facility across from his home. He wanted the by-law defeated. He also criticized the members of Town Council. No doubt, they did not like being called liars and communists. Mr. Brady did not like Mr. Bracken calling for him to be fired. On crossexamination, he stated that Mr. Bracken had no right to say so. He viewed it as a threat to his livelihood. The language was neither polite nor restrained. But as this Court pointed out in Cusson v. Quan, 2007 ONCA 771, 87 O.R. (3d) 241, rev'd 2009 SCC 62, [2009] 3 S.C.R. 712, at para 125: "(d)emocracy depends upon the free and open debate of public issues and the freedom to criticize the rich, the powerful and those ... who exercise power and authority in our society... Debate on matters of public interest will often be heated and criticism will often carry a sting and yet open discussion is the lifeblood of our democracy."

The Oakes Test

The Oakes Test¹³ is the burden which Hamilton City Council must meet to prove to the Divisional Court that this ban is a reasonable limit on free expression that can be demonstrably justified in a free and democratic society.

Firstly, is the objective of the bylaw both "pressing and substantial"?

The objective of the bylaw is to stifle public discourse regarding the behaviour of elected officials. This is not a pressing government objective.

It is not even a government objective. It is a personal political desire of some members of Hamilton City Council.

This bylaw fails this first test.

Most laws subjected to Charter challenge meet this hurdle, that this bylaw will not speaks to just how far outside the range of reasonableness it is.

¹³ R. v. Oakes, [1986] 1 S.C.R. 103

For the sake of legal argument, let's assume a Court decided to allow the Bylaw to pass the first test.

The propose of this assumption is purely to demonstrate how the bylaw will fail on the second stage proportionality analysis.

The infringement is rationally connected to "the law's purpose" which is to stifle expression critical of Council. There is no other means to achieve this purpose. Therefore, it will meet this part of the analysis.

The infringement is not a minimal impairment. It is a blanket ban on all discourse regarding Council behaviour. It fails on minimal impairment.

The proportionate effects are much too grave for any Court to entertain allowing this bylaw to stand. Beyond stifling debate regarding Council behaviour, the bylaw will allow behaviours which intimidate residents and prevent them from engaging in democratic processes.

Ultra Vires

Ontario Regulation 55/18 requires municipal councillors to codify a Code of Conduct which includes a section regarding "respectful conduct, including conduct toward officers and employees of the municipality or the local board, as the case may be."

The proposed ban on critical letters regarding the conduct of members of council not only violates the Charter. It is *ultra vires*.

Council is required by regulation to consider submissions from the public regarding Council behaviour. The Council cannot consider the letters in closed session as per the open meeting requirements.

The Council cannot pass bylaws beyond the powers granted to it, or restricted by, the Province of Ontario.

Conclusion

The proposed ban on correspondence critical of Council behaviour fails all legal tests.

If Council wishes to proceed to pass this illegal bylaw, I will take the necessary steps to challenge at the Superior Court of Justice to uphold the fundamental foundations of our democracy.

Submitted by:

Joey Coleman Resident, City of Hamilton

Request to Speak to Committee of Council

Submitted on Thursday, December 24, 2020 - 1:00 pm

==Committee Requested==

Committee: Audit, Finance and Administration Committee

==Requestor Information==

Name of Individual: Sanaullah Chaudhry

Name of Organization:

Contact Number:

Email Address:

Mailing Address:

Reason(s) for delegation request:

Urgent Request: My Water Bill of total \$2,193.34 (for October 18 - Dec. 18th through Alectra-Bill) is very shocking for me. To survive, special request to waive one time this 'water-bill of \$2193' to be highly appreciated. All details of my property / House available to present upon acceptance of my request.

Regards Sanaullah Chaudhry

Will you be requesting funds from the City? No

Will you be submitting a formal presentation? No

Sanaullah Chaudhry Delegation to Audit, Finance & Administration Committee – February 4, 2021

Four Points:

- #1. Water Bill (Killing \$2K++) in Covid 2020-2021.
- #2. Property Tax Highest within Neighbourhood
- #3. City didn't Issue parking permit (req'd 4)
- #4. Zoning CMU5 Property Commercial 25% Versus Residential 75%. (Why can't total residential to avoid terrible costs I have to pay).

'Hamilton City & SC 4-Pts Agenda' Ref. XXXXXXX

Owner (Lucky or UnLucky? - 'XXXXXXXXXXXX, Hamilton'

Sanaullah M. Chaudhry B. Eng, PE, CET, PMP, RPM (Sr. PM Consultant – GOC)

Basic Introduction; Special Thanks to City for Opportunity Background 2018 & CMU Zoning & Fact Sheet & Where to Go Next!)

'City & 4-Pt. AGENDA/ Action-Items 'XXXXXXXX'

- 1. My Water Bill (Killing \$2300+ Bill & Covid 2020-2021Damages
- 2. My Property Taxes 'XXXXXXX' Highest in Neighbourhood (Discriminated \$12K; MPAC & Tax Office Inerpretation!!)
- 3. My Parking Permits Why Zero (Even High Tax Paid)- Reqd 4 Permits!
- 4. **Zoning Change CMU5?** (Commenrcial 25% Vs residential 75% Residential; Why Can't Total Residential (as Comm Expenses Killing)

XXXXXXXXXXXX, Hamilton Water Billing Dispute Delegation to Audit, Finance & Administration Committee

Summary

The current property owner purchased XXXXXXXXXXXXXXXX, Hamilton in October 2018, which consists of rental commercial and residential units. There is a single water service line/water meter for the entire property. The commercial tenant is the water account holder (commercial tenant held accounts are permitted whereas, all residential accounts established after June 1, 2018, must be opened and billed in the name of the registered owner(s) of the property). An online review shows the commercial unit is operating as a café/hookah bar. The property tax bill is directed to the owner/landlord with a Mississauga mailing address. There have not been any water arrears transfers to the property tax roll but there has been three Municipal Law Enforcement (MLE) charges applied to the tax roll during 2020.

Table 1 on page two provides the property's water and wastewater/storm billings for the past year. As reflected on the July 2020 water bill, the water consumption for the property essentially doubled from previous months. The water consumption further increases by over 200% with the October 2020 water bill. The following two month's bills reflect massive increases in water consumption suggestive of leaking condition(s) that have progressively worsened over time.

Property owners need to be aware that although billing and payment notifications may be sent to a commercial tenant, subsection 398 (2) of the *Municipal Act*, 2001 authorizes the City to add unpaid water and wastewater fees and charges to the tax roll of the property to which the public utility was supplied.

All internal plumbing is private property and as such, any repairs are the responsibility of property owners. Property owners of rental properties that have redirected the water bill to a commercial tenant need to keep in regular contact with their tenants to be aware of changes in the property's water consumption that may indicate potential plumbing issues that lead to high water bills. Additionally, as set out in the *Residential Tenancies Act*, 2006, landlords may carry out periodic inspections of rental units to ensure they remain in a good state of repair.

In December 2020, following several months of increasing water bills, the property owner contacted Alectra Utilities (the City's water billing agent) requesting that the November and December 2020 water billings be outright waived. The City's Water Leak Adjustment Policy (Policy) may provide limited financial relief to eligible customers, however, commercial landlords (apart from non-profit housing) are not eligible applicants.

City's Water Bill Adjustment Policies

City Council has approved water bill adjustment polices (Water Leak Adjustment Policy and the Extraordinary Circumstance Policy) that provide staff the authority to allow bill adjustments under specific circumstances, however, an "Income Producing Residential

Rental Property" is excluded under these policies. The term "income property" should not be confused to mean that rental income exceeds the property's operating costs (mortgage, property taxes, utilities, maintenance, etc). Many landlords do not "profit" from the rent exceeding the aforementioned costs but recoup operating cost recovery shortfalls when selling the property by means of the capital appreciation of the property over time.

The *Municipal Act, 2001* does include an "anti-bonusing" clause whereby a municipality shall not assist directly or indirectly any commercial enterprise through the granting of bonuses that includes giving a total or partial exemption from any levy, charge or fee. This consideration is why landlords are excluded from being eligible under the bill adjustment policies that any Ontario municipality may offer. This is also why Hamilton may provide an adjustment to registered non-profit housing (social housing) providers. The City's Legal Services division have reviewed the existing water bill adjustment policies and continue to recommend that income producing properties be considered as commercial so that such policies be restricted to residential and non-profit customers to comply with the *Municipal Act*.

TABLE 1

Read Date	To	tal Bill (\$)	Consumption (m3)	Read From	Read To	Avg Daily (m3)	Billing Days
12/18/20	\$	902	259	8717	8976	8.355	31
11/17/20	\$	1,292	375	8342	8717	13.889	27
10/21/20	\$	575	162	8180	8342	4.765	34
9/17/20	\$	205	54	8126	8180	1.862	29
8/19/20	\$	283	76	8050	8126	2.303	33
7/17/20	\$	237	63	7987	8050	2.032	31
6/16/20	\$	116	27	7960	7987	0.818	33
5/14/20	\$	117	28	7932	7960	0.966	29
4/15/20	\$	121	29	7903	7932	0.967	30
3/16/20	\$	122	30	7873	7903	1.111	27
2/18/20	\$	150	37	7836	7873	1.156	32
1/17/20	\$	110	26	7810	7836	0.839	31

Alectra Utilities (AUC) Customer Contact Timeline

12/04/20 - Owner calls AUC advised that account is in tenant's name but since there are multiple units, he pays the water portion. AUC advised water consumption began increasing in June 2020 and suggested a water meter read be taken by the customer from the water meter to confirm reads and to check for leaks. AUC advised of use of dye test with toilets to identify flapper issues, use of water meter to identify low flow leakage and availability of optional payment arrangements.

12/07/20 – Owners contacts AUC inquiring about relief programs and was advised no credit programs were available however can offer payment arrangements (tenant has authorized owner). AUC suggested waiting for December bill to include in a payment arrangement.

12/24/20 – Owner contacts AUC requesting high water billings be waived. AUC explained can only offer payment arrangement only. Owner advised heard a hissing noise and had it fixed. Owner escalated call with AUC supervisor who also advised no credit relief available but could offer payment arrangement. Owner stated the water supplies residential units, as well as, commercial and wanted the water billing to be moved to the residential electric account. AUC supervisor advised doing so still does not permit consideration under the City's Water Leak Adjustment Policy. Owner stated he would dispute City Policy with Council.

12/24/20 - Owner entered into a 12-month payment arrangement for outstanding water/wastewater balance of \$2,193.34 with the first installment commencing with the January 2021 water bill.

Request to Speak to Committee of Council

Submitted on Monday, January 11, 2021 - 11:36 pm

==Committee Requested==

Committee: Audit, Finance & Administration Committee

==Requestor Information==

Name of Individual: Robert Ridley - on behalf of Olympia Gerl (homeowner)

Name of Organization:

Contact Number:

Email Address:

Mailing Address:

Reason(s) for delegation request:

Water Billing Overcharging Anomaly – Senior level escalation Ref#XXXXXXXX

- usage billed for January (104m3) and March (76m3) were entirely unprecedented, while February reverted back to normal (3 m3)
- we have been attempting to remedy this over the last months with Alectra, but they have been unwilling to date
- A plumber has done a full inspection of all pipes and fixtures, and has confirmed in writing that there is no issue or evidence of flooding, leakage or anything else that would account for the high charges; his final assessment is that the problem is the meter.
- There are further details I would like to inform the committee and request she is credited back for the overbilling

Will you be requesting funds from the City? No

Will you be submitting a formal presentation? No

XXXXXXXXXX, Hamilton Water Billing Dispute Delegation to Audit, Finance & Administration Committee

Summary

Table 1 on page 3, provides the recent water and wastewater/storm billings for XXXXXXXXX (the property owner is the account holder). The crux of the owner's water billing issue lies with high water bills experienced in January and March 2020, while the owner was out of the country for a 3 month period. Unfortunately, the water supply to the home was not shut off despite the property being unoccupied. The City has long recommended shutting off water to one's property when away to avoid unauthorized water use or undetected leakage occurring at unoccupied properties.

Council approved water bill adjustment policies (City's Water Leak Adjustment Policy or Extraordinary Circumstance Policy) explicitly exclude adjustments where a dwelling is unoccupied and/or vacant for more than 72 hours. Staff conducted an online review to determine if Hamilton's bill adjustments policies differ with respect to the unoccupied/vacancy exclusion condition. Very few Canadian municipalities offer water bill adjustment policies recognizing that property owners are responsible for private plumbing and the need to maintain/safeguard water using private infrastructure. The online review identified a small number of municipalities that offer water bill adjustment policies (Brantford, Guelph, London, Kitchener, City of Waterloo, Bancroft and Winnipeg) and without exception, the policies exclude adjustments where the home was unoccupied/vacant when the high water consumption occurred.

For extended absences, customers should consider shutting off the water supply (except where water is used for heating) and draining all the pipes and appliances. The City has long promoted the water supply be shut off when a property will be unoccupied/vacant for extended periods via Alectra customer service staff, newspapers ads, bill inserts (including the October 2019 and November 2020 Hamilton Water newsletter bill inserts) and on the City/Alectra websites. The use of on-bill messaging (message appears just below the water consumption bar chart on the right-hand side of the water bill) has also been introduced with the January to April monthly water bills providing the following advice:

Leaving on vacation? Make sure your home's water valve is turned off before you leave to prevent expensive water bills from flooded basements due to leaky pipes or water theft from your outside taps. www.hamilton.ca/leaks

Furthermore, the City's Waterworks Bylaw R84-026, Section 18(4) requires of all property owners: "Whenever any premises having a water service becomes vacant, the owner is required within twenty-four (24) hours of vacancy to notify in writing, the City to shut off the water at the street line." This requirement recognizes that vacant properties often experience undetected water leaks and/or unauthorized water usage. Following the ice storm in December 2013, the City began suggesting specifically to "snowbirds"

to request the City shut the water service at the street as extended power outages can cause water pipes to burst.

In some circumstances, customers raise concerns with respect to the accuracy/functioning of the water meter suggesting high water consumption is a result of inaccurate measurement. Hamilton like most water utilities purchase meters that must be manufactured in accordance with American Waterworks Association (AWWA) standards.

The City uses an industry standard positive displacement meter for residential metering purposes. These meters have a design that is extremely reliable and inherently mitigates over recording. It also provides an accurate record of consumption for many years before requiring replacement. 98% of water meters installed in Canada are positive displacement meters. The replacement of these devices is driven by the wear of mechanical parts that leads to the *under recording* of water consumption. As such, Staff is confident that the City's meters don't account for the high water bills.

When customers raise doubts that the water meter is malfunctioning leading to high consumption billing, like most municipalities Hamilton has a process whereby a customer may have their meter tested for accuracy with a <u>Water Meter Testing Request Form</u> available on Alectra's website. The customer is required to pay an applicable service fee that varies with the meter size; the 2021 meter accuracy test fee for residential meters (15 – 16mm) is \$314.56 plus HST (fee includes removal of existing meter and installation of replacement meter). Unless the meter is found to be registering more than 2% in error, then the property owner is responsible for the cost of the test and the replacement cost of the water meter; otherwise the cost is borne by the City. If the meter is found to be over-registering the water account would be adjusted down to reflect the amount of over-registering consumption that required a billing adjustment. As there is no evidence that the water meters over-register, the application of the meter test fee avoids unnecessarily removing meters from service before the end of their useful life.

The City is not obligated at its expense to remove a water meter for accuracy testing where a customer contends the meter is inaccurate. The City was successful in defending at trial a Small Claims Court claim from a customer that alleged that his water meter was not functioning however the plaintiff chose not to have the meter tested for accuracy (Radassao v. City of Hamilton et al). The decision noted that the customer by not having the meter tested could not prove any malfunction.

TABLE 1

Read Date	To	otal Bill (\$)	Consumption (m3)	Read From	Read To	Avg Daily (m3)	Billing Days
12/15/20	\$	37	7	1,792	1,799	0.212	33
11/12/20	\$	32	7	1,785	1,792	0.259	27
10/16/20	\$	35	7	1,778	1,785	0.233	30
9/16/20	\$	37	9	1,769	1,778	0.31	29
8/18/20	\$	50	12	1,757	1,769	0.343	35
7/14/20	\$	48	12	1,745	1,757	0.375	32
6/12/20	\$	29	3	1,742	1,745	0.097	31
5/12/20	\$	30	3	1,739	1,742	0.091	33
4/09/20	\$	30	5	1,734	1,739	0.179	28
3/12/20	\$	262	76	1,658	1,734	2.714	28
2/13/20	\$	26	3	1,655	1,658	0.111	27
1/17/20	\$	353	104	1,551	1,655	3.152	33
12/15/19	\$	36	10	1,541	1,551	0.37	27
11/18/19	\$	37	8	1,533	1,541	0.25	32
10/17/19	\$	33	8	1,525	1,533	0.296	27

The water billings have been paid in full to date as the account is on pre-authorized payment plan. There was no contact from the customer until her return in late March 2020 which suggests the water billings during the owner's absence were not reviewed when issued. If the January water bill was reviewed when issued in late January, at that point an investigation could have taken place to find the source of the high water usage.

As noted below, there have been repeated requests by the customer for financial compensation and Alectra consistently has advised (following consultation with City Finance) that due to the property being unoccupied during the periods of high metered water usage the requests were ineligible under City policies.

Additionally, the customer, via the ward councillor's office, has had several discussions with Hamilton Water Community Outreach between May and December 2020 that mirrored the discussion the customer has had with Alectra.

As previously noted, there are two adjustment policies approved by City Council for City staff to adjust a high water billing (City's Water Leak Adjustment Policy and the Extraordinary Circumstance Policy) however, both polices exclude adjustments where a dwelling is unoccupied and/or vacant for more than 72 hours.

Alectra Utilities (AUC) Customer Contact Timeline

Jan 29/20 - As the Jan 17th water bill was high, AUC attempts to contact customer by phone however the phone number listed was not in service, no number found on Canada 411 & no email on file. Clerk issued meter re-read which confirmed meter reading so bill issued.

Mar 24 - Olympia (owner) called regarding high usage, she advised she was away for 3 months, AUC customer service representative (CSR) advised to check for leaks. Customer stated toilets are brand new. CSR asked for read from meter (WLA (water leak adjustment) not mentioned as property vacant).

Mar 25 – Olympia called with read which confirmed reads were OK and that usage had returned to normal – she advised she was away Dec 25 to Mar 24 and requested the meter be tested, she was advised work order was pending to confirm reads.

Apr 20 – Olympia called requesting reimbursement and the CSR suggested in future to turn water off when going away for extended period, customer requested escalation.

Apr 22 – AUC supervisor called Olympia and advised this type of usage usually suggests leaks, suggested read meter overnight to monitor and do dye test on toilets. Customer responded received our stickers and everything is fine. AUC supervisor advised would call back to discuss overnight reads, also advised she may need to get a plumber and understood her concerns about having someone in home during Covid 19.

Apr 24 – AUC supervisor called back to discuss overnight reads.

May 12 - Olympia's son in law Bob called, he advised same info Olympia – she was away, no leaks, no run on toilets, he requested relief as she is a senior, was advised by CSR no relief available.

May 14 – Email received from the City to call Bob. AUC supervisor called and Bob was concerned we could not explain the high – low – high invoices and wanted compensation.

May 19 - CSR spoke with Olympia – same matters as raised on May 14 were discussed.

June 3 – Bob spoke to Supervisor who pointed out that both years Olympia was away she experienced higher usage (2018 and 2020) in 2019 when she was home there were no issues (2018 Jan 5-Feb 8 16m3, Feb 8-Mar 6 28m3, Mar 6 to Apr 3 13m3)

June 9 – Bob spoke to AUC supervisor, requested credit as Olympia is senior on fixed income, was explained cannot offer compensation, also explained meter test option and payment arrangements. Bob requested further escalation before going to the media.

June 18 – AUC Manager called and spoke with Bob. Bob advised Olympia was away, 2 people were checking on home in her absence. No one detected water being used, they checked for leaks including toilet test and found none. Manager asked if plumber had checked premises, Bob advised no. Bob asked for compensation as Olympia is on fixed income and this is causing financial strain. Bob also stated he feels she shouldn't have to pay for the charges because she was not home. Manager explained how meter works and that since water had flowed through the meter charges apply. Customer requested a call back.

July 28 – Unexplained High Water Usage Billing Adjustment Request Application Form received from customer via email – was sent to call centre supervisors who forwarded to billing supervisors.

July 30 – Form reviewed and declined after billing clerk reviewed with COH. Denied as criteria not met, email was sent to billing supervisor to advise whomever was dealing with customer to advise the customer.

Aug 10 – CSR called and spoke to Olympia and advised did not meet criteria under the Extraordinary Circumstance Bill Adjustment Policy.

Aug 11 – Bob spoke with CSR who transferred him to COH to advise why criteria was not met.

Aug 18 – Email received from Hamilton Water Meter Operations inquiring why was customer contacting them.

Nov 25 – Email received from COH by call centre supervisor to contact Bob. Bob stated that they did not receive written notification for reason their request was declined under Extraordinary Circumstance Bill Adjustment Policy, advised declined verbally on Aug 10. Was advised of option to appear before City's Audit, Finance & Administration Committee. Supervisor advised would review request form and call back.

Dec 2 – Supervisor called Bob to advise reviewed compensation request and advised request is declined - reiterated option to appear before City Committee, Bob advised he would contact councillor, MPP and would take further action.



CITIZEN COMMITTEE REPORT

То:	Audit, Finance and Administration Committee; General Issues Committee and City Council
From:	Hamilton Status of Women Committee
	(Autumn Getty, Chair)
Date:	December 16, 2020
Re:	Status of Women Advisory Committee Donation of Remaining 2020 Budget Allocation – November 26, 2020 Motions 4.3 & 4.4

Recommendation:

That the budget allocations approved at the November 26,2020 Status of Women Committee Meeting for motions 4.3 and 4.4 be approved as follows (motions listed below under background).

Motion 4.3

- That the Status of Women Committee allocate the remainder of their annual Budget and reserves amount for 2020 of approximately \$4000 as one-time donations of \$444.00 to each of the following organizations that are in alignment with the mandate and goals of the committee and are working to promote the safety and well-being of woman-identified and non-binary Hamiltonians
 - An allocation of \$444.00 from the Status of Women Committee's budget to be provided to Native Women's Centre's Honoring the Circle program to support ongoing community activities that assist women during COVID.
 - An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Sexual Assault Centre of Hamilton and Area

(SACHA) to support ongoing community activities that assist women during COVID.

2 |

- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to the Hamilton Jewish Family Services to support ongoing community activities that assist women during COVID.
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Interval House Hamilton to support ongoing community activities that assist women during COVID
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Good Shepherd Centres: Martha House to support ongoing community activities that assist women during COVID
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Mission Services Hamilton: Inasmuch House to support ongoing community activities during COVID
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Mission Services Hamilton: Willow's Place to support ongoing community activities during COVID
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Care-Mongering via the appropriate flowthrough charitable organization to support ongoing community activities during COVID
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to Keeping 6 via the appropriate flow-through charitable organization to support ongoing community activities during COVID

Motion 4.4

That the Status of Women Committee support A. Getty, as the Chair, to find out
who the flow through charities are for Keeping 6 and Care-Mongering, and to
then contact the flow through charities, to provide donation of \$444.00 each to
Keeping 6 and Care mongering on behalf of the Status of Women Committee.

Background:

The Status of Women Committee carried the following 2 motions at its meeting on November 26,2020.

- 4.3 That the Status of Women committee provide a donation of \$444.00 to each of the following 9 organizations that are in alignment with the mandate and goals of the committee:
 - 1) Native Women's Centre Honouring the Circle program
 - 2) SACHA

3 |

- 3) Hamilton Jewish Family Service
- 4) Interval House
- 5) Good Shepherd Centres: Martha House
- 6) Mission Services Hamilton: Inasmuch House
- 7) Mission Services Hamilton: Willow's Place
- 8) Care-mongering
- 9) Keeping 6
- 4.4 That the Status of Women Committee support A. Getty, as the Chair, to find out who the flow through charities are for Keeping 6 and CareMongering, and to then contact the flow through charities, to provide donation of \$444.00 each to Keeping 6 and Care mongering on behalf of the Status of Women Committee.

This is the corresponding Citizen Committee Report.

More information about the charities listed above can be found on the following websites.

- Native Women's Centre Honouring the Circle program http://www.nativewomenscentre.com/honouring-the-circle
- Sacha- https://sacha.ca/
- Hamilton Jewish Family Service- https://www.hamiltonjfs.ca/
- Interval House- https://intervalhousehamilton.org/
- Martha House https://www.goodshepherdcentres.ca/services/martha-house/
- Inasmuch house https://mission-services.com/programs-andservices/inasmuch-house/
- Willow's Place https://mission-services.com/programs-and-services/willowsplace/
- Care-Mongering https://redbook.hpl.ca/redbook/record/HAM5638

Keeping Six - https://redbook.hpl.ca/redbook/record/HAM4868

Analysis/Rationale:

These recommendations are in line with the mandate and goals of the Status of Women committee. These amounts are allocated for in the 2020 budget submission. The Status of Women Committee want to allocate the remainder of its \$4000 available

4 |

funding as part of its budget due to it not regularly meeting during the COVID-19 pandemic.

With respect to motion 4.3; The Status of Women committee selected organizations that are either primarily women serving or contains programs that are primarily women serving. All organizations have been recognized for providing tremendous support to their communities during the COVID-19 pandemic. As such, the Status of Women Committee would like to provide the donation of \$444.00 to assist each organization with their ongoing COVID-19 relief efforts that support and assist women-identified and non-binary Hamiltonians.

With respect to motion 4.4; As Care-mongering and Keeping Six are not stand-alone registered charities, A. Getty, on behalf of the Status of Women Committee, as the Chair, will confirm who the flow through charities for these two programs prior to providing the donation of \$444.00.

Update: With respect to Motion 4.4, A. Getty has confirmed that The Aids Network is the flow through charitable organization for Keeping Six and The Hamilton Centre for Civic Inclusion is the flow through Charity for Care-mongering. Therefore;

- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to The Aids Network: Care-Mongering program to support ongoing community activities during COVID
- An allocation of \$444.00 from the Status of Women Committee's 2020 budget to be provided to The Hamilton Centre for Civic Inclusion: Keeping 6 program to support ongoing community activities during COVID



CITY OF HAMILTON

City Manager's Office Digital Innovations & Strategic Partnerships

TO:	Chair and Members Audit, Finance and Administration Committee
COMMITTEE DATE:	February 4, 2021
SUBJECT/REPORT NO:	2020 City Enrichment Fund Update (GRA21001) (City Wide)
WARD(S) AFFECTED:	City Wide
PREPARED BY:	Monique Garwood 905-546-2424 ext.3991
SUBMITTED BY:	Cyrus Tehrani Director, Digital Innovations & Strategic Partnerships City Manager's Office
SIGNATURE:	

RECOMMENDATION

That the overall 2020 City Enrichment Fund surplus (attached as Appendix "A" to Report GRA21001), in the amount of \$732,342 be transferred to the City Enrichment Fund Reserve # 112230.

EXECUTIVE SUMMARY

The total budget for the 2020 City Enrichment Fund program is \$6,088,340. During the year, as many programs and events were impacted by COVID, a phased approach was used to allocate funding to applicants. In all, the funding recommendations were submitted in 4 phases which totalled \$5,434,052. As at the 2020 year-end, the City Enrichment Fund's budget shows an unspent balance of \$732,342.

The City Enrichment Fund Reserve # 112230 currently has a balance of \$101,594.

Alternatives for Consideration - See Page 2

FINANCIAL – STAFFING – LEGAL IMPLICATIONS (for recommendation(s) only)

Financial: The overall surplus in the City Enrichment Fund is \$732,342. If transferred to the City Enrichment Fund Reserve, the balance in this reserve will be \$833,936.

Staffing: N/A Legal: N/A

SUBJECT: 2020 City Enrichment Fund Update (GRA21001) (City Wide) Page 2 of 3

HISTORICAL BACKGROUND (Chronology of events)

Prior to the final year-end closing entries, staff prepare a final draft of the City Enrichment Fund actual costs, inorder to provide the Grants Sub-Committee and Council a review of any unclaimed or unallocated Grant Funds. This report outlines all the grants paid, due or allocated for the year ending December 31, 2020. This is consistent with the practice of previous years.

POLICY IMPLICATIONS AND LEGISLATED REQUIREMENTS

N/A

RELEVANT CONSULTATION

Financial Policy & Planning – Corporate Services

ANALYSIS AND RATIONALE FOR RECOMMENDATION

The unallocated balance for the City Enrichment Fund is \$732,342. The summary of funding is below:

2020 Approved Budget	\$6	,088,340
Grants Issued per Category:		
Community Services	\$2	,102,033
Agriculture	\$	78,658
Environment	\$	146,390
Sport & Active Lifestyle	\$	127,646
Arts	\$2	,696,275
Communities, Culture & Heritage (CCH)	\$	202,255
Administration Costs	\$	2,741
Total Payments	\$5	,355,998
Balance/Surplus	\$	732,342

ALTERNATIVES FOR CONSIDERATION

The Grants Sub-Committee could choose not to transfer the surplus to the City Enrichment Fund Reserve # 112230. The surplus would form part of the City's overall surplus/deficit for 2020.

SUBJECT: 2020 City Enrichment Fund Update (GRA21001) (City Wide) Page 3 of 3

ALIGNMENT TO THE 2016 - 2025 STRATEGIC PLAN

Community Engagement & Participation

Hamilton has an open, transparent and accessible approach to City government that engages with and empowers all citizens to be involved in their community.

Healthy and Safe Communities

Hamilton is a safe and supportive city where people are active, healthy, and have a high quality of life.

Culture and Diversity

Hamilton is a thriving, vibrant place for arts, culture, and heritage where diversity and inclusivity are embraced and celebrated.

APPENDICES AND SCHEDULES ATTACHED

Appendix "A" to GRA21001 – 2020 City Enrichment Fund Summary

2020 City Enrichment Fund SUMMARY

Category	Total Applications	2020 Approved Budget		2020 Total Approved Funding *		2020 Total Payment		Variance (Budget vs Payment)	
Community Services Total	107	\$	2,164,360	\$	2,102,033	\$	2,102,033	\$	62,327
Agriculture Total	18	\$	143,360	\$	78,658	\$	78,658	\$	64,702
Environment Total	14	\$	146,390	\$	146,390	\$	146,390	\$	-
Sport and Active Lifestyles Total	44	\$	245,990	\$	143,055	\$	127,646	\$	118,344
Arts Total	82	\$	2,770,540	\$	2,701,910	\$	2,696,275	\$	74,265
CCH Total	69	\$	567,700	\$	212,017	\$	202,255	\$	365,445
Total Traditional Grant Categories	334	\$	6,038,340	\$	5,384,063	\$	5,353,257	\$	685,083
CEF Administration		\$	50,000	\$	50,000	\$	2,741	\$	47,259
Total City Enrichment Fund	334	\$	6,088,340	\$	5,434,063	\$	5,355,998	\$	732,342

* Funds were approved in 4 phases:

Phase 1	2,541,713
Phase 2	2,640,836
Phase 3	84,663
Phase 4	166,851
	5,434,063



CITIZEN COMMITTEE REPORT

То:	Audit, Finance and Administration Committee; General Issues Committee; and City Council		
From:	LGBTQ Advisory Committee Cameron Kroetsch, Chair		
Doto	January 10, 2021		
Date:	January 19, 2021		
Re:	All Advisory Committee Event		

Recommendation

That the LGBTQ Advisory Committee (LGBTQAC) host a virtual all Advisory Committee event to be held in 2021 with all other City of Hamilton Advisory Committees (there are 14 in total according to the City of Hamilton's website).

Background

This proposed event will be similar to December events that have been held in the past by the Committee (but will be virtual due to the COVID restrictions that are currently in place).

This event is intended to restart the tradition of ACs coming together informally (i.e. not using a formal meeting structure). In the past, it was a regular occurrence for Chairs, Vice Chairs, and other members of ACs to hold informal events to gather, discuss best practices, and offer support to one another.

If approved by City Council, each AC would be sent information from the LGBTQAC as an item of correspondence for its next available meeting. The Chair of the LGBTQAC and its Staff Liaison will coordinate with other ACs and their Staff Liaisons to select a

date that will work for the other ACs and then send an invitation to all ACs to invite them to attend.

As an event, and not a formal public meeting, there will be no requirement to host the event through Webex, so it may be hosted using another virtual meeting software. The event will not be conducting or advancing any official City business.

Analysis / Rationale

The purpose is for each AC to have the opportunity to introduce itself and to educate the other ACs about what it does. As part of this presentation, each AC will talk about the successes it has had and the challenges it has faced. There will be an open discussion after these presentations to allow for comments, suggestions, feedback, and support.

The Chair of the LGBTQAC delegated to 2 other ACs in 2020 (Arts Advisory Commission and the Hamilton Cycling Committee) and made a brief presentation to them about holding this event. Members of both ACs were very supportive of attending the event as described in this report.

The Chair of the LGBTQAC and its Staff Liaison will co-facilitate the event.

NOTE - We ask that the LGBTQAC Chair be updated as to when and at what Committee the recommendation(s) in this Citizen Committee Report will be addressed so that they can speak to the matter formally at that time.



GOVERNANCE REVIEW SUB-COMMITTEE REPORT 21-001

Monday, January 25, 2021 9:30 am Council Chambers Hamilton City Hall

Present:

Councillors T. Whitehead (Chair), M. Wilson (Vice-Chair), B. Clark, L.

Ferguson, M. Pearson, and A. VanderBeek

THE GOVERNANCE REVIEW SUB-COMMITTEE PRESENTS REPORT 21-001 AND RESPECTFULLY RECOMMENDS:

1. 2020 Review of the City's Procedural By-law Amendments (FCS21004 / LS21001) (Item 9.1)

- (a) That the revised Summary of the Proposed Revisions, as detailed in Appendix 'B' attached to Governance Review Sub-Committee Report 21-001, be approved;
- (b) That By-law 18-270, the Council Procedural By-law, and it's amending By-laws 19-090, 19-212,19-308, 20-042, 20-055, 20-103, 20-129, 20-145, 20-146, 20-151, 20-184, be repealed; and,
- (c) That the Council Procedural By-law attached as Appendix 'A' to Governance Review Sub-Committee Report 21-001, as amended, be enacted by Council.

2. Civil Marriage Solemnization Update (CL19012(b)) (City Wide) (Item 9.2)

That Report CL19012(b) respecting Civil Marriage Solemnization Update, be received.

3. Legal Advice on Contract Terms (Item 13.1)

That the direction provided to staff in Closed Session respecting Legal Advice on Contract Terms, be approved and remain confidential.

FOR INFORMATION:

(a) CHANGES TO THE AGENDA (Item 1)

The Committee Clerk advised of the following change to the agenda:

CHANGE TO THE ORDER OF ITEMS:

Item 13.1, respecting Legal Advice on Contract Terms, to be considered immediately following Item 9.2.

The January 25, 2021 Agenda of the Governance Review Sub-Committee was approved, as amended.

(b) DECLARATIONS OF INTEREST (Item 2)

There were no declarations of interest.

(c) APPROVAL OF MINUTES (Item 3)

(i) December 2, 2020 (Item 3.1)

The Minutes of the December 2, 2020 meeting of the Governance Review Sub-Committee were approved, as presented.

(d) COMMUNICATIONS (Item 4)

(i) Correspondence from Cameron Kroetsch, respecting suggested changes to consolidated Procedural By-law 18-270 (Item 4.1)

The correspondence from Cameron Kroetsch, respecting suggested changes to consolidated Procedural By-law 18-270 was received and referred to consideration of Item 9.1 - Review of the City's Procedural By-law Amendments (FCS21004 / LS21001).

(e) DELEGATION REQUESTS (Item 5)

(i) Cameron Kroetsch, respecting changes to the Procedural By-law (For today's meeting) (Item 5.1)

The delegation request from Cameron Kroetsch, respecting changes to the Procedural By-law, was approved for today's meeting.

(f) DELEGATIONS (Item 7)

(i) Cameron Kroetsch, respecting changes to the Procedural By-law (Added Item 7.1)

Cameron Kroetsch addressed the Committee respecting changes to the Procedural By-law, with the aid of a PowerPoint presentation.

The delegation from Cameron Kroetsch, respecting changes to the Procedural By-law, was received.

(g) DISCUSSION ITEMS (Item 9)

- (i) 2020 Review of the City's Procedural By-law Amendments (FCS21004 / LS21001) (Item 9.1)
 - (a) That the Summary of the Proposed Revisions, as detailed in Appendix 'B' attached to Report FSC21004 / LS21001, be approved;
 - (b) That By-law 18-270, the Council Procedural By-law, and it's amending By-laws 19-090, 19-212,19-308, 20-042, 20-055, 20-103, 20-129, 20-145, 20-146, 20-151, 20-184, be repealed; and,
 - (c) That the Council Procedural By-law attached as Appendix 'A' to Report FSC21004 / LS21001, be enacted by Council.

The Committee made the following amendments to the recommendations in Report FCS21004 / LS21001 and directed staff as follows:

- (i) That staff be directed to report back to the next Governance Review Sub-Committee meeting, on a further amendment to the Council Procedural By-law, to permit electronic participation by members of Council at Council and Standing Committee meetings beyond the Emergency Order currently in place.
- (ii) That Sections 3.3 (2) and 5.5 (2) Location of Meetings to Appendix "A" of Report FCS21004 / LS21001, be deleted in their entirety as follows:

SECTION 3 – COUNCIL MEETINGS

- 3.3 Location of Meetings
- (2) When necessary, Meetings of Council may be held in an adjacent municipality (Halton Region; Niagara Region or Haldimand County), with the details posted on the City's website.

<u>SECTION 5 – COMMITTEE MEETINGS</u>

- 5.5 Location of Meetings
- (2) When necessary, Standing Committee Meetings may be held in an adjacent municipality (Halton Region; Niagara Region or Haldimand County), with the details posted on the City's website.
- (iii) That staff be directed to review and provide the Audit, Finance and Administration Committee with clarification on the wording of Sections 3.10 (10) and 5.14 (10) Communications Items to Appendix "A" of Report FCS21004 / LS21001 on February 4, 2021.

Governance Review Sub-Committee Minutes 21-001

- (iv) (a) That Section 8.4 Order and Decorum of Appendix "A" to Report FCS21004 / LS21001, *be amended* by deleting the words "without challenge"; and,
 - (b) That staff be directed to provide the Audit, Finance and Administration Committee with alternative language on February 4, 2021.
 - 8.4 Should a Member of Council persist in conducting themselves in a manner contrary to the rules set forth in subsection <u>8</u>.1 after having been called to order by the Mayor or the Chair of the Committee, the Mayor or the Chair may order them *without challenge* to vacate the place the meeting is being held. If the Member of Council apologizes, they may, by vote of a majority of the Council or Committee members, be permitted to remain at the meeting.
- (v) That Section 5.9 (3) Rules of Procedure, of Appendix "A" to Report FCS21004 / LS21001, be amended by deleting the words "when a subject matter directly affects their Ward"; as follows:
 - (3) A Member of Council who is not a member of a Standing Committee is entitled to attend such Standing Committee meetings and to participate when a subject matter directly affects their Ward, provided that the Member of Council shall:
 - (a) not be counted for quorum purposes, and;
 - (b) not move any motion or vote on any matter.

(h) PRIVATE AND CONFIDENTIAL (Item 13)

That Committee move into Closed Session respecting Item 13.1, pursuant to Section 8.1, Sub-section (f) of the City's Procedural Bylaw 18-270, and Section 239(2), Sub-section (f) of the *Ontario Municipal Act, 2001*, as amended, as the subject matter pertains to advice that is subject to solicitor-client privilege, including communications necessary for that purpose.

(i) Legal Advice on Contract Terms (Item 13.1)

Staff were provided direction in Closed Session.

For disposition of this matter refer to Item 3.

(i) DISCUSSION ITEMS (Item 9) (Continued)

(ii) Integrity Commissioner Work Plan (FCS20016(b)) (City Wide) (Item 9.3)

The consideration of Report FCS20016(b), respecting the Integrity Commissioner Work Plan was deferred to the next Governance Review Sub-Committee meeting.

(j) GENERAL INFORMATION/OTHER BUSINESS (Item 12)

(i) Amendments to the Outstanding Business List (Item 12.1):

The following amendments to the Outstanding Business List, were approved:

(a) Items to be Removed (Item 12.1(a)):

Civil Marriage Solemnization (CL19012) (City Wide) (Item 10.2) That report CL19012 respecting Civil Marriage Solemnization be referred back to staff for public consultation on a 1 year Pilot Project, to a maximum of 150 ceremonies, using current resources. Added: November 26, 2019 at Governance Review Sub-Committee (Item 10.2)

Completed: February 12, 2020 at Governance Review Sub-

Committee (Item 10.1)
OBL Item: D-2019

Civil Marriage Solemnization (CL19012(a)) (City Wide) (Item 10.1) That staff report back to the Governance Review Sub-Committee within a one year period to provide an update on Civil Marriage Solemnization services.

Added: February 12, 2020 at Governance Review Sub-Committee (Item 10.1)

Completed: January 25, 2021 at Governance Review Sub-

Committee (Item 9.2) OBL Item: 2020-A

(k) ADJOURNMENT (Item 14)

There being no further business, the Governance Review Sub-Committee meeting adjourned at 1:18 p.m.

Respectfully submitted,

Councillor T. Whitehead, Chair Governance Review Sub-Committee

Angela McRae Legislative Coordinator Office of the City Clerk

Page 101 of 261

Appendix A to Item 1 of Governance Review Sub-Committee Report 21-001 as amended by Governance Review Sub-Committee on January 25, 2021

Authority:	Item, Audit, Finance & Administration Committee Report 21-0 (Governance Review Sub Committee Report 21) Date:, 2021
	Bill No.

CITY OF HAMILTON

BY-LAW NO. 21-

A BY-LAW TO GOVERN THE PROCEEDINGS OF COUNCIL AND COMMITTEES OF COUNCIL

WHEREAS Section 238 of the *Municipal Act, 2001* requires that Council of the City of Hamilton adopt, by By-law, procedures governing the calling, place, and proceedings of meetings; and

WHEREAS, the Council of the City of Hamilton, in adopting Item _____ of Audit, Finance & Administration Committee Report 21-____ on ______, 2021 recommended that Bylaws 18-270, 19-090, 19-212,19-308, 20-042, 20-055, 20-103, 20-129, 20-145, 20-146, 20-151, 20-184 be repealed and replaced;

NOW THEREFORE, BE IT RESOLVED, the Council of the City of Hamilton enacts as follows:

INDEX

TITLE Definitions	SECTION 1	PAGE 3
Application	2	7
Council Meetings	3	7
First Council Meeting	3.1	7
Regular Council Meeting Times	3.2	7
Location of Meetings	3.3	8
Special Meetings of Council	3.4	8
Meetings of Council for Emergencies	3.5	9
Cancellation of a Council Meeting	3.6	9
Quorum	3.7	10
Remedy for Lack of Quorum	3.8	10
Delegations	3.9	11
Communication Items	3.10	11
Order of Business	3.11	12
Bills and By-laws	3.12	12
Editorial and Other Changes	3.13	13
Shareholder and Board of Director Meetings	3.14	14
Conflict of Interest Declarations	3.15	14
Provisional Governance Committee During a Declared Emergency	3.16	14

Page 102 of 261
Page 2 of 66

Procedural By-law

Occupall Occupanientian Hadatas	0.47	4.5
Council Communication Updates	3.17	15
Committee of the Whole	4	15 15
Council may, by resolution:	4.1 -	15
Standing Committee Meetings	5	16
Standing Committees	5.1	16
Appointment of Standing Committee Chairs and		
Vice Chairs	5.2	16
Standing Committee Membership	5.3	16
Quorum	5.4	17
Location of Meetings	5.5	18
Standing Committee Meeting Times	5.6	18
Cancellation of a Committee Meeting	5.7	19
Committee Reports	5.8	19
Rules of Procedure	5.9	20
Duties of Standing Committees	5.10	20
Order of Business	5.11	20
Delegations	5.12	21
Public Hearings at Standing Committees	5.13	22
Communication Items	5.14	23
Reconsideration of a Matter Decided Within a	5.15	24
Standing Committee Meeting		
Conflict of Interest Declarations	5.16	25
Special Meetings of a Standing Committee	5.17	25
Change to a Schedules Committee Meeting	5.18	26
Rules of Debate, Voting Procedures and Motion	6	26
Procedures		
Rules of Debate	6.1	26
Relinquishing the Chair	6.1(9)	27
Voting Procedures	6.2	27
Motion Procedures	6.3	29
Notice of Motion	6.3(1)	29
Dispensing with Notice	6.3(2)	30
Motion to Withdraw	6.3(3)	30
Motion to Amend	6.3(4)	30
Motion to Defer	6.3(5)	30
Motion to Refer	6.3(6)	31
Motion to Adjourn	6.3(7)	31
Motion to Call the Question	6.3(8)	31
Motion to Reconsider a Matter Decided by Council	6.3(9)	31
Order of Procedural Motions	6.3(10)	32
Non-Debatable Motions	6.3(11)	33
Motion to Suspend the Rules of Procedure	6.3(12)	33
Motion Beyond Jurisdiction of Council	6.3(13)	33
Establishment of Advisory Committees or Task	7	33
Forces		
Order and Decorum	8	33
Meetings Closed to the Public	9	35

1 Toocdara By law		Page 3 of 66
Virtual Participation at Council and/or Committee Meetings	10	36
Virtual Participation During an Emergency	10.1	36
Virtual Meetings during an Emergency	11	36
Role of Clerk	12	37
General	13	37
Appendices		
General Issues Committee	Appendix A	39
Board of Health	Appendix B	41
Public Works Committee	Appendix C	42
Planning Committee	Appendix D	43
Emergency & Community Services Committee	Appendix E	44
Audit, Finance & Administration Committee	Appendix F	45
Roles of Council, Mayor and Committee Chair	Appendix G	54
Criteria for the Creation of Advisory Committees or	Appendix H	57
Task Forces		
Hamilton Advisory Committee/Task Force Code of Conduct	Appendix I	58
Hearing Procedures for Development Charge	Appendix J	60

Procedural By-law

Page 103 of 261

SECTION 1 - DEFINITIONS

1.1 In this By-law,

Complaints – Section 20

[&]quot;Act" means the Municipal Act, 2001.

[&]quot;Amend" means to alter or vary the terms of a main motion without materially changing its purpose, and amendment has a corresponding meaning.

[&]quot;Approve" means to support or endorse the action, report or document.

[&]quot;Bill" means a proposed form of By-law.

[&]quot;By-law" means a Bill, in a form approved by Council, passed for the purpose of giving general effect to a previous decision or proceedings of Council.

[&]quot;Chair" means the member presiding at any meeting of Council or Committee.

[&]quot;Challenge" means to appeal a ruling of the Chair.

[&]quot;City" means the geographical area of the City of Hamilton or the municipal corporation as the context requires.

[&]quot;Clerk" means the City Clerk, or their designate.

[&]quot;Closed Meeting" means a meeting, or part of a meeting, which is closed to the public as permitted by the Act, also referred to as an "in-camera meeting".

- "Committee" means a Standing Committee, Sub-Committee, Selection Committee or an Advisory Committee or Task Force established by Council from time to time.
- "Committee of the Whole" means a Committee comprised of all Members of Council that directly reports to Council.
- "Confirming By-law" means a by-law to confirm and ratify the actions of Council.
- "Consent Items" means those items on an agenda of a Standing Committee; Sub-Committee or Advisory Committee that may be approved by Committee without debate.
- "Council" means the City Council.
- "Defer" means to delay consideration of a matter by Council or a Committee until a specified time or event.
- "Deputy Mayor" means a Member of Council who is appointed for a one month period from amongst members of Council on a rotation basis determined by the Clerk at the beginning of the term of Council:
 - (a) to assist the Mayor; and
 - (b) to act from time to time in the place of the Mayor, with:
 - (i) all the rights, powers and authority of the Mayor when the Mayor is absent from the City or the office of Mayor is vacant; and
 - (ii) subject to paragraph (i), those rights, powers and authority of the Mayor that:
 - the Mayor is unable to act for the time being due to illness, incapacity or other reason, including acting in the place of the Mayor at any Council or Committee meeting when the Mayor is absent; or,
 - 2. the Mayor refuses to act;
- "ELE" means Electronic meeting procedures and applies to Virtual and Hybrid meetings during an emergency and used only when the procedures for Electronic meetings differ from In-Person (INP) procedures.
- "Emergency" means a situation or impending situation that constitutes a danger of major proportions that could result in serious harm to persons or substantial damage to property and that is caused by the forces of nature, a disease or other health risk, an accident or an act whether intentional or not.
- "Ex-officio" means by virtue of the office or position and the Mayor is an ex officio member of all Standing Committees who, when attending a Standing Committee meeting, shall have full voting privileges and shall be counted for the purposes of the Standing Committee's quorum as such quorum is set out in subsection 5.4 of this By-law.

- "Friendly Amendment" means an amendment made to the motion under debate with the consent of the mover and seconder, and without the requirement for an amending motion to be made.
- "Hybrid" means a proceeding where some members participate virtually and some members participate in-person at Council and Committee meetings during an emergency.
- **"INP"** means In-Person meeting procedures and applies only when the procedures for In-Person meetings differs from Electronic (ELE) meeting procedures.
- "In-Person" means a proceeding where members participate In-Person at Council and Committee meetings.
- "Majority" for the purposes of voting, means more than half the total of members of Council or Committee present and not prohibited by statute from voting.
- "Mayor" means the Head of the Council and in this By-law means the Deputy Mayor when the Deputy Mayor is acting in place and stead of the Mayor.
- "Meeting" means any scheduled, special or other meeting of a Council, local board or of a committee of either of them, where, a quorum of members is present, and members discuss or otherwise deal with any matter in a way that materially advances the business or decision-making of the Council, local board or committee.
- "Member of Council" is a person who has been elected to City Council.
- "Motion" means a proposal by a member for the consideration of Council or a Committee included in the published agenda, that is moved by a member and seconded by another member.
- "Notice of Motion" means advanced, written notice, given by a member and received by the Clerk, advising Council or Committee that a Motion will be brought forward at a future Meeting of Council or Committee that may be included in the published agenda or in the addendum.
- "Point of Order" means a question raised by a member with respect to any departure from the Procedural By-law or in the practised conduct of Council or Committee business.
- "Point of Privilege" means a matter that affects the credibility, reputation, and dignity of a Member individually or that Council or Committee collectively hold, that ensure Council's or a Committee's ability to function freely; a need for assistance, to be excused for illness or a personal emergency.
- "Public Hearing" means a Meeting or portion of a Meeting held pursuant to the requirements of a statute, regulation or by-law to hold a public meeting to hear representations on a matter from any person who attends the public meeting.

- "Question" means an issue before a meeting on which a decision has to be made. A question cannot be debated, amended or voted on until it has been proposed as a motion.
- "Quorum" means the number of members required to be present at a meeting to validate the transactions of its business.
- "Receive" means to acknowledge the receipt of information and to include it in the official record.
- "Recess" means a short intermission in a meeting's proceedings, which does not close the meeting, and after which business will immediately be resumed at exactly the point where it was stopped.
- "Recorded Vote" means a written record of the name and vote of each member of Council or a Committee who votes on a question and of each member present who does not vote.
- "Refer" means to send a matter currently under consideration to an individual; a department; a Committee or Council for their consideration.
- "Rules of Procedure" means the rules and procedures set out in this By-law, for the calling, place and proceedings of the meetings of Council and its Committees.
- "Selection Committee" means a Committee established by Council, comprised entirely of Members of Council, to interview and report back to Council on the appointment of citizen representatives to agencies, boards and Committees, and reports directly to Council.
- "Senior Leadership Team" means the City Manager and General Managers appointed by Council and anyone designated by the City Manager.
- "Standing Committee" means a Committee established by Council, comprised entirely of Members of Council, to carry out duties on an ongoing basis, as specified by Council, and that reports directly to Council.
- "Time Sensitive" means useful for a specific amount of time only or has to be used or completed by a specific time.
- "Virtual" means a proceeding where members participate online or on a computer screen, rather than In-Person during Council and Committee meetings during an emergency.
- "Website" means the official City website.

SECTION 2 - APPLICATION

- 2.1 The rules of procedure shall be observed in all IN-PERSON / VIRTUAL / HYBRID proceedings of Council and shall be the rules for the order and dispatch of business in Council and unless specifically provided, with necessary modifications, apply to all Committees.
- 2.2 All matters relating to the IN-PERSON / VIRTUAL / HYBRID proceedings of Council and Committees for which rules have not been provided for in this By-law and its Appendices shall be decided, as far as is reasonably practicable, with reference to the parliamentary rules as contained in the most recent edition of *Bourinot's Rules of Order*, Geoffrey H. Stanford, Fourth Edition.

SECTION 3 - COUNCIL MEETINGS

3.1 First Council Meeting

- (1) The First Meeting of Council shall be held on the first Monday of December in an election year at 5:00 p.m.
- (2) The IN-PERSON / VIRTUAL / HYBRID Proceedings of the First Meeting of Council shall include:
 - (a) Ceremonial special address by guests;
 - (b) Filing of Declarations of Office in accordance with the provisions of the Act;
 - (c) Council appointments to Committees, boards and agencies; and
 - (d) Confirming By-law.

3.2 Regular Council Meeting Times

- (1) Unless otherwise decided by Council, Regular Council meetings shall be held:
 - in January:
 on the fourth Wednesday of the month, commencing at 9:30 a.m.
 unless such day is a public or civic holiday, in which case Council
 shall set an alternate day and time;
 - (b) in February, April, May, June, September, October and November: every second and fourth Wednesday of each month, commencing at 9:30 a.m., unless such day is a public or civic holiday, in which case Council shall set an alternate day and time;
 - (c) in March:
 on the fourth Wednesday of the month, commencing at 9:30 a.m.
 unless such day is a public or civic holiday, in which case Council shall set an alternate day and time;
 - in July and August:on a Friday of each month (one meeting per month), commencing at 9:30 a.m., unless such a day is a public or civic holiday, in which case Council shall set an alternate day and time;

- (e) in December:
 on the second Wednesday of the month, commencing a 9:30 a.m.,
 unless such a day is a public or civic holiday, in which case Council
 shall set an alternative day and time; or
- (f) in accordance with the schedule approved by Council.
- (2) That, from time to time, there may be a need for Special Council Meeting at 5:00 p.m. to accommodate a Ceremonial Activity where the individuals being recognized are unable to attend a meeting during the day; and
- (3) Advance notice of meetings for the public's information may be found by accessing the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

3.3 Location of Meetings

(1) IN-PERSON / VIRTUAL / HYBRID Meetings of Council shall be held in the Council Chambers at City Hall in the City of Hamilton or at such other place within the City of Hamilton as Council may from time to time determine.

3.4 Special Meetings of Council

- (1) In addition to scheduled Council meetings, the Mayor may, at any time, summon a special meeting of Council by giving written direction to the Clerk stating the date, time, and purpose of the special meeting.
- (2) The Clerk shall summon a special meeting of Council when requested to do so in writing by a majority of Members of Council.
- (3) The Clerk shall give each Member of Council, or their designated staff, notice of a special meeting of Council at least 2 days before the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (4) The written notice to be given under subsection 3.4(3) shall state the nature of the business to be considered at the special meeting of Council and no business other than that which is stated in the notice shall be considered at such meeting.

- (5) Notwithstanding subsection 3.4(4), other business may be considered at a special meeting with at least two-thirds of the Members of Council present and voting in the affirmative.
- (6) The Clerk shall prepare the Order of Business for a special meeting of Council with the appropriate selection of agenda categories from subsection 3.11.

3.5 Meetings of Council for Emergencies

- (1) Despite any other provisions of this By-law, a meeting of Council for an emergency may be called by the Mayor, without written notice, to deal with an emergency, provided that an attempt has been made by the Clerk to notify Members of Council about the meeting as soon as possible and in the most expedient manner available.
- (2) The only business to be dealt with at a meeting of Council for an emergency shall be business dealing directly with the emergency.
- (3) The lack of receipt of a notice of, or an agenda for, a meeting of Council for an emergency by any Member of Council shall not affect the validity of the meeting or any action taken at such meeting.
- (4) The Clerk shall prepare the Order of Business for an emergency meeting of Council with the appropriate selection of agenda categories from subsection 3.11.
- (5) In the case of an emergency, Council may hold its meetings at any convenient location within or outside the municipality, and such meetings may be held as IN-PERSON / VIRTUAL / HYBRID meetings, with the details posted on the City's website.

3.6 Cancellation of a Council Meeting

- (1) The Clerk may cancel a Council meeting in consultation with the Mayor or the Deputy Mayor in the Mayor's absence.
- (2) The Clerk shall give each Member of Council, or their designated staff, notice of a cancellation of a meeting of Council at least 2 days prior to the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (3) The written notice to be given under subsection 3.6(2) shall state the reason for the cancellation of the meeting.

(4) The Clerk shall ensure that the meeting cancellation information is posted to the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

3.7 Quorum

- (1) As soon as there is a quorum after the time set for the meeting, the Mayor shall take the chair and call the meeting to order.
- (2) A majority of Members of Council is necessary to constitute a quorum of Council.
- (3) If the Mayor does not attend within 15 minutes of the time appointed for a Council meeting and a quorum of Council is present, the Deputy Mayor shall take the chair and call the meeting to order and will preside until the arrival of the Mayor.
- (4) If a quorum is not achieved within 30 minutes after the time appointed for a meeting of Council, the Clerk shall record the names of the Members of Council present and the meeting shall stand adjourned until the date of the next scheduled Council meeting or other meeting called in accordance with the provisions of this By-law.
- (5) **INP** If Quorum cannot be maintained during an IN-PERSON Council meeting, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the decision to continue the meeting will rest with the Mayor or Deputy Mayor. During the absence of quorum, no decisions may be approved.
 - ELE If Quorum cannot be maintained during a VIRTUAL / HYBRID Council meeting, due to a loss of members' electronic connection, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the meeting will be recessed for up to 15 minutes to allow for the member's electronic connection to be restored. If quorum cannot be restored within 15 minutes the meeting will be recessed to the next available business day. The Clerk will ensure that details regarding the time of the continuation of the Council meeting is on the City's website. Where Quorum during a VIRTUAL/HYBRID Council meeting is lost for any other reason the INP procedure shall apply.

3.8 Remedy for Lack of Quorum

Municipal Conflict of Interest Act

Where the number of Members of Council who, by reason of provisions of the *Municipal Conflict of Interest Act*, are disabled from participating in a meeting is such that at that meeting the remaining Members of Council are not of sufficient number to constitute a quorum, then, the remaining number of Members of Council shall be deemed to constitute a quorum, provided such number is not less than 2.

3.9 Delegations

Delegations are not permitted at meetings of Council.

3.10 Communication Items

- (1) Communication Items shall be addressed to the Mayor and Members of Council or the City Clerk and shall be presented to Council for consideration.
- (2) Communication Items shall be forwarded to the Office of the City Clerk no later than 12 Noon on the Wednesday of the week prior to the City Council meeting for inclusion in the printed Council Agenda.
- (3) Communication Items not received for inclusion in the Agenda package, shall be introduced under Changes to the Council Agenda, provided that they are received in the Office of the City Clerk no later than 12 Noon on the business day prior to the Council meeting.
- (4) Communication Items shall not be anonymous or contain any defamatory, obscene, offensive, indecent, improper, rude or vulgar language.
- (5) Communication Items shall not contain the names of individuals (staff or public) with the exception of the author's name. Any names found in the communication will be redacted pursuant to the obligations of *Municipal Freedom of Information and Protection of Privacy Act* (MFIPPA), this Procedural By-law and other relevant legislation or procedure.
- (6) Debate on Council communications shall be limited to the recommended disposition, or limited to reason for or against the referral of the matter to an appropriate Standing Committee.
- (7) Resolutions from other municipalities requesting support will be included on the Council Agenda to be received, unless otherwise directed by a Member of Council.
- (8) Receipt of resolutions from other municipalities by City Council shall not be construed as endorsement by the City of any recommendations or actions contained therein.
- (9) Communications respecting labour relation matters, conduct of a member(s) of staff and matters before another body (i.e. Police Services Board) will not be placed on the Council Agenda and will be forwarded forthwith by the City Clerk to the relevant City Department or body for consideration and appropriate action.
- (10) Communications respecting the conduct of a member(s) of Council or any member of a body subject to review by the Integrity Commissioner will not be placed on a Council Agenda and the author will be advised of the process for filing a complaint with the Integrity Commissioner.

(11) Notwithstanding subsections 3.10(9) and 3.10(10) Council may add any communication item to a Council Agenda.

3.11 Order of Business

The Order of Business for scheduled IN-PERSON / VIRTUAL / HYBRID meetings of Council, unless amended by Council in the course of the meeting, shall be as follows:

- (a) Approval of Agenda
- (b) Declarations of Interest
- (c) Ceremonial Activities (**ELE** During a VIRTUAL/HYBRID meeting ceremonial activities shall only be permitted to take place Virtually)
- (d) Approval of Minutes of Previous Meeting
- (e) Communications
- (f) Committee Reports
- (g) Motions
- (h) Notice of Motions
- (i) Statement by Members (non-debatable)
- (j) Council Communication Updates
- (k) Private and Confidential
- (I) By-laws and Confirming By-law
- (m) Adjournment

3.12 Bills and By-laws

- (1) All Bills coming before Council shall be in typewritten form and shall contain no blanks except as may be required to conform to accepted procedure or to comply with provisions of any statute or regulation and shall be complete with the exception of the number and date.
- (2) All Bills coming before Council shall receive one reading, unless otherwise requiring more than one reading, pursuant to a Provincial or Federal statute or regulation.
- (3) Every Bill shall be introduced upon motion by a Member of Council, and any number of Bills may be introduced together in one motion, but Council may at the request of a Member of Council, deal separately with any Bill. If Council deals separately with a Bill:
 - (a) all amendments or debate on the content of the Bill shall be held in committee of the whole:
 - (b) Council will move into committee of the whole for discussion on any Bill following a successful verbal motion to go into committee of the whole;

- (c) all amendments made in committee of the whole shall be reported by the Mayor to Council, which shall receive the same forthwith, and after the report has been received, a vote on the motion to pass the Bill shall be taken:
- (d) every By-law passed by Council shall be numbered and dated and shall be sealed with the seal of the municipality and signed by the Mayor and Clerk.
- (4) No Bill or By-law may be defeated on the floor of Council without reconsideration of the original motion which gave rise to the consideration of the Bill or By-law.

3.13 Editorial and Other Changes

- (1) The Clerk may make changes to By-laws or resolutions to:
 - (a) Correct spelling, punctuation or grammatical errors, or errors that are of a clerical, typographical or similar nature.
 - (b) Alter the style or presentation of text or graphics to improve electronic or print presentation.
 - (c) Replace a description of a date or time with the actual date or time.
 - (d) If a provision provides that it is contingent on the occurrence of a future event and the event occurs, remove text referring to the contingency and make any other changes that are required as a result.
 - (e) When the name, title, location or address of a body, office, person, place or thing has been altered, change any reference to the name, title, location or address to reflect any alteration in name, title, location or address.
 - (f) Correct errors in the numbering of provisions or other portions of a By-law and make any changes in cross-references that are required as a result.
 - (g) If a provision of a transitional nature is contained in a By-law, make any changes that are required as a result.
 - (h) Make a correction, if it is patent both that an error has been made and what the correction should be.

(2) The Clerk:

- (a) may provide notice of the changes made under subsections 3.13(1)(a) to 3.13(1)(c) inclusive in the manner that they consider appropriate.
- (b) shall provide notice of the changes made under subsections 3.13(1)(d) to 3.13(1)(h) inclusive in the manner that they consider appropriate.
- (c) in determining whether to provide notice under subsection 3.13(2)(a), shall consider:
 - (i) the nature of the change; and
 - (ii) the extent to which notice, and the information provided in it, would provide assistance in understanding the relevant legislative history.

- (d) in providing notice under subsection 3.13(2)(a) or 3.13(2)(b), shall state the change or the nature of the change.
- (3) No legal significance shall be inferred from the timing of the exercise of a power under this section.
- (4) Regardless of when a change is made to a By-law under this section, the change may be read into the By-law as of the date it was enacted if it is appropriate to do so.

3.14 Shareholder and Board of Director Meetings

Where Council acts as the Shareholder or the Board of Directors, or both, of a corporation:

- (a) Council shall meet as the Shareholder or the Board of Directors as necessary to complete such business as would normally be completed by the Shareholder or Board of Directors and as required by the corporation's by-laws and resolutions and any laws that govern the corporation; and
- (b) the Mayor and Clerk are authorized, upon approval by Council, to sign any necessary resolutions.

3.15 Conflict of Interest Declarations

- (1) Where a member has disclosed a conflict of interest at a meeting, the member, shall file a written statement with the Clerk.
- (2) Where a member, has disclosed an interest and filed a written statement with the Clerk, as required by subsection 3.15(1), the Clerk, shall:
 - (a) file the written statement by the member in the Conflict of Interest Registry;
 - (b) record the interest in the minutes of the meeting:
 - (c) file a record of the interest recorded in the minutes of the meeting in the Conflict of Interest Registry; and
 - (d) make the Conflict of Interest Registry available for public inspection.

3.16 Provisional Governance Committee During a Declared Emergency

- (1) The Provisional Governance Committee may carry out the duties and responsibilities of the Council only at such times when, to the satisfaction of at least 5 Members of Council attending a properly scheduled meeting of Council, at least 8 Members of Council are not able to attend such meeting due to death, illness or circumstances attributable to an emergency which has been declared by the Mayor or their designate in accordance with the City's Emergency Plan.
- (2) The Provisional Governance Committee shall be comprised of eight Members of Council and shall elect its own Chair from amongst the attending Members.

- (3) The Provisional Governance Committee shall have all the powers and duties that may legally be delegated to the Provisional Governance Committee pursuant to the Act, or any other applicable Provincial or Federal legislation, subject to the following conditions and restrictions:
 - (a) the delegation shall be effective only during the times specified; and
 - (b) the Provisional Governance Committee shall conduct its meetings in accordance with this By-law, providing that the quorum for the constitution of a valid meeting of the Provisional Governance Committee shall be 5 Members of Council.

3.17 Council Communication Updates

- (1) All staff Communication Update documents sent to Council and posted on the website will be published by the Clerk on the Council agenda for receipt into the public record.
- (2) A member may refer any of the items listed within the Council Communication Updates to a Standing Committee by contacting the Clerk and any item so referred shall be placed on the next available agenda of the respective Standing Committee.

SECTION 4 - COMMITTEE OF THE WHOLE

- **4.1** Council may, by resolution:
 - (a) move into Committee of the Whole to consider, including debate, Standing Committee Reports; Selection Committee Reports and the Reports of Committees who report directly to Council on the Council meeting agenda; and
 - (b) move out of Committee of the Whole to report to Council.
- **4.2** During the consideration of Reports a member may lift an item from the Information Section with the approval of a majority of the members present, for the matter:
 - (a) to be voted upon by Council; or
 - (b) to be debated and/or amended by Council.
- 4.3 An information item that is lifted from a Report and amended, will be given an item number and added to the appropriate section of the Standing Committee Report, Selection Committee Report, or the Report of Committee who reports directly to Council.

<u>SECTION 5 – COMMITTEE MEETINGS</u>

5.1 Standing Committees

The Standing Committees that report directly to Council shall be as follows:

- (a) General Issues Committee
- (b) Board of Health
- (c) Public Works Committee
- (d) Planning Committee
- (e) Emergency & Community Services Committee
- (f) Audit, Finance & Administration Committee

5.2 Appointment of Committee Chairs and Vice Chairs

- (1) Each Standing Committee shall recommend to Council the appointment of a Chair, Vice Chair and, where applicable, 2nd Vice Chair for the required term.
 - (a) The term of the Chair, Vice Chair and 2nd Vice Chair shall be for a calendar year, or part of a calendar year in an election year.
 - (b) No Member of Council may serve as Chair, Vice Chair or 2nd Vice Chair of the same Standing Committee for more than one year in a Council term.
 - (c) The role of the Chair of the General Issues Committee shall be rotated amongst the Deputy Mayors.
- (2) Despite subsection 5.2(1)(a) and (b), the Mayor shall stand as Chair for the term of Council for the Board of Health.
- (3) The Board of Health shall recommend to Council the appointment of a Vice Chair in accordance with 5.2(1). Despite subsection 5.2(1)(a) and (b), the Vice Chair shall stand as Vice Chair for the term of Council for the Board of Health.
- (4) Each Sub-Committee, Advisory Committee and Task Force shall recommend to Council the appointment of a Chair and Vice Chair. The Chair or Vice Chair of a Sub-Committee, Advisory Committee or Task Force established by Council may serve for more than one year in a Council term.

5.3 Standing Committee Membership

- (1) General Issues Committee and Board of Health shall be comprised of all Members of Council.
- (2) Public Works Committee shall be comprised of a minimum of 8 Members of Council.
- (3) Planning Committee shall be comprised of a minimum of 7 Members of Council.
- (4) Emergency & Community Services Committee shall be comprised of a minimum of 5 Members of Council.

- (5) Audit, Finance & Administration Committee shall be comprised of a minimum of 7 members of Council.
- (6) Each Member of Council shall sit on a minimum of 2 Standing Committees, in addition to the Board of Health and General Issues Committee.
- (7) Membership on Standing Committees shall be for the duration of the term of Council.
- (8) The Mayor is an ex-officio member of all Standing Committees who, when attending a Standing Committee meeting, shall have full voting privileges and shall be counted for the purpose of the Standing Committee's quorum as set out in subsection 5.4.

5.4 Quorum

- (1) The quorum for all Committees shall be a half of the membership rounded up to the nearest whole number.
- (2) As soon as there is a quorum after the time set for the meeting, the Chair shall take the chair and call the meeting to order.
- (3) If the Chair does not attend within 15 minutes of the time appointed for a Committee meeting and a quorum of Committee is present, the Vice Chair shall take the chair and call the meeting to order and will preside until the arrival of the Chair.
- (4) If a quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of Committee, the Clerk shall record the names of the Members of Council present and the meeting shall stand adjourned until the date of the next scheduled Committee meeting or another meeting may be called in accordance with the provisions of this Bylaw, except as provided for in 5.4(6) and 5.4(7).
- (5) **INP** If Quorum cannot be maintained during an IN-PERSON Committee meeting, the Clerk will advise the Chair that quorum is lost and the decision to continue the meeting will rest with the Chair. During the absence of quorum, no decisions may be approved.
 - ELE

 If Quorum cannot be maintained during a VIRTUAL / HYBRID Standing Committee meeting, due to a loss of members' electronic connection, the Clerk will advise the Chair that quorum is lost and the meeting will be recessed for up to 15 minutes to allow for the member's electronic connection to be restored. If quorum cannot be restored within 15 minutes the meeting will be recessed to the next available business day. The Clerk will ensure that details regarding the time of the continuation of the Standing Committee meeting is on the City's website. Where Quorum during a VIRTUAL/HYBRID Standing Committee meeting is lost for any other reason the INP procedure shall apply.

- (6) If quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of Committee, the Chair may decide to proceed informally to hear scheduled presentations. Ratification of any decisions will be postponed to a future meeting.
- (7) If quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of a Committee, the Chair may allow the Committee to discuss items of interest. Ratification of any decisions will be postponed to a future meeting.

5.5 Location of Meetings

(1) IN-PERSON / VIRTUAL / HYBRID Standing Committee Meetings shall be held in the Council Chambers at City Hall in the City of Hamilton or at such other place within the City of Hamilton as Council may from time to time determine.

5.6 Standing Committee Meeting Times

- (1) Unless otherwise decided by Council, scheduled meetings of the Standing Committees will be as follows:
 - (a) (i) General Issues Committee shall meet at least once per month on a Wednesday commencing at 9:30 a.m.;
 - (ii) Board of Health shall meet at least once per month commencing at 9:30 a.m.;
 - (iii) Public Works Committee shall meet at least once per month commencing at 1:30 p.m.;
 - (iv) Planning Committee shall meet at least once per month on a Tuesday commencing at 9:30 a.m.;
 - (v) Audit, Finance & Administration Committee shall meet at least once per month on a Thursday commencing at 9:30 a.m.;
 - (vi) Emergency & Community Services Committee shall meet at least once per month on a Thursday commencing at 1:30 p.m.;
 - (b) In the event a meeting date, as shown in subsections 5.6(1)(a)(i) to 5.6(1)(a)(vi) falls on a day which is a public or civic holiday, an alternate date and time will be selected:
 - (c) In accordance with the schedule approved by Council.

- (2) In addition to scheduled Standing Committee meetings, a special meeting of a Standing Committee may be scheduled when required, at the call of the Chair, at locations and times to permit convenient access for members of the public most affected by such a matter with at least 2 days notice, delivering notice in accordance with subsections 5.17(3) and 5.17(4).
- (3) Advance notice of Committee meetings for the public's information may be found by accessing the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca .

5.7 Cancellation of a Committee Meeting

- (1) The Clerk may cancel a Committee meeting in consultation with the Committee Chair or the Committee Vice Chair in the Committee Chair's absence.
- (2) The Clerk shall give each Member of the Committee, or their designated staff, notice of a cancellation of a Committee meeting at least 2 days prior to the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (3) The written notice to be given under subsection 5.7(2) shall state the reason for the cancellation of the meeting.
- (4) The Clerk shall ensure that the meeting cancellation information is posted to the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

5.8 Committee Reports

- (1) Each Standing Committee, the Selection Committee, and other Committees that report directly to Council shall make recommendations to Council.
- (2) A minimum of 2 days shall pass before the Report of a Standing Committee, the Selection Committee, or other Committee that reports directly to Council is presented to Council to provide adequate opportunity for review. Every effort will be made for Standing Committee and/or Selection Committee Reports to be released at an earlier date.
- (3) Despite subsection 5.8(2), this rule may be suspended by the affirmative vote of two-thirds of the Members of Council present and voting.

5.9 Rules of Procedure

- (1) No member of Council shall speak more than once, until every member of a Standing Committee, followed by every Member of Council who is not a member of the Standing Committee, has spoken, and then only to provide information or seek clarification from the Chair, other Members of Council or staff.
- (2) A Member of Council may place an item on a future agenda of a Standing Committee meeting by delivering a Notice of Motion as set out in subsection 6.3(1) to the Chair of the Standing Committee and to the Committee Legislative Coordinator.
- (3) A Member of Council who is not a member of a Standing Committee is entitled to attend such Standing Committee meetings and to participate, provided that the Member of Council shall:
 - (a) not be counted for quorum purposes, and;
 - (b) not move any motion or vote on any matter.

5.10 Duties of Standing Committees

The duties of the Standing Committees shall be as set out in Appendices A to F inclusive.

5.11 Order of Business

- (1) The general Order of Business for the IN-PERSON / VIRTUAL / HYBRID scheduled meetings of Committees, unless amended by the Committee in the course of the meeting, shall be as follows:
 - (a) Ceremonial Activities (**ELE** During a VIRTUAL/HYBRID meeting ceremonial activities shall only be permitted to take place Virtually)
 - (b) Approval of Agenda
 - (c) Declarations of Interest
 - (d) Approval of Minutes of Previous Meeting
 - (e) Communications
 - (f) Delegation Requests
 - (g) Consent Items
 - (h) Staff Presentations
 - (i) Public Hearings/ Delegations
 - (i) Discussion Items
 - (k) Motions
 - (I) Notice of Motions
 - (m) General Information/Other Business

- (n) Private and Confidential
- (o) Adjournment
- (2) The Clerk may prepare the Order of Business for the scheduled meetings of Committees with slight adjustments to the general Order of Business when deemed necessary.

5.12 Delegations

- (1) **INP** Persons who wish to appear as a delegation at an IN-PERSON meeting will appear in-person to address a Committee
 - Persons who wish to appear as a delegation at a VIRTUAL/
 HYBRID meeting will appear virtually to address a Committee
 - (a) Persons wishing to delegate on a matter that is not listed on a Committee agenda, shall make a request in writing to the Clerk and the Clerk shall list the delegation request on the respective Committee's upcoming agenda. The requester will be notified of the date of the meeting where their delegation will be heard following Council's ratification of the Committee's Report.
 - (b) Persons wishing to delegate on a matter that is listed on the Committee agenda, shall make a request in writing to be listed as a delegation to the Clerk no later than 12:00 noon the business day before the meeting. If the Committee is meeting on a Monday, the deadline will be 12:00 noon on the Friday before.
 - (c) All delegates (appearing VIRTUALLY or IN-PERSON) shall complete the "Request to Speak to a Committee of Council" form (available on the City's website at https://www.hamilton.ca/council-committee-council-committee-council.
- (2) A request to attend IN-PERSON or VIRTUALLY as a delegation shall include the person's name, reason(s) for the delegation, and if applicable the name, address and telephone number of any person, corporations or organizations which they represent.
- (3) Upon receipt of a request to attend IN-PERSON or VIRTUALLY as a delegation, the Clerk shall list the delegation request on the next appropriate agenda for the relevant Committee and the decision whether or not to entertain the delegation will be made by the Committee.
- (4) When a person is listed as a delegation on a Committee agenda and such person, without notification to the Clerk, does not attend said meeting, that person must make a new request to be considered as a delegation and must re-apply in accordance with subsections 5.12(1) and 5.12(2).
- (5) An IN-PERSON or VIRTUAL delegation of one or more persons, shall be limited to an oral presentation of not more than five minutes.

- (6) Notwithstanding subsection 5.12(5), an extension of a specific allotment of time to the five-minute restriction may be granted with the approval of a majority of the members of the Committee present.
- (7) Delegations are encouraged to provide a written brief of their presentation to the Clerk, at least 5 days in advance, which will be provided to the members of the Committee and become part of the public record.
- (8) An IN-PERSON or VIRTUAL delegation may provide additional written material regarding the subject matter, for the public record, by providing a copy of such written material to the Clerk at least 5 days in advance of the meeting.
- (9) Except as required by by-law, any IN-PERSON or VIRTUAL delegations who have previously appeared before a particular Committee on a subject matter shall be limited to providing only new information in their second and subsequent appearances at that Committee.
- (10) Notwithstanding subsection 5.12(9), a Committee member may add the delegation on a Committee Agenda.
- (11) Delegates appearing IN-PERSON or VIRTUALLY will adhere to Section 8, Order and Decorum of this by-law.
- (12) Members of Committees can ask questions of the delegations for clarification purposes only and not for debate.

5.13 Public Hearings at Standing Committees

- (1) Public Hearings shall be held at Standing Committee meetings.
- (2) Advertising or notice of a Public Hearing shall be undertaken as required by applicable legislation, by-law, or according to City policy, to advise interested persons.
- (3) Public Hearings shall commence with:
 - (a) an introduction of the subject matter by the Chair or by staff;
 - (b) the staff presentation, if any and if not waived by the members of the Standing Committee; and
 - (c) **INP** The signing in of all delegations on a sheet, provided by staff, with their name, address, including postal code, phone number and email address (optional).
 - **ELE** Presentations by delegations who have registered to attend VIRTUALLY, will be heard.
- (4) Persons who wish to appear as a delegation IN-PERSON or VIRTUALLY to address Standing Committee on a matter that is the subject of a Public Hearing may make a delegation in accordance with section 5.12.

- (5) **INP** Notwithstanding subsection 5.13(4), Standing Committee shall hear oral representations from any person in attendance at a Public Hearing, whether or not a request was made in advance.
 - ELE Notwithstanding subsection 5.13(4), Standing Committee shall only hear oral representations from persons who have registered to attend VIRTUALLY.
- (6) IN-PERSON or VIRTUAL delegations made at a Public Hearing shall be limited to an oral presentation of not more than five minutes.
- (7) Where a delegation wishes to provide written material to become part of the public record to the members of the Standing Committee at a Public Hearing, the delegation shall:
 - (a) comply with subsection 5.12(8); or
 - (b) if after the deadline, provide a copy of the written material to the Clerk at the IN-PERSON meeting for distribution after the meeting; or
 - (c) if after the deadline, provide a copy of the written material to the Clerk via email at a VIRTUAL / HYBRID meeting for distribution after the meeting.

5.14 Communication Items

- (1) Communication Items shall be addressed to the Chair and Members of Committee or the City Clerk and shall be presented to the Standing Committee for consideration if the matter is being considered at the Standing Committee meeting.
- (2) Communication Items shall be forwarded to the Office of the City Clerk no later than 12 Noon on the Monday two weeks prior to the Standing Committee meeting for inclusion in the printed Standing Committee Agenda.
- (3) When the date described in subsection 5.14(2) is a public or civic holiday, the communication item shall be forwarded to the Office of the City Clerk no later than 12 Noon on the last business day prior to the date as set out in subsection 5.14(2), for inclusion in the printed Standing Committee Agenda.
- (4) Communication Items not received for inclusion in the Standing Committee Agenda package, shall be introduced under Changes to the Agenda, provided that they are received in the Office of the City Clerk no later than 12 Noon on the business day prior to the Standing Committee meeting.
- (5) Communication Items shall not be anonymous or contain any defamatory, obscene, offensive, indecent, improper, rude or vulgar language.

- (6) Communication Items shall not contain the names of individuals (staff or public) with the exception of the author's name. Any names found in the communication will be redacted pursuant to the obligations of *Municipal Freedom of Information and Protection of Privacy Act* (MFIPPA), this Procedural By-law and other relevant legislation or procedure.
- (7) Communication Items that are relevant to a matter that appears on the Standing Committee Agenda, will be received and referred to the consideration of that matter.
- (8) Communication Items received after the prescribed deadline as set out in subsection 5.14(4) will be included on the Council Agenda, to be received and referred to the consideration of that matter within the Standing Committee Report.
- (9) Communications respecting labour relation matters, conduct of a member(s) of staff and matters before another body (i.e. Police Services Board) will not be placed on the Standing Committee Agenda and will be forwarded forthwith by the City Clerk to the relevant City Department or body for consideration and appropriate action.
- (10) Communications respecting the conduct of a member(s) of Council or any member of a body subject to review by the Integrity Commissioner will not be placed on a Standing Committee Agenda and the author will be advised of the process for filing a complaint with the Integrity Commissioner.
- (11) Notwithstanding subsections 5.14(9) and 5.14(10) a Standing Committee member may add any communication item to a Standing Committee Agenda.

5.15 Reconsideration of a Matter Decided Within a Standing Committee Meeting

- (1) At any Standing Committee Meeting, after a matter has been decided by the Standing Committee, including both successful or failed motions, a Member of the Standing Committee who voted in the majority, may present a Motion to reconsider the matter prior to the adjournment of the meeting. The Chair may ask the Member of the Standing Committee to confirm that they voted with the majority on the issue in question.
- (2) A motion to reconsider a decided matter shall require the approval of at least two-thirds of Standing Committee members present.
- (3) No motion to reconsider may, itself, be the subject of a motion to reconsider.
- (4) Debate on a motion to reconsider shall be confined to reasons for or against the reconsideration.
- (5) The following motions cannot be reconsidered:
 - (a) to adjourn;
 - (b) to recess; and
 - (c) to suspend the rules of procedure.

5.16 Conflict of Interest Declarations

- (1) Where a member has disclosed a conflict of interest at a meeting, the member shall file a written statement with the Clerk.
- (2) The Clerk shall establish and maintain a registry in which shall be kept,
 - (a) a copy of each statement filed under section 5.16(1); and
 - (b) a copy of each declaration recorded in the minutes of the meeting.
- (3) The Clerk shall make the Conflict of Interest Registry available for public inspection.

5.17 Special Meetings of a Standing Committee

- (1) In addition to scheduled Standing Committee meetings, the Mayor or the Standing Committee Chair may, at any time, summon a special meeting of a Standing Committee by giving written direction to the Clerk stating the date, time, and purpose of the special meeting.
- (2) The Clerk shall summon a special meeting of a Standing Committee when requested to do so in writing by a majority of Members of the Standing Committee.
- (3) The Clerk shall give each Member of the Standing Committee or their designated staff, notice of a special meeting of a Standing Committee at least 2 days before the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (4) The written notice to be given under subsection 5.17(3) shall state the nature of the business to be considered at the special meeting of a Standing Committee and no business other than that which is stated in the notice shall be considered at such meeting.
- (5) Notwithstanding subsection 5.17(4), other business may be considered at a special meeting with at least two-thirds of the Standing Committee members present and voting in the affirmative.
- (6) The Clerk shall prepare the Order of Business for a special meeting of a Standing Committee with the appropriate selection of agenda categories from subsection 5.11.

5.18 Change to a Scheduled Committee Meeting

- (1) The Clerk may change the time or location of a Committee meeting in consultation with the Committee Chair or the Committee Vice Chair in the Committee Chair's absence.
- (2) The Clerk shall give each Member of the Committee, or their designated staff, notice of the change to the Committee meeting at least 2 days prior to the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (3) The written notice to be given under subsection 5.18(2) shall state the reason for the change to the meeting.
- (4) The Clerk shall ensure that the change to the Committee meeting is posted to the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

SECTION 6- RULES OF DEBATE, VOTING PROCEDURES AND MOTION PROCEDURES

6.1 Rules of Debate

- (1) No Member of Council shall be deemed to have precedence or seniority over any other Member of Council.
- (2) Before speaking to a question or motion, a Member of Council shall be recognized by the Mayor or the Chair and be placed on the speakers' list.
- (3) When a Member of Council is speaking, no other Member of Council shall pass between that Member of Council and the Mayor or the Chair or interrupt that Member of Council except to raise a point of order or privilege.
- (4) A Member of Council may require the question or motion under discussion to be read at any time during the debate but not so as to interrupt a Member of Council who is speaking.
- (5) A Member of Council may not speak on any matter other than the matter being debated.
- (6) All Members of Council are restricted to a five minute time limit when asking a question and/or making a statement during Council and Committee meetings.
- (7) Notwithstanding subsection 6.1(6), an extension of a specific allotment of time to the five minute speaking restriction may be granted with the approval of a majority of the members of the Council or Committee present.

- (8) The following matters may be introduced orally without written notice and without leave:
 - (a) a point of order
 - (b) a point of privilege
 - (c) a motion to adjourn
 - (d) a motion to call the question
 - (e) a motion to go into committee of the whole
 - (f) a motion to recess
 - (g) a motion to refer
 - (h) a motion to defer
 - (i) a motion to go into a closed meeting
 - (j) a motion to suspend the rules of procedure
 - (k) a motion to lift

(9) Relinquishing the Chair

The Mayor or the Chair may designate another Member as acting Chair during any part of a meeting and while presiding, such Member shall have all the powers of the Mayor or the Chair with respect to chairing the meeting.

6.2 Voting Procedures

- (1) Once a motion is moved and seconded it shall be deemed to be in the possession of Council or Committee.
- (2) Where a Member of Council present at a meeting of Council or Committee is prohibited from voting by statute, the Clerk shall record the name of the Member of Council and the reason that they are prohibited from voting.
- (3) INP When an electronic voting system is available at an IN-PERSON meeting every Member of Council or Committee present and participating in a vote on a motion, shall vote electronically excluding those motions referred to in (i) and (ii) below which shall be by show of hands:
 - (i) procedural (majority vote: refer, defer, withdraw, recess, call the question and adjourn); and
 - (ii) Consent Items.
 - When an electronic voting system is available at a VIRTUAL / HYBRID meeting every Member of Council or Committee present and participating in a vote on a motion, shall vote electronically.
- (4) When an electronic voting system is available for a vote, the results will be displayed and recorded in the proceedings.

- (5) When an electronic voting system is not available every Member of Council or Committee present shall vote by a show of hands.
- (6) When an electronic vote is taken and a member advises the Mayor or the Chair of a point of privilege with respect to their vote being omitted or in error immediately following and prior to the taking of another vote, the vote will be considered void and a new electronic vote will be retaken immediately.
- (7) When an electronic vote is taken and a member advises the Mayor or the Chair of a point of privilege with respect to their vote being omitted or in error following the taking of another vote, the member's request will be considered through a vote to reconsider the matter:
 - (i) If a two-thirds majority vote to reconsider is decided in the affirmative, the electronic vote will be retaken immediately;
 - (ii) If a two-thirds majority vote to reconsider is decided in the negative, the original electronic vote stands.
- (8) When a vote is taken and a tie results, a question is deemed to be lost.
- (9) INP When a Member of Council at an IN-PERSON meeting of Council or Committee does not disclose a pecuniary interest, and does not vote on a Motion, the member will be deemed as having voted in the negative and when an electronic voting system is in use the recusal will be displayed and noted in the proceedings.
 - When a Member of Council at a VIRTUAL / HYBRID meeting of Council or Committee does not disclose a pecuniary interest, and does not vote electronically on a Motion, the Mayor/Chair will call on the member to vote 3 (three) times, if the member doesn't respond the member will be noted as being 'Not Present' during the vote.
- (10) When an electronic voting system is not available and a recorded vote is requested by a Member of Council or Committee immediately before or after a question is put, the Clerk shall record the name and vote of every Member of Council or Committee on the question.
- (11) When the Mayor or the Chair calls for a vote on a question, each Member of Council or Committee shall occupy their seat and shall remain in their seat until the result of the vote has been declared by the Mayor or the Chair, and during such time no Member of Council or Committee shall walk across the room or speak to any other Member of Council or Committee or make any noise or disturbance.
- (12) After a question is put by the Mayor or the Chair, no Member of Council or Committee shall speak to the question nor shall any other motion be made until after the vote is taken and the result has been declared.
- (13) No vote shall be taken by ballot or by any other method of secret voting and every vote so taken is of no effect, except where permitted by statute.

- (14) Despite subsection 6.2(13), a meeting may be closed to the public during a vote if:
 - (a) the vote relates to an issue permitting the meeting to be closed to the public in accordance with section 9 (Meetings Closed to the Public) of this By-law; and
 - (b) the vote is for a procedural matter or for giving directions or instructions to officers, employees or agents of the City or persons retained by or under a contract with the City.
- (15) The Mayor or the Chair shall, upon request of a Member of Council or Committee, divide a question, if the question under consideration contains distinct propositions, and the vote upon each proposal shall be taken separately.
- (16) If a Member of Council or Committee disagrees with the announcement of the Mayor or the Chair that a question is Carried or Lost on a vote by show of hands, they may immediately after the declaration by the Mayor or the Chair, object to the Mayor's or the Chair's declaration and require that a recorded vote be taken.
- (17) Members of Council who are appointed as representatives of an Advisory Committee, do not have voting rights while attending an Advisory Committee meeting.

6.3 Motion Procedures

(1) Notice of Motion

- (a) Notice of all new motions, except motions listed in subsection 6.1(8) shall be given in writing:
 - (i) at a meeting of Council or Committee, but shall not be debated until the next meeting of Council or Committee; or
 - (ii) delivered to the Clerk at any time before noon on the last business day preceding the date of the meeting at which the notice of motion is to be introduced.
- (b) Where a Member of Council's or Committee's motion has been called by the Mayor or the Chair at the subsequent meeting and not proceeded with, it shall be dropped from the agenda unless Council or Committee decides otherwise.
- (c) Where Council or Committee has determined not to drop a motion from the agenda, and at the second meeting such motion is called by the Mayor or the Chair and not proceeded with, it shall be deemed to have been withdrawn.

(2) Dispensing with Notice

Where a matter is time sensitive or is an emergency, the Motion may be introduced without notice if Council or Committee, without debate, dispenses with notice on the affirmative vote of two-thirds of the Members of Council or Committee present and voting.

(3) Motion to Withdraw

After a motion is moved and seconded it shall be deemed to be in possession of Council or Committee and may be withdrawn by the mover at any time before decision or amendment.

(4) Motion to Amend

- (a) A motion to amend:
 - (i) shall be presented in writing, duly moved and seconded;
 - (ii) shall be made only to a previous question or to amend an amendment to the question;
 - (iii) shall be relevant to the question to be decided; shall not be received if, in essence it constitutes a rejection of the main question;
 - (iv) shall be put to a vote in the reverse order to that which it is moved.
- (b) If the amendment is not considered a "Friendly Amendment", then the motion to amend:
 - (i) If more than one, will be put to a vote in the reverse order to that in which they were moved;
 - (ii) Will be decided upon or withdrawn before the main motion is put to the vote;
 - (iii) Will not be amended more than once, provided that further amendments may be made to the main motion; and,
 - (iv) Will not be directly contrary to the main motion.

(5) Motion to Defer

A motion to defer a question:

- (a) is debatable; and
- (b) shall include the date when the question is to be considered.

(6) Motion to Refer

- (a) A motion to refer the question:
 - shall include the name of the individual or body to whom the question is to be referred;
 - (ii) may include instructions respecting the terms upon which the question is to be referred;
- (b) A motion to refer a question is not debatable except where instructions are included, in which case, only the instructions shall be debatable.
- (c) A motion to refer a question may be amended in accordance with subsection 6.3(4).
- (d) When a motion to refer a question to Council is defeated by a Standing Committee, the Chair of the Standing Committee shall call the question on the matter.
- (e) A member who has spoken to the motion under debate shall not be permitted to move a motion to refer a question.
- (f) A motion to refer a question is only in order after all first time speakers have been heard.

(7) Motion to Adjourn

A motion to adjourn a Council or Committee meeting shall always be in order except:

- (a) when a Member of Council or Committee is speaking or during the taking of a vote;
- (b) immediately following the affirmative resolution of a motion that a vote on the question now be taken;
- (c) when a Member of Council or Committee has already indicated to the Mayor or the Chair, that they desire to speak on the question;
- (d) when decided in the negative, cannot be made again until Council or Committee has conducted further proceedings.

(8) Motion to Call the Question

A motion to call the question shall be in order except when a Member of Council or Committee is speaking or before the first-time speakers' list is exhausted.

(9) Motion to Reconsider a Matter Decided by Council

(a) At any Council Meeting, after a matter has been decided by Council, including both successful or failed motions, a Member of Council who voted in the majority, may present a Notice of Motion to reconsider the matter. The Chair may ask the Member of Council to confirm that they voted with the majority on the issue in question. Such Notice of Motion shall be referred to the next scheduled meeting of Council and shall be included on the agenda under "Motions".

- (b) No decided matter shall be reconsidered more than once during the term of Council.
- (c) A motion to reconsider a decided matter shall require the approval of at least two-thirds of Council present.
- (d) It is not in order to reconsider a decided matter when the motion has been implemented, resulting in a legally binding commitment that is in place on the date that a motion to reconsider is to be debated.
- (e) No motion to reconsider may, itself, be the subject of a motion to reconsider.
- (f) Debate on a motion to reconsider shall be confined to reasons for or against the reconsideration.
- (g) The following motions cannot be reconsidered:
 - (i) to adjourn;
 - (ii) to recess; and
 - (iii) to suspend the rules of procedure.
- (h) Subsections 6.3(9)(a) to (f) do not apply when:
 - (i) a motion pertains to a decision of a previous Council;
 - (ii) a motion that, while pertaining to a previously decided motion, does not alter the core purpose or intent of the previously decided motion;
 - (iii) a motion presented in respect of a previously decided matter for the purpose of strengthening or clarifying the original decision; or
 - (iv) a motion presented in respect of a previously decided matter for the purpose of moving a matter through a series of necessary steps or phases.

(10) Order of Procedural Motions

The following motions are procedurally in order in the following sequence:

- (a) to adjourn;
- (b) to recess;
- (c) to request information;
- (d) to call the question;
- (e) to defer;
- (f) to refer;
- (g) to amend.

(11) Non-Debatable Motions

The following motions/actions are not debatable:

- (a) a motion to adjourn;
- (b) a motion to refer (except where instructions are included, in which case, only the instructions shall be debatable);
- (c) a motion to suspend the rules of procedure;
- (d) a point of order; and
- (e) a point of privilege.

(12) Motion to Suspend the Rules of Procedure

The rules governing the procedure of the Council or Committee at Council or Committee meetings may be suspended and shall require the approval of at least two-thirds of members present.

(13) Motions Beyond Jurisdiction of Council

A motion in respect of a matter which is beyond the jurisdiction of Council shall not be in order.

SECTION 7 – ESTABLISHMENT OF ADVISORY COMMITTEES OR TASK FORCES

- **7.1** Council may from time to time, as needed, establish Advisory Committees or Task Forces in response to specific matters requiring immediate or long term attention.
- **7.2** When Council approves of the formation of the Advisory Committee or Task Force the procedures as outlined in Appendix H shall be followed.
- **7.3** When an Advisory Committee or Task Force has completed its mandate, the Standing Committee to which the Advisory Committee or Task Force reports shall dissolve it by motion.

SECTION 8 - ORDER AND DECORUM

- **8.1** No Person in attendance IN-PERSON or VIRTUALLY at a Council or Committee meeting shall:
 - (a) be disrespectful or disrupt the Meeting in any manner;
 - (b) bring food or beverages, water excepted, into the Council Chamber when an IN-PERSON meeting is being held;
 - (c) address remarks to anyone but the Chair;
 - (d) interrupt a person who has the floor;
 - (e) applaud participants in debate.

- 8.2 Should a person persist in conducting themselves in a manner contrary to the rules set forth in subsection 8.1 after having been called to order by the Mayor or a Chair of a Committee, the Mayor or a Chair of a Committee may expel that person and may request the assistance of security and/or Police in doing so.
- **8.3** Notwithstanding subsection 8.2, in the event a member of the public is conducting themselves in a manner which poses a threat to the personal safety of themselves or others in attendance at the meeting, they shall be removed from the meeting, without warning.
- 8.4 Should a Member of Council persist in conducting themselves in a manner contrary to the rules set forth in subsection 8.1 after having been called to order by the Mayor or the Chair of the Committee, the Mayor or the Chair may order them to vacate the place the meeting is being held. If the Member of Council apologizes, they may, by vote of a majority of the Council or Committee members, be permitted to remain at the meeting.
- **8.5** No person except members of Council or a Committee, the Clerk, and officials authorized by the Clerk, shall:
 - (a) be allowed on the floor while an IN-PERSON meeting is being held, namely where members of Council or Committee, the Clerk, and officials authorized by the Clerk are seated and from which they speak;
 - (b) before or during an IN-PERSON meeting of the Council or a Committee, place on the desks of members or otherwise distribute any material whatsoever unless such person is so acting with the approval of the Clerk.
- **8.6** Signs or placards are permitted at Council and Committee Meetings, provided:
 - (a) they do not contain disrespectful or offensive language:
 - (b) they do not disrupt the Meeting;
 - (c) they do not disrupt an attendee's ability to view the proceedings; and
 - (d) they do not contain solid handles or hard backings.
- **8.7** Electronic devices at IN-PERSON / VIRTUAL / HYBRID meetings:
 - (a) Every person shall have all electronic devices, including but not limited to phone, computers, and similar electronic devices, etc. switched to a non-audible function during Council and Committee meetings.
 - (b) The use of electronic devices at IN-PERSON / VIRTUAL / HYBRID Closed Meetings is strictly prohibited.
 - (c) Staff from the Office of the City Clerk are exempted from subsection 8.7(b) when using electronic devices for record-keeping purposes.

SECTION 9 – MEETINGS CLOSED TO THE PUBLIC

For the purposes of this section, "Committee" means any advisory or other committee, subcommittee or similar entity of which at least 50% of the members are also members of Council.

- **9.1** No meeting of Council or a Committee or part thereof shall be closed to the public unless the subject matter being considered is:
 - (a) the security of the property of the City or a local board;
 - (b) personal matters about an identifiable individual, including City or a local board employees;
 - (c) a proposed or pending acquisition or disposition of land for City or a local board purposes;
 - (d) labour relations or employee negotiations;
 - (e) litigation or potential litigation, including matters before administrative tribunals, affecting the City or a local board;
 - (f) the receiving of advice that is subject to solicitor-client privilege, including communications necessary for that purpose;
 - (g) a matter in respect of which Council or a Committee may hold a closed meeting under an Act other than the *Municipal Act, 2001*;
 - (h) information explicitly supplied in confidence to the City or a local board by Canada, a province or territory or a Crown agency of any of them;
 - (i) a trade secret or scientific, technical, commercial, financial or labour relations information, supplied in confidence to the municipality or local board, which, if disclosed, could reasonably be expected to prejudice significantly the competitive position or interfere significantly with the contractual or other negotiations of a person, group of persons, or organization;
 - (j) a trade secret or scientific, technical, commercial or financial information that belongs to the City or a local board and has monetary value or potential monetary value; or
 - (k) a position, plan, procedure, criteria or instruction to be applied to any negotiations carried on or to be carried on by or on behalf of the City or a local board.
- **9.2** A meeting of Council shall be closed to the public if the subject matter relates to the consideration of:
 - (a) a request under the *Municipal Freedom of Information and Protection of Privacy Act* if Council is designated as head of the institution for the purposes of that Act; and
 - (b) an ongoing investigation respecting the City or a municipally-controlled corporation by the Ombudsman appointed under the *Ombudsman Act*.
- **9.3** A meeting of Council or a Committee may be closed to the public if:
 - (a) it is held for the purposes of educating or training members; and
 - (b) at the meeting, no member discusses or otherwise deals with any matter in a way that materially advances the business or decision-making of Council or the Committee.

- **9.4** Before holding a meeting or part of a meeting that is to be closed to the public, a Council or Committee shall:
 - (a) state specifically by resolution the fact of the holding of the closed meeting and the general nature of the matter to be considered at the closed meeting;
 - (b) state specifically that members of the public will be invited to return to hear any further deliberations when Council or Committee reconvenes in Open Session, indicated by the removal of the frosting on the windows and the elimination of the white noise; and
 - (c) Council or Committee will, in the event they are in Closed Session for more than 30 minutes, wait up to 5 minutes upon reconvening in Open Session before proceeding with the meeting, to provide members of the public and the media time to return to the meeting room.

SECTION 10 – VIRTUAL PARTICIPATION AT COUNCIL AND/OR COMMITTEE MEETINGS

10.1 Virtual participation is prohibited at Council and/or Committee meetings, except where permitted during an Emergency as set out in subsection 10.2.

10.2 Virtual Participation During an Emergency

- (1) Virtual participation, in accordance with the *Municipal Act, 2001* and/or any other applicable legislation, is permitted by members of Council at Council and/or Committee meetings during an emergency when attending in-person is not possible.
- (2) Virtual participation, in accordance with the *Municipal Act, 2001* and/or any other applicable legislation, is permitted by Citizen Committee members at Committee meetings during an emergency when attending in-person is not possible.
- (3) Virtual participation, that is consistent with the obligation to hold public meetings in accordance with the *Municipal Act, 2001* and/or any other applicable legislation, or at a Standing Committee of Council is permitted by members of the public at Committee meetings.

SECTION 11 – VIRTUAL MEETINGS DURING AN EMERGENCY

- **11.1** All VIRTUAL/ HYBRID Council and Committee meetings shall be streamed live.
- **11.2** Upon being apprised that a virtual meeting is interrupted and cannot proceed with its live feed, due to loss of connection, sound or video, the meeting will be recessed for up to 15 minutes.

- 11.3 If the live feed of a Virtual meeting cannot be resumed within 15 minutes, the meeting will be considered adjourned. The Council/ Committee meeting will be rescheduled to another day or the next regularly scheduled meeting date.
- 11.4 If the Virtual meeting is interrupted and its live feed cannot be resumed, the names of the Council/Committee members present and any decisions of Council/Committee up to the point in time of the interruption shall be recorded in the minutes of the meeting.

SECTION 12 - ROLE OF CLERK

- **12.1** The Clerk shall make the agendas of Council and Committee meetings available to members of Council and Senior Leadership Team electronically at least 5 days prior to the scheduled meetings.
- 12.2 The Clerk shall make the agendas of Council and Committee meetings available to the media and general public electronically, simultaneously with the distribution set out in subsection 12.1 on the City's website.
- **12.3** The Clerk shall advise the Mayor or Chair when items are required to be added to or removed from an agenda.
- 12.4 The Clerk shall submit for confirmation the minutes of the previous Council or Committee meeting. Upon approval of the majority of the members of Council or Committee, the Mayor or Chair and Clerk shall sign the minutes as confirmed or as amended.
- 12.5 The Clerk shall advise the Mayor or Chair, if in their opinion, a matter or portion of a matter being discussed in a meeting that is closed to the public is not procedurally appropriate in accordance with section 239 of the Act and section 9 of this By-law.
- 12.6 The Mayor or Chair may call upon the Clerk to provide advice regarding procedure. The Clerk shall provide this advice following which the Mayor or Chair will announce their ruling.
- 12.7 The Clerk shall assume the role of the Mayor or Chair as the chair of Council or a Committee, when a ruling of the Mayor or Chair is challenged by a Member of Council or Committee. The Clerk shall take a vote of the challenge for Members of Council or Committee in favour or opposed to the ruling. The ruling fails when there is a majority vote in favour of the challenge.

SECTION 13 - GENERAL

- **13.1** This By-law shall not be amended or repealed except by a majority vote of all Members of Council.
- **13.2** General guidelines describing the respective roles and responsibilities of Council, the Mayor and the Chair of a Committee are set out in Appendix G to this By-law.
- **13.3** This By-law comes into force on the date on which it is passed.
- **13.4** The short title of this By-law is the Procedural By-law or the Council Procedural By-law.

	3.5 By-law Nos. 18-270, 19-090, 19-212,19-308, 20-042, 20-055, 20-103, 20-129, 20-145, 20-146, 20-151, 20-184, governing and amending the rules and procedures of Council of the City of Hamilton are repealed.			
PASSI	ED this	day of	, 2021.	
F. Eise Mayor	enberger		A. Holland City Clerk	

Appendix A

GENERAL ISSUES COMMITTEE

COMPOSITION

The General Issues Committee shall be comprised of all 16 members of Council.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Council Strategic Plan
- Corporate Strategic Plan
- Municipal Benchmarking Network Canada (MBNC)
- Annual Operating and Capital Budgets
- Economic Development matters
- Portfolio Management Strategy Real Estate Acquisitions/Disposals
- Hamilton International Airport matters
- Growth Related Integrated Development Strategy (GRIDS)
- Vision 20/20
- Upper Levels of Government Funding Matters
- Department Work Programs: Planning and Economic Development
- Legal Services all matters having definite litigation
- Human Resources labour negotiations
- Departmental Organizational Structure Changes
- Boards and Agencies
- Hamilton Utilities Corporation
- Climate Change
- any and all other matters which Council chooses to refer to the General Issues Committee for consideration

Specific duties shall include:

- To review and monitor the City's and Council's Strategic Plan
- To review corporate and program objectives and performance measures and make recommendations to Council
- To consider and recommend to Council on matters relating to budgets, budget monitoring, re-assessment and related tax policies
- To consider and recommend to Council on matters relating to Business Development, the Small Business Enterprise Centre, Incentive Loans/Grants programs and approvals, BIA initiatives, the Hamilton Incubator of Technology and Tourism
- To consider and recommend to Council on matters relating to GRIDS
- To receive briefings on legal matters involving the City and give direction to the City Solicitor on litigation matters
- To receive information on labour negotiations and provide direction to the Director of Labour Relations

Appendix A

- To consider and make recommendations to Council on matters regarding Boards and Agencies
- To Meet as Shareholders and/or Board of Directors of a corporation when required.
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee

Page 41 of 66

Appendix B

BOARD OF HEALTH

COMPOSITION

The Board of Health shall be comprised of all 16 members of Council.

MANDATE

General:

To ensure the City of Hamilton meets or exceeds its obligations as a Board of Health under the *Health Protection and Promotion Act*, and in all matters relating to Public Health in order to promote well-being and create opportunities to enhance the quality of life in our community.

Specific duties shall include:

- To consider and recommend to Council on policy matters and emerging issues related to Public Health
- To provide advice and guidance to the Public Health Services including input to Strategic Planning initiatives to ensure alignment with Council's Mission, Vision, Values and Goals
- To receive delegations from the public and conduct public hearings as required by statute and Council
- To consider and recommend to Council the overall service levels in relation to the delivery of Public Health programs
- To consider and recommend to Council policies governing service delivery of the Public Health Services
- To consider and monitor program implementation and performance through staff reports
- To consider the public health needs of the community and recommend advocacy by Council to address these needs
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Appendix C

PUBLIC WORKS COMMITTEE

COMPOSITION

The Public Works Committee shall be comprised of a minimum of 8 Members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Water & Wastewater (Collection/Distribution, Treatment, and Compliance)
- Waste Management (Solid Waste Planning, Collection, and Disposal)
- Operations & Maintenance (Roads, Traffic, Forestry & Beautification, Parks & Cemeteries, and Trails)
- Transit (Operations and Transit Fleet, Accessible Transportation Services (ATS))
- Fleet and Facilities (Central Fleet, Corporate Buildings, Energy Office)
- Capital Planning and Implementation (Strategic and Environmental Planning, Asset Management, Design and Construction, Open Space Planning)
- Major road construction projects

Specific duties shall include:

- To consider and recommend to Council, Service Programs and Service Levels for all direct and indirect services provided by the Department (i.e. business plans, management plans, operating plans, and other relevant plans)
- To consider and recommend to Council, Policies, By-laws, and procedures governing service delivery implemented by the Department
- To consider and recommend to Council, Asset Management Plans and Forecasts for all infrastructure managed by the Department
- To consider and monitor program implementation and performance through staff reports and make recommendations to Council respecting program management initiatives (i.e. continuous improvement, budget performance, revenue generation and other relevant issues)
- To consider and recommend to Council, procurement in accordance with corporate policies
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Public Works Committee
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Page 43 of 66

Appendix D

PLANNING COMMITTEE COMPOSITION

The Planning Committee shall be comprised of a minimum of 9 Members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Long Range Planning, land use management, development planning and engineering, transportation planning and downtown planning and implementation
- Parking Operations and Enforcement, School Crossing Guards
- By-law Enforcement, Municipal Licensing, Lottery Licensing, Building Code issues

Specific duties shall include:

To consider and recommend to Council on:

- the administration and enforcement of the Ontario Building Code, Zoning Bylaws, the Property Standards By-laws, Licensing By-law, Animal Control Bylaw, Sign By-law and other relevant By-laws
- all matters related to the *Planning Act*, the *Ontario Heritage Act*, the *Municipal Act*, 2001, the Niagara Escarpment Act and other applicable legislation regarding planning, development engineering, by-law enforcement and licensing and downtown planning issues
- the City's Official Plan and Zoning By-laws and amendments thereto, pursuant to the *Planning Act*, and to conduct such related public meetings as may be required
- applications for subdivision and condominium approval pursuant to the *Planning Act* and *Condominium Act*, as applicable
- matters such as community planning, urban design guidelines, heritage policy and related housing policy and programs in accordance with the directions contained in the Official Plan
- matters relating to Parking Operations and Enforcement
- To advise and assist Council in any matters arising from the work of the Committee of Adjustment in considering applications for consent and minor variance under the *Planning Act*, including possible City participation at any Local Planning Appeal Tribunal (LPAT) Hearings to consider the appeal of Committee of Adjustment decisions
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Planning Committee
- To receive delegations for the public and conduct public hearings as required by statute and Council, specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Appendix E

EMERGENCY & COMMUNITY SERVICES COMMITTEE

COMPOSITION

The Emergency & Community Services Committee shall be comprised of a minimum of 5 members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Children's Services
- Ontario Works
- Neighbourhood Development
- Housing Services
- Lodges: Macassa Lodge and Wentworth Lodge
- Recreation
- Fire Services
- Paramedic Services
- Advisory Committees that report to the Emergency & Community Services Committee

Specific duties shall include:

To consider and recommend to Council on:

- service levels for all direct and indirect services provided by the Department
- policies governing service delivery implemented by the Department
- procurement in accordance with corporate policies
- plans for community facility infrastructure
- the delivery of protective services including fire operations, fire prevention, corporate emergency management, and corporate radio communications
- the delivery of emergency medical (paramedic) services
- To consider and monitor program implementation and performance through staff reports
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Emergency & Community Services Committee
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Appendix F

AUDIT, FINANCE & ADMINISTRATION COMMITTEE

COMPOSITION

The Audit, Finance & Administration Committee shall be comprised of a minimum of 7 members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Internal/external audits;
- Oversight of internal control, financial reporting;
- Human Resources programs and services;
- Programs related to Financial Planning and Policy, Treasury Services, Customer Service, Information Technology, Legislative Services, Records Management, Legal Services, Procurement and Risk Management Access & Equity; and
- To hold hearing on complaints made pursuant to Section 20 of the *Development Charges Act, 1997*

Specific duties shall include:

To consider and make recommendations to Council on:

- matters of policy respecting human resources, including health and safety, union/management relations, organizational planning and development and compensation administration
- matters of policy involving communications issues
- matters of policy and oversight involving financial management, investment, reserves, debt, procurement and risk management
- matters of policy involving general policies and procedures and administrative By-laws
- all audit matters, including promoting an appropriate environment for the management of public funds and the economy, efficiency and effectiveness of operations and a high level of accountability. Ensure compliance with laws, regulations, policies and support high standard of ethical conduct
- matters of policy and direction related to Access and Equity, including the annual corporate accessibility plan
- To act as liaison to the Access & Equity volunteer advisory committees (with the exception of the Advisory Committee for Persons with Disabilities reporting to the General Issues Committee)
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Audit, Finance & Administration Committee
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Schedule F1 Office of the City Auditor Charter
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Page 46 of 66

Schedule F1 to Appendix F

CITY OF HAMILTON OFFICE OF THE CITY AUDITOR CHARTER

INTRODUCTION

The Office of the City Auditor provides independent, objective audit assurance and consulting services designed to add value and improve the City of Hamilton's operations. The Office of the City Auditor (OCA) brings a systematic, disciplined approach to evaluating and improving the effectiveness of risk management, control and governance processes.

In the capacity of Auditor General, the City Auditor assists City Council in holding itself and its administrators accountable for the quality of stewardship over public funds and for the achievement of value for money in City operations.

SCOPE

The Office of the City Auditor completes the following types of work:

- Compliance Auditing
- Value for Money Auditing
- Special Investigations (Fraud and Waste, Whistleblower)
- Risk assessments
- Consulting

The first three activities are conducted by the Office in its capacity as Auditor General.

Audit Assurance and Special Investigations

Audit and Assurance work includes the examination and evaluation of the adequacy and effectiveness of the City's governance, risk management process, systems of internal control and the quality of performance in carrying out assigned responsibilities to achieve the organization's goals and objectives. This includes both compliance and value for money audits. The City's processes should function in a manner that ensures:

- Risks are appropriately identified and managed.
- Significant financial, managerial, and operating information is accurate, reliable and timely.
- Actions are in compliance with policies, standards, procedures, and applicable laws and regulations.
- Resources are acquired economically, used efficiently and are adequately protected.
- Programs, plans and objectives are achieved.
- Significant legislative and regulatory issues impacting the City are recognized and addressed properly.

 Quality and continuous improvement are fostered in the organization's internal control processes.

In addition to audits, special investigations may be undertaken by the OCA pursuant to Council having given that authority under the Whistleblower By-law (19-181) and the City of Hamilton's Fraud Policy and Protocol. These include reviews or investigations of matters of alleged or suspected wrongdoing, per the Whistleblower By-law or fraud, waste or other similar activities reported by employees or residents through the Fraud and Waste Hotline or other mechanisms.

The City Auditor has been appointed by By-law 19-180 (City Auditor By-law) as an Auditor General under Section 223.19 of the *Municipal Act, 2001*, with the responsibilities, including the powers, duties and protections, under Sections 223.19 to 223.23 of the *Municipal Act, 2001*. These responsibilities apply to the extent authorized by these sections of the *Municipal Act, 2001*. The powers and protections provided to the City Auditor (by appointment as an Auditor General) strengthens the position of independence and institutional authority of the City Auditor while providing information to Council for its oversight role.

Consulting and Risk Assessment

In addition to audit engagements, the Office of the City Auditor provides advisory or other consulting services, as planned, or at the request of Council or senior management. These types of services may include:

- Risk assessments and related workshops;
- Performing research;
- Providing education/training on audit related topics such as risk management, fraud awareness, performance measurement and internal controls;
- Performing Lessons Learned/Opportunities Assessments on project/program implementations that have experienced difficulties or challenges; and
- Providing counsel or advice (e.g. on the adequacy of draft procedures).

<u>AUTHORITY</u>

The Office of the City Auditor is granted full, free and unrestricted access to any and all records, property and personnel relevant to any function under review. Access to personal information is provided for under the *Municipal Freedom of Information and Protection of Privacy Act* (in particular, Subsections 31(c) and 32(d)).

The OCA has the authority to conduct audits and reviews of all City departments, Members of Council, agencies, boards and commissions, as well as other entities the City is related to or has an interest in.

All employees will assist the OCA in fulfilling its objectives.

As an Auditor General, the OCA has the responsibilities, including the powers, duties and protections, under Sections 223.19 to 223.23 of the *Municipal Act, 2001* for:

- City Departments;
- Members of Council;
- Local boards (not including the Board of Health, the Hamilton Public Library Board, the Police Services Board or other local boards in accordance with the current definition of "local board" under Section 223.1 of the *Municipal Act*, 2001);
- Municipally-controlled corporations (a corporation that has 50 per cent or more of its issued and outstanding shares vested in the City or that has the appointment of a majority of its board of directors made or approved by the City, not including a corporation established in accordance with Section 203 of the *Municipal Act*, 2001); and
- Grant recipients (as currently defined under Section 223.1 of the *Municipal Act, 2001* as "a person or entity that receives a grant directly or indirectly from the municipality, a local board or a municipally-controlled corporation");

These responsibilities under Section 223.19 to 223.23 of the *Municipal Act, 2001* include the powers to access information and to examine persons under Section 33 of the *Public Inquiries Act, 2009*; the duty to preserve secrecy with respect to all matters that come to its knowledge in the course of performing its functions; and the protection of not being a competent or compellable witness in a civil proceeding.

The relevant excerpts from the *Municipal Act* (accessed on July 16, 2019 at https://www.ontario.ca/laws/statute/01m25#BK272) are included below for reference:

Auditor General

223.19 (1) Without limiting sections 9, 10 and 11, those sections authorize the municipality to appoint an Auditor General who reports to council and is responsible for assisting the council in holding itself and its administrators accountable for the quality of stewardship over public funds and for achievement of value for money in municipal operations. 2006, c. 32, Sched. A, s. 98.

Same

(1.1) The Auditor General shall perform his or her responsibilities under this Part in an independent manner. 2009, c. 33, Sched. 21, s. 6 (11).

Exceptions

(2) Despite subsection (1), the responsibilities of the Auditor General shall not include the matters described in clauses 296 (1) (a) and (b) for which the municipal auditor is responsible. 2006, c. 32, Sched. A, s. 98.

Powers and duties

(3) Subject to this Part, in carrying out his or her responsibilities, the Auditor General may exercise the powers and shall perform the duties as may be assigned to him or her by the municipality in respect of the municipality, its local boards and such municipally-controlled corporations and grant recipients as the municipality may specify. 2006, c. 32, Sched. A, s. 98.

Grant recipients

(4) The authority of the Auditor General to exercise powers and perform duties under this Part in relation to a grant recipient applies only in respect of grants received by the grant recipient directly or indirectly from the municipality, a local board or a municipally-controlled corporation after the date on which this section comes into force. 2006, c. 32, Sched. A, s. 98.

Delegation

(5) The Auditor General may delegate in writing to any person, other than a member of council, any of the Auditor General's powers and duties under this Part. 2006, c. 32, Sched. A, s. 98.

Same

(6) The Auditor General may continue to exercise the delegated powers and duties, despite the delegation. 2006, c. 32, Sched. A, s. 98.

Status

(7) The Auditor General is not required to be a municipal employee. 2006, c. 32, Sched. A, s. 98.

Duty to furnish information

223.20 (1) The municipality, its local boards and the municipally-controlled corporations and grant recipients referred to in subsection 223.19 (3) shall give the Auditor General such information regarding their powers, duties, activities, organization, financial transactions and methods of business as the Auditor General believes to be necessary to perform his or her duties under this Part. 2006, c. 32, Sched. A, s. 98.

Access to records

(2) The Auditor General is entitled to have free access to all books, accounts, financial records, electronic data processing records, reports, files and all other papers, things or property belonging to or used by the municipality, the local board, the municipally-controlled corporation or the grant recipient, as the case may be, that the Auditor General believes to be necessary to perform his or her duties under this Part. 2006, c. 32, Sched. A, s. 98.

No waiver of privilege

(3) A disclosure to the Auditor General under subsection (1) or (2) does not constitute a waiver of solicitor-client privilege, litigation privilege or settlement privilege. 2006, c. 32, Sched. A, s. 98.

Powers re examination

223.21 (1) The Auditor General may examine any person on oath on any matter pertinent to an audit or examination under this Part. 2006, c. 32, Sched. A, s. 98.

Application of Public Inquiries Act, 2009

(2) Section 33 of the Public Inquiries Act, 2009 applies to an examination by the Auditor General. 2009, c. 33, Sched. 6, s. 72 (3).

Duty of confidentiality

223.22 (1) The Auditor General and every person acting under the instructions of the Auditor General shall preserve secrecy with respect to all matters that come to his or her knowledge in the course of his or her duties under this Part. 2006, c. 32, Sched. A, s. 98.

Same

- (2) Subject to subsection (3), the persons required to preserve secrecy under subsection (1) shall not communicate information to another person in respect of any matter described in subsection (1) except as may be required,
- (a) in connection with the administration of this Part, including reports made by the Auditor General, or with any proceedings under this Part; or
- (b) under the Criminal Code (Canada). 2006, c. 32, Sched. A, s. 98.

Same

(3) A person required to preserve secrecy under subsection (1) shall not disclose any information or document disclosed to the Auditor General under section 223.20 that is subject to solicitor-client privilege, litigation privilege or settlement privilege unless the person has the consent of each holder of the privilege. 2006, c. 32, Sched. A, s. 98.

Section prevails

(4) This section prevails over the Municipal Freedom of Information and Protection of Privacy Act. 2006, c. 32, Sched. A, s. 98.

Testimony

223.23 Neither the Auditor General nor any person acting under the instructions of the Auditor General is a competent or compellable witness in a civil proceeding in connection with anything done under this Part. 2006, c. 32, Sched. A, s. 98.

INDEPENDENCE

Independence is an essential component to maintaining public trust and preserving objectivity and integrity associated with the audit function.

To provide for the independence of Office of the City Auditor, its personnel report to the City Auditor, who reports administratively to the City Manager and functionally to the Audit, Finance & Administration Committee of Council. Audit and other reports are sent directly to the Audit, Finance & Administration Committee for discussion and approval and then to Council. These reporting relationships help ensure independence, promote comprehensive audit objectivity and coverage and assure adequate consideration of audit recommendations.

All OCA activities shall remain free of influence by any element in the organization, including matters of audit selection, scope, procedures, frequency, timing or report content to permit maintenance of an independent and objective attitude necessary in rendering reports. The City Auditor has the authority to revise and extend the scope of any audit or investigation in the course of their examination.

The OCA shall have no direct operational responsibility or authority over any of the activities it reviews. Accordingly, it shall not develop nor install systems or procedures, prepare records or engage in any other activity which would normally be audited.

RESPONSIBILITIES

The City Auditor and the staff of the OCA have the responsibility to:

- Review operations within the City at appropriate intervals to determine whether planning, organizing, directing and controlling are in accordance with management instructions, policies and procedures and in a manner that is consistent with both City objectives and high standards of administrative practice.
- Determine the adequacy and effectiveness of the systems of internal accounting, financial and operating controls.
- Review the reliability, utility and integrity of financial information and the means used to identify, measure, classify and report such information.
- Review the established systems to ensure compliance with those policies, plans, procedures, laws and regulations which would have a significant impact on operations and reports and determine whether the organization is in compliance.
- Review the means of safeguarding assets and, as appropriate, verify the existence of such assets.
- Carry out value for money (VFM) / performance audits to determine the efficiency and effectiveness of services and evaluate attainment of corporate objectives and value to residents.
- Report to those members of management who should be informed or who should take corrective action, the results of audit examinations, the audit opinions formed, and the recommendations made.

- Evaluate any plans or actions taken to correct reported conditions and provide timely follow-up to ensure satisfactory disposition of audit findings in the manner and timeframe committed to by management in the original audit report. If the corrective action is considered unsatisfactory, hold further discussions to achieve acceptable disposition.
- Develop a flexible multi-year work plan, including any risks or control concerns identified by management or other audits as well as appropriate special tasks or projects requested by management.
- Undertake assessments, investigations, or refer issues to other appropriate parties as a result of disclosures under a Whistleblower By-law or the Fraud and Waste Hotline.
- Maintain a professional audit staff with sufficient knowledge, skills and experience.

AUDIT PLANNING

Each term of Council, the City Auditor shall prepare a multi-year work plan, setting out the proposed work for the term.

For the projects, the following sources are considered:

- Prioritization of the audit universe using a risk-based methodology;
- Requests from Members of Council, senior management and staff;
- Any audits planned for the past term but delayed or not completed;
- Any conditions or concerns discovered or communicated throughout past years;
 and
- Input from members of the public, via the Fraud and Waste Hotline and resident complaints or suggestions.
- Areas likely to provide significant payback in terms of increased revenues, reduced costs, operational efficiencies and quality of services.

The multi-year work plan will be presented to the Audit, Finance & Administration Committee for approval. Any changes to the work plan requested by Council or individual Members of Council will require a majority of at least two-thirds the total members of Council present and not prohibited by statute from voting for the City Auditor to consider.

REPORTING

A written report is prepared and issued by the City Auditor following the conclusion of each audit assurance project. The report will include management's responses to the report findings and recommendations. Management's response will include a statement of general agreement or disagreement with the stated findings and recommendations as well as a timeframe for anticipated completion of action to be taken and an explanation for any recommendation not addressed.

The OCA is responsible for appropriate follow up on audit findings and recommendations. All significant findings will remain open until the City Auditor has determined management has appropriately taken action to resolve the finding.

By-law 19-181 (Whistleblower By-law), Section 19 – Responsibility of the City Auditor requires a quarterly report for, in the aggregate, on the number, nature and outcome of disclosures of serious wrongdoing made under this By-law. Quarterly reports known as "Whistleblower, Fraud and Waste Information Updates" are submitted to the Audit, Finance & Administration Committee and capture at a high level the special investigations activity undertaken by the OCA.

Consulting and risk assessment project reports are issued at the discretion of the City Auditor. Considerations for reports include project size and type, topic, risk profile and organizational impact.

All reports are generally presented to the Audit, Finance & Administration Committee. Reports may be presented to the General Issues Committee, if appropriate.

PROFESSIONAL STANDARDS

Under the City Auditor's authority granted in this Office of the City Auditor Charter, all OCA activity will be guided by the following professional standards and/or legislative requirements:

- *Municipal Act*, Sections 223.19-223.23 (Auditor General), as applicable.
- City of Hamilton Whistleblower By-law (19-181).
- International Standards for the Professional Practice of Internal Auditing issued by the Institute of Internal Auditors, for research and general consulting work.
- Certified Fraud Examiners Code of Professional Standards issued by the Association of Certified Fraud Examiners for Fraud, Waste and Whistleblower assessments and investigations.

When working as an Auditor General performing audit assurance engagements, work activities will be guided by professional standards and/or legislative requirements, as applicable.

Audit Services professional staff are also bound by the standards and ethics of their respective professional organizations, which include the Chartered Professional Accountants of Canada (CPA), Institute of Internal Auditors (IIA) and Association of Certified Fraud Examiners (ACFE).

(Approved by Council – September 11, 2019 – Report AUD19005)

Appendix G

ROLES OF COUNCIL, MAYOR AND COMMITTEE CHAIR

PREAMBLE

The general requirements for holding office for elected officials for the City of Hamilton are contained in the *Municipal Act*, 2001.

GENERAL ROLE/GUIDELINES (COUNCIL)

- (a) Within the authority of enabling legislation, to develop policies for the purpose of guiding the administration of municipal government in the City of Hamilton.
- (b) Subject to legislative restrictions, develop regulations to be adopted in By-laws and resolutions for the overall benefit of the citizens of the community.
- (c) Appoint statutory officers and senior officials to ensure that a functional management system is in place and which will administer the City within the adopted policies of Council.
- (d) To collectively oversee the administrative functions as carried out by appointed officials within the delegated authority and the policies adopted by Council.
- (e) To monitor the administrative process and ensure that the will and direction of the City is placed into effect.
- (f) To be prepared to attend regularly scheduled meetings and special meetings of Council and Committees as required and to participate in the debate for the purpose of developing and adopting policies and directions for the City of Hamilton.
- (g) Absences from the meetings of Council for three successive months results in a member's seat being declared vacant unless authorized by resolution of Council.
- (h) To act as liaison between the citizens they represent and the City, to ensure that the intention of established policies and regulations are applied in a manner that is conducive to the interests of the citizens as a whole.
- (i) To oversee the financial affairs and delivery of City services through the adoption of policies and budget control guidelines and to ensure that appropriate audit procedures and monitoring programmes are in effect.

Appendix G

ROLE OF THE MAYOR

(including the Deputy Mayor while fulfilling the duties of Mayor)

The Mayor is responsible to act as the Head of Council, as detailed in the *Municipal Act,* 2001 providing leadership to other Members of Council.

- (a) To act as the Council's corporate representative when dealing with other government agencies and the private sector consistent with the vision and direction expressed by the Council of the day.
- (b) The Mayor and the City Manager must work in close liaison as the pivotal link between the policy-making body of Council and the administrative organization of the City.
- (c) The role of Mayor is considered as statutory and policy-related, to act as the Head of Council and to co-ordinate political representation on behalf of the City when required at meetings, receptions, functions, and community activities, and to direct administrative functions to the attention of the City Manager.

ROLE OF COMMITTEE CHAIR

- (a) To ensure that the general functions noted in the roles of Council and in the policies and procedures established by the City are maintained.
- (b) To ensure that the rules of procedure with respect to conduct are followed.
- (c) The role of Chair, in co-ordinating the meetings of a Committee, will be considered statutory and policy-related, with administrative functions to be delegated to appointed staff and officials through the appropriate General Manager or Medical Officer of Health, following appropriate reporting procedures of the City.
- (d) It is recognized that the role of Chair, as outlined above, does not provide any administrative authority over staff, and that any administrative requests should be brought to the attention of the City Manager, appropriate General Manager or Medical Officer of Health or appropriate delegated staff.

DUTIES OF COMMITTEE CHAIR

- (a) Open the meetings of the Committee by taking the chair and calling the members to order.
- (b) Announce the business before the Committee and the order in which it is to be acted upon.
- (c) Receive and submit, in the proper manner, all motions presented by the members of the Committee.
- (d) Put to a vote all motions, which are moved, or necessarily arise in the course of the proceedings, and to announce the result.

Appendix G

- (e) Enforce the rules of procedure and rule upon all procedural matters.
- (f) Guide the members when engaged in debate in accordance with the rules of procedure.
- (g) Enforce on all occasions the observance of order and decorum among the members and the attending public.
- (h) Order any member persisting in a breach of the rules of procedure to vacate the meeting room.
- (i) Permit questions to be asked through the Chair of any staff in order to provide information to assist any debate when the Chair deems it proper.
- (j) Rule on any points of order or points of privilege without debate or comment.
- (k) Rule whether a motion or proposed amendment is in order.
- (I) Determine which member has the right to speak.
- (m) Ascertain that all members who wish to speak on a motion have spoken and that the members are ready to vote, and shall then put the vote.
- (n) May call a member to order.
- (o) Where it is not possible to maintain order, the Chair may, without any motion being put, adjourn the meeting to a time to be named by the Chair.
- (p) Adjourn the meeting when the business is concluded.
- (q) The Chair may state relevant facts and the Chair's position on any matter before the Committee without leaving the chair, but it shall not be permissible for the Chair to move a motion or debate a question without first leaving the chair.
- (r) If the Chair desires to leave the chair to move a motion, or to take part in the debate, the Chair shall call on the Vice Chair, or in the absence of the Vice Chair, on another member to preside until the Chair resumes the chair.
- (s) The Committee Chair shall vote on any questions before the Committee and in the event of an equality of votes (tie vote) the Committee Chair will not have an extra casting vote and the question being voted upon is deemed lost.

Appendix H

CRITERIA FOR THE CREATION OF ADVISORY COMMITTEES OR TASK FORCES

- (a) That all requests for the formation of an Advisory Committee or Task Force be presented to a Standing Committee for consideration.
- (b) That all requests, upon approved motion by the Standing Committee, be referred to the applicable General Manager for a report back on the following;
 - (i) inventory of previous and existing activities related to the issue
 - (ii) inclusion of a sunset clause, which outlines the time needed to complete the mandate
 - (iii) reporting structure, membership composition, mandate, objectives, legislative requirements, work plan and timelines for the Special Purpose Body
 - (iv) responsibilities of the members
 - (v) membership expertise requirements for the Special Purpose Body
 - (vi) proposed budget allocation requirement and source of funding
 - (vii) staffing requirements
 - (viii) other necessary resources
- (c) The Clerk may invite citizens to serve as members and such invitations shall include the Advisory Committee or Task Force's mandate including any membership expertise requirements and the time needed to complete the mandate;
- (d) Applications received for membership shall be forwarded by the Clerk to the Selection Committee;
- (e) The Selection Committee may consider balanced geographical representation in selecting the members, where appropriate;
- (f) The rules of procedure shall be observed so far as they are applicable;
- (g) A Code of Conduct, setting out general standards for acceptable conduct by Members of Advisory Committees and Task Forces in the performance of their duties, is set out in Appendix I to this By-law.
- (h) All minutes and reports shall be submitted to the Standing Committee to which the Advisory Committee or Task Force reports; and
- (i) Staff may act as resource persons in a non-voting capacity.

Appendix I

HAMILTON ADVISORY COMMITTEE/TASK FORCE CODE OF CONDUCT

Council has adopted this Code of Conduct for the guidance of Appointees to Advisory Committees and Task Forces providing recommendations to Standing Committees and to assist Appointees in performing their duties in a manner which will promote the public's confidence in these Advisory Committees and Task Forces operating with integrity, transparency and courtesy.

It is recognized that the Code of Conduct cannot anticipate all possible fact situations in which Appointees may be called upon to exercise judgement as to the appropriate standard of conduct. When this occurs, Appointees are to ensure that their decisions maintain the Advisory Committee or Task Force's integrity, transparency and courtesy.

This Code of Conduct does not apply to Members of Council who are subject to the Council Code of Conduct.

Failure to comply with this Code of Conduct may result in the Advisory Committee or Task Force:

- (1) requesting an apology from the Appointee; and/or
- (2) removing the Appointee from the Advisory Committee or Task Force for a portion or all of their term.

1. GOOD CONDUCT

Appointees shall act with honesty and integrity including:

- acting in a manner that contributes to the public's confidence in the Advisory Committee or Task Force; and,
- not engaging in conduct that may, or may appear to, constitute an abuse of their position as an Appointees.

2. MEETINGS

Appointees shall maintain proper control over meetings demonstrating respect for everyone who is involved in a proceeding.

Appointees are expected to attend all meetings of the Advisory Committee or Task Force. If an Appointee misses more than three meetings during their term, the Chair, after hearing and considering any explanation provided by the Appointee, may remove the Appointee from the Advisory Committee or Task Force for the remainder of their term.

3. COLLEGIALITY

Appointees shall respect and co-operate with other Appointees and the Advisory Committee or Task Force staff.

4. GIFTS OR BENEFITS

Appointees shall not accept a gift or benefit that may appear as being offered because they are Appointees.

5. CONFIDENTIAL INFORMATION

Appointees shall not disclose to any member of the public any confidential information acquired by virtue of their position.

Page 59 of 66

Appendix I

6. COMMUNICATION

Appointees should accurately communicate a recommendation or direction.

Appointees may provide their own personal opinion on a matter, provided that it is made clear to the party they are speaking to that the comments are their own and are not being made on behalf of the Committee providing it does not relate to In Camera discussions.

Appointees may refer the media or others making inquiries to the Chair, or, in the absence of the Chair, to the Vice Chair.

HEARING PROCEDURES FOR DEVELOPMENT CHARGE COMPLAINTS – SECTION 20

COMMITTEE COMPOSITION

The Audit, Finance & Administration Committee has delegated authority to hold hearings for Section 20 Development Charges complaints. The Audit, Finance & Administration (AF&A) shall be comprised of a minimum of 7 members of Council, plus the Mayor as exofficio.

SECTION 20 COMPLAINT

As Per the *Development Charges Act, 1997,* a person may submit a formal complain to Council respecting one of three issues:

- **20** (1) A person required to pay a development charge, or the person's agent, may complain to the council of the municipality imposing the development charge that,
 - (a) the amount of the development charge was incorrectly determined;
 - (b) whether a credit is available to be used against the development charge, or the amount of the credit or the service with respect to which the credit was given, was incorrectly determined; or
 - (c) there was an error in the application of the development charge by-law. 1997, c. 27, s. 20 (1).

The formal complaint must be submitted within 90 days of the charge being payable (Development Charges Act, 1997) and must be in accordance with the grounds permitted under the Development Charges Act.

(2) A complaint may not be made under subsection (1) later than 90 days after the day the development charge, or any part of it, is payable. 1997, c. 27, s. 20 (2).

The formal complaint must be in writing, stating the complainant's name, address where notice can be given, and the reason for the complaint.

DEVELOPMENT CHARGES COMPLAINT HEARING - SECTION 20 COMPLAINT

A hearing (or tribunal) must be held within 60 days of receiving the complaint (Section 22(2)). A Hearing Notice will be mailed to the complainant at least 14 days before the hearing.

The Complaint Hearing is governed by the procedures set out in the Council Procedure By-law and the *Statutory Powers Procedure Act* ("SPPA").

The Audit, Finance & Administration Committee, in conducting the hearing of a Development Charges Complaint, is required to only consider the grounds permitted under subsection 20(1) of the *Development Charges Act*.

The grounds set out in the *Development Charges Act* are very specific and quite focused. They do not include a request to be exempt from a Development Charge otherwise applicable pursuant to a Development Charge By-law nor do they include a request to create a new category of development not found in a Development Charge By-law. A Development Charge Complaint cannot be used to amend the Development Charge By-law to alter the Development Charge rate otherwise validly applicable or to add a credit or exemption not already within the Development Charge By-law.

1. Party Representatives

There are two parties to the hearing of a Development Charge Complaint, the City and the Complainant. The City's Finance Division is responsible for administering the Development Charge By-law and a solicitor from the Legal Division will represent the City at the hearing. The Complainant may represent themselves or have legal representation as well.

In addition, the Committee will have a solicitor from the Legal Division to provide legal advice to the Committee during a hearing and deliberations. This practice is permissible provided that the Solicitor does not take part in making findings of fact or in making the ultimate decision on the matter.

2. Live Stream & Electronic Voting

The Development Charge Hearing is not live-streamed when meeting IN-PERSON, unless specifically requested by someone in attendance and then agreed upon by both parties.

Electronic votes are not taken during a Development Charge Hearing.

Subject to Section 4 of this Appendix J, the Development Charge Hearing is live-streamed when meeting VIRTUALLY, in order to have all proceedings open to the public and the parties.

3. Conflict of Interest

If any Committee member feels that they should not be hearing a particular matter, the member may inform the Committee that they will not participate, giving a general or specific reason why not, e.g. one of the witnesses is a relation.

4. Open and Closed Proceedings / Deliberations

All proceedings are to be open to the public and the parties unless one of the exceptions under the SPPA or the *Municipal Act, 2001* applies. Section 9(1) of the SPPA provides that a hearing may be closed to the public if:

- (a) a matter involving public security may be disclosed; or
- (b) intimate financial or personal matters or other matters may be disclosed at the hearing of such a nature, having regard to the circumstances, that the desirability of avoiding disclosure thereof in the interest of any person affected or in the public interest outweighs the desirability of adhering to the principle that hearings be open to the public.

If the Committee believes that such matters could be disclosed, it should ask each of the parties if this is the case. If the Committee is convinced that the harm outweighs the desirability of the hearing being open to the public, the hearing may be closed to the public.

When a hearing is closed to the public under the SPPA, only the parties and their representatives remain in attendance.

The Committee may also rely on the authority under section 239 of the *Municipal Act*, 2001 in closing proceedings to the public or the public and the parties - for example, if it wishes to receive advice subject to solicitor client privilege.

The Committee may, but is not required to, retire to deliberate in the absence of the public and the parties. Deliberations occur when the Committee considers the evidence and submissions in arriving at a decision. The decision itself is announced in the presence of the public and the parties. The authority for retiring to deliberate is found at common law and is referred to as the "confidentiality principle" or the "rule on deliberative secrecy".

5. Adjournments

Adjournments may be requested by either party at the start of or during a hearing. The Committee may grant or refuse an adjournment request in light of a number of considerations including: the legitimate inability of the complainant or a witness to attend or, within reason, the counsel of their choice; or, the necessity for time to prepare before a hearing or to respond to new and unexpected issues or allegations arising in the course of a hearing.

6. Agreed Upon Statements of Fact and Joint Submissions

The parties may submit an agreed upon statement of facts which means that they will inform the Committee that some or all of the facts relevant to the hearing are not in dispute. Agreed upon facts need not be proven and should be accepted by the Committee.

The parties may go beyond an agreed upon statement of facts to make a joint submission, asking for a final decision that is acceptable to both. The Committee must give serious consideration to a joint submission and must not reject it without good cause. While the Committee may reject all or part of a joint submission, if this is being considered, both parties must be given the opportunity to make representations before the final decision is made.

7. Witnesses

The parties may call witnesses during the Hearing. (Section 10.1, SPPA)

At the request of a party or on its own initiative, the Committee may require the attendance of a witness to give evidence by issuing a summons.

A witness who is summoned to testify before the Committee cannot refuse to answer a proper and relevant question and Section 13 of the *Canadian Charter of Rights and Freedoms* and Section 14 of the SPPA protect witnesses when providing answers.

The Committee should be hearing only the relevant evidence of the parties' witnesses.

Witnesses should be sworn or affirmed by the Chair of the Committee prior to commencing their testimony.

Lawyers or representatives acting as an advocate for a complainant cannot be a witness.

8. Evidence

Administrative tribunals are given much more latitude than courts with respect to the evidence which they may receive and consider in arriving at a decision. Accordingly, the Committee may receive hearsay evidence and unsworn evidence. (Section 15, SPPA)

The fundamental test with respect to the admissibility of evidence is that it must be relevant to the issues which are involved in the hearing. Relevance for a Development Charge hearing is determined by reference to the permitted grounds set out in subsection 20(1) of the *Development Charges Act* and by the grounds set out in the complaint, but only if those grounds are permitted grounds.

When the Committee is confronted with an objection to the admissibility of a relevant piece of evidence, the evidence should generally be admitted unless it is clearly irrelevant. The Committee should consider the objection with respect to the weight it gives to that particular evidence when arriving at its decision. The general principle is that indirect evidence (hearsay) should be given less weight than direct evidence such as a witness' own observations, unless there is a valid reason to conclude that the direct evidence is not credible.

The Committee may make a finding of credibility in considering the testimony of a witness – giving little or no weight to testimony it does not find credible. An administrative tribunal may find it very difficult to indicate in a decision that a witness was not credible. It is advisable for the administrative tribunal to fully and clearly explain itself, for example, by stating X's testimony was not relied upon because they admitted to a direct financial interest in the outcome of the hearing and because their answers were influenced by this.

Although unaffirmed or unsworn evidence is admissible, testimony to the Committee should be given under affirmation or oath. Each witness should be affirmed or sworn immediately before giving their testimony. A witness should be asked whether they prefer to be sworn or affirmed.

An unrepresented complainant is acting in two capacities, both as their own representative and as a witness. When they are acting as a witness – for example, telling the Committee what did or did not happen – they should be under affirmation.

The parties may examine their own witnesses and cross-examine other witnesses. (Section 10.1, SPPA) The Committee may also question witnesses. Generally, this should be done after the parties have finished questioning the witness.

The Chair or Vice Chair presiding should allow each party to ask any further questions of the witness they may have arising from questions posed by a member of the Committee.

The onus is on the complainant to satisfy the Committee that the Development Charge amount was incorrectly determined, that a credit should have been issued against the Development Charge, that a credit was incorrectly applied, or there was an error in the application of the Development Charge by-law.

9. Note-taking

Notes taken by Committee members during the Hearing should be kept for a reasonable period of time, enough time for whatever appeal right there may be to be exercised. A problem arises when protection of privacy and freedom of information legislation applies. In the case of the *Municipal Freedom of Information and Protection of Privacy Act*, there is no exception made for the notes of members sitting on a municipal tribunal such as a Development Charge Complaint Hearing. As a consequence, if the City, in this case Clerks, has custody or control of the notes, these notes may be subject to disclosure under MFIPPA. In addition, the notes arguably would be subject to the Records Retention By-law and could only be destroyed in accordance with that By-law.

Consequently, to ensure that adjudicative independence can be maintained, it is recommended that if a Committee member chooses to take notes, they:

- (1) Do so for the purposes of helping:
 - (a) to remember and understand what occurred during a hearing; and
 - (b) to make a decision in respect of a hearing;
- (2) Keep their notes confidential, not allowing any other person to see, read or use the notes for any purpose;
- (3) maintain responsibility for the care and safe-keeping of their notes;
- (4) store their notes at their office or home; and
- (5) destroy their notes after some reasonable period of time such as one year.

10. Decisions

In the context of a Development Charge Complaint hearing, the Committee makes two types of decisions:

- (1) procedural decisions such as adjourning the hearing to another date; and,
- (2) final decisions concerning whether:
 - (a) the amount of the development charge was incorrectly determined;
 - (b) whether a credit is available to be used against the development charge, or the amount of the credit or the service with respect to which the credit was given, was incorrectly determined; or
 - (c) there was an error in the application of the development charge by-law.

The procedural decision or final decision of the majority of the members of the Committee is the procedural decision or final decision of the Hearing, allowing for a vote with dissent. However, it is recommended that the Committee operate on a consensual basis in respect of Development Charge Complaint decisions.

The Committee may retire to deliberate in the absence of the public and the parties. At the conclusion of its deliberations, the Committee gives its procedural decision or final decision on the matter in the presence of the parties and the public.

The Committee is not required to give reasons for a procedural decision that is made with the consent of the parties. The Committee must give oral and/or written reasons for a procedural decision that does not have the consent of both parties.

For a final decision, the Committee must announce it orally in the presence of the parties and the public. Additionally, it must make its decision in writing to the complainant and/or their legal counsel or agent within a reasonable time subsequent to the completion of the hearing.

Subsection 17(1) of the SPPA requires a written decision to include reasons for the decision only if a party to the complaint requests reasons. If reasons are requested the decision should summarize the facts and arguments presented by the parties, the findings of fact made by the Committee. The decision must explain the relationship between the evidence and its conclusions, including why evidence was rejected or given little credibility.

Committee members should refrain from expressing opinions on the merits or strength of a case until after all of the evidence and submissions have been heard and they are giving their decision.

It is important that the Committee provides clear instructions to the Legislative Coordinator about the decisions it makes, however, it is acceptable for the Legislative Coordinator to prepare a draft decision for approval by the Committee.

Page 66 of 66

Appendix J

11. Appeal

The *Development Charges Act* requires notice of the decision of the Hearing to be mailed within 20 days after the day the decision is made.

The decision of the Hearing is subject to appeal by the complainant to the LPAT. An appeal must be made no later 40 days after the day the decision is made. The appeal is an appeal *de novo* which means LPAT is not required to consider the Audit, Finance & Administration Committee's reasons in its decision regarding an appeal.

Summary of the Proposed Revisions

Throughout the by-law:

The terms IN-PERSON / VIRTUAL / HYBRID have been added to refer to the various ways the City of Hamilton could be conducting their meetings.

The term 'regularly scheduled meetings' has been amended to 'scheduled meetings'.

The contents of Appendices J, K and L attached to By-law 18-270, as amended, which were appendices that addressed the holding of Electronic meetings, have been incorporated into the proposed Procedural By-law, in order to keep the reference in the by-law to one area, rather than 4 (four) areas in some cases (i.e. sub-section 2.1 could be found in the by-law as well as Appendices J, K and L.

SECTION 1 - DEFINITIONS

- Added the following definitions, in order to define the new terms when they appear in the by-law:
 - "ELE" means Electronic meeting procedures and applies to Virtual and Hybrid meetings during an emergency and used only when the procedures for Electronic meetings differ from In-Person (INP) procedures.
 - "Hybrid" means a proceeding where some members participate virtually and some members participate in-person at Council and Committee meetings during an emergency.
 - "INP" means In-Person meeting procedures and applies only when the procedures for In-Person meetings differs from Electronic (ELE) meeting procedures.
 - "In-Person" means a proceeding where members participate In-Person at Council and Committee meetings.
 - "Virtual" means a proceeding where members participate online or on a computer screen, rather than In-Person during Council and Committee meetings during an emergency.
- Revised the following definitions to enhance their meanings for better understanding:
 - "Point of Order" means a question raised by a member with respect to any departure from the Procedural By-law or in the practised conduct of Council or Committee business.

"Point of Privilege" means a matter that affects the credibility, reputation, and dignity of a Member individually or that Council or Committee collectively hold, that ensure Council's or a Committee's ability to function freely; a need for assistance, to be excused for illness or a personal emergency.

SECTION 3 - COUNCIL MEETINGS

3.5 Meetings of Council for Emergencies

- Added Section 3.5(5), to ensure that Council, has the ability to meet in an adjacent municipality, when necessary during an emergency:
 - (5) In the case of an emergency, Council may hold its meetings at any convenient location within or outside the municipality, and such meetings may be held as IN-PERSON / VIRTUAL / HYBRID meetings, with the details posted on the City's website.

3.7 Quorum

- Section 3.7(5), has been divided into 2 sections, to address quorum for in IN-PERSON and VIRTUAL / HYBRID meetings:
 - (5) **INP** If Quorum cannot be maintained during an IN-PERSON Council meeting, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the decision to continue the meeting will rest with the Mayor or Deputy Mayor. During the absence of quorum, no decisions may be approved.
 - ELE

 If Quorum cannot be maintained during a VIRTUAL / HYBRID Council meeting, due to a loss of members' electronic connection, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the meeting will be recessed for up to 15 minutes or until the members' electronic connection is resumed, which could result in a delay to the next available business day. the Clerk will ensure that the proper messaging respecting the continuation of a Council meeting to the following day or next available day is on the City's website.

3.10 Communication Items

- Sections 3.10(1), 3.10(2) and 3.10(4) have been amended to reflect 'Communication Items.....'.
- Added Section 3.10(5) to apprise those who send in communication items for inclusion on Council agendas that reference another person or staff member will be redacted as per the relevant legislation:

- (5) Communication Items shall not contain the names of individuals (staff or public) with the exception of the author's name. Any names found in the communication will be redacted pursuant to the obligations of *Municipal Freedom of Information and Protection of Privacy Act* (MFIPPA), this Procedural By-law and other relevant legislation or procedure.
- Section 3.10(10) has been amended to include 'or any member of a body subject to review by the Integrity Commissioner':
 - (10) Communications respecting the conduct of a member(s) of Council or any member of a body subject to review by the Integrity Commissioner will not be placed on a Council Agenda and the author will be advised of the process for filing a complaint with the Integrity Commissioner.

3.11 Order of Business

 Added a note on Item (c), Ceremonial Activities, to address the format of Ceremonial Activities at electronic meetings; and 'Council Communication Updates' as per the addition of Section 3.17 (below):

The Order of Business for scheduled IN-PERSON / VIRTUAL / HYBRID meetings of Council, unless amended by Council in the course of the meeting, shall be as follows:

- (a) Approval of Agenda
- (b) Declarations of Interest
- (c) Ceremonial Activities (**ELE** During a VIRTUAL/HYBRID meeting ceremonial activities shall only be permitted to take place Virtually)
- (d) Approval of Minutes of Previous Meeting
- (e) Communications
- (f) Committee Reports
- (g) Motions
- (h) Notice of Motions
- (i) Statement by Members (non-debatable)
- (j) Council Communication Updates
- (k) Private and Confidential
- (I) By-laws and Confirming By-law
- (m) Adjournment

3.17 Council Communication Updates

- Added Section 3.17, Council Communication Updates, to initiate the preparation and distribution of a package that contains the Communication Updates forwarded to Council in a two-week time period, as follows:
 - (1) All staff Communication Update documents sent to Council and posted on the website will be published by the Clerk on the Council agenda for receipt into the public record.
 - (2) A member may refer any of the items listed within the Council Communication Updates to a Standing Committee by contacting the Clerk and any item so referred shall be placed on the next available agenda of the respective Standing Committee.
- Sections 3.12 (Rules of Debate); 3.13 (Voting Procedures) and 3.14 (Motion Procedures) have been moved to their own section (Section 6), as they apply to both Council and Committees.

3.15 Conflict of Interest Declarations

- Section 3.15(1) has been amended to remove the reference to the Conflict of Interest Act, 1990:
 - (1) Where a member has disclosed a conflict of interest at a meeting, the member, shall file a written statement with the Clerk.

SECTION 4 - COMMITTEE OF THE WHOLE

- Added 'and the Reports of Committees who report directly to Council', as we have and have had several Committees who report directly to Council:
 - **4.1** Council may, by resolution:
 - (a) move into Committee of the Whole to consider, including debate, Standing Committee Reports; Selection Committee Reports and the Reports of Committees who report directly to Council on the Council meeting agenda; and
 - **4.2** During the consideration of Reports a member may lift an item from the Information Section with the approval of a majority of the members present, for the matter:
 - 4.3 An information item that is lifted from a Report and amended, will be given an item number and added to the appropriate section of the Standing Committee Report Selection Committee Report or the Report of Committee who reports directly to Council.

SECTION 5 – COMMITTEE MEETINGS

 The title of Section 5, as well as any references to 'Standing' Committee throughout the section has been removed as this section applies to all Committees, unless the sub-section is specific to only 'Standing' Committees:

5.2 Appointment of Committee Chairs and Vice Chairs

- Section 5.2(4) has been amended to reflect that each Sub-Committee, Advisory Committee and Task Force recommends to Council the appointment of a Chair and Vice-Chair:
 - (4) Each Sub-Committee, Advisory Committee and Task Force shall recommend to Council the appointment of a Chair and Vice Chair. The Chair or Vice-Chair of a Sub-Committee, Advisory Committee or Task Force established by Council may serve for more than one year in a Council term.

5.4 Quorum

- Section 5.4(4) has been amended to reflect the loss of quorum with the exception of sections 5.4(6) and 5.4(7):
 - (4) If a quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of Committee, the Clerk shall record the names of the Members of Council present and the meeting shall stand adjourned until the date of the next scheduled Committee meeting or another meeting may be called in accordance with the provisions of this By-law, except as provided for in 5.4(6) and 5.4(7).
- Section 5.4(5) has been divided into 2 sections, to address quorum for in IN-PERSON and VIRTUAL / HYBRID meetings:
 - (5) **INP** If Quorum cannot be maintained during an IN-PERSON Committee meeting, the Clerk will advise the Chair that quorum is lost and the decision to continue the meeting will rest with the Chair. During the absence of quorum, no decisions may be approved.
 - ELE If Quorum cannot be maintained during a VIRTUAL / HYBRID Standing Committee meeting, due to a loss of members' electronic connection, the Clerk will advise the Chair that quorum is lost and the meeting will be recessed for up to 15 minutes to allow for the member's electronic connection to be restored. If quorum cannot be restored within 15 minutes the meeting will be recessed to the next available business day. The Clerk will ensure that details regarding the time of the continuation of the Standing Committee meeting is on the City's website. Where Quorum during a VIRTUAL/HYBRID

Standing Committee meeting is lost for any other reason the INP procedure shall apply.

- Section 5.4(6), reflects the continuation of a meeting without quorum to address when the Committee has an outside consultant; organization, etc., who is in attendance at meeting where quorum isn't achieved, it permits the Chair to decide to proceed with hearing the scheduled presentation without making any decisions:
 - (6) If quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of Committee, the Chair may decide to proceed informally to hear scheduled presentations. Ratification of any decisions will be postponed to a future meeting.
- Section 5.4(7), reflects the continuation of a meeting without quorum to address when the Committee members who are in attendance at a meeting where quorum isn't achieved, want to talk informally on a matter of interest without making any decisions:
 - (7) If quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of a Committee, the Chair may allow the Committee to discuss items of interest. Ratification of any decisions will be postponed to a future meeting.
- Section 5.5 was added to include a section on the location of Committee meetings:

5.5 Location of Meetings

(1) IN-PERSON / VIRTUAL / HYBRID Standing Committee Meetings shall be held in the Council Chambers at City Hall in the City of Hamilton or at such other place within the City of Hamilton as Council may from time to time determine.

5.8 Committee Reports

- Added 'and other Committees that report directly to Council', as we have and have had several Committees who report directly to Council:
 - (1) Each Standing Committee, the Selection Committee, and other Committees that report directly to Council shall make recommendations to Council.
 - (2) A minimum of 2 days shall pass before the Report of a Standing Committee, the Selection Committee, or other Committee that reports directly to Council is presented to Council to provide adequate opportunity for review. Every effort will be made for Standing Committee and/or Selection Committee Reports to be released at an earlier date.

5.9 Rules of Procedure

- The original Section 5.9(1) has been removed as Section 2.1 (below) addresses the rules of procedure for all Committees:
 - (1) The rules governing the procedure of the Council at Council meetings shall be observed at Committee meetings, so far as they are applicable.
 - **2.1** The rules of procedure shall be observed in all proceedings of Council and shall be the rules for the order and dispatch of business in Council and unless specifically provided, with necessary modifications, apply to all Committees.
- 5.9 (3) was amended at Governance Review Sub-Committee to remove "when a subject matter directly affects their Ward" to permit a member to attend a Standing Committee meeting on any matter of interest:
- (3) A Member of Council who is not a member of a Standing Committee is entitled to attend such Standing Committee meetings and to participate, provided that the Member of Council shall:
 - (a) not be counted for quorum purposes, and;
 - (b) not move any motion or vote on any matter.

5.10 Order of Business

- Section 5.10 has been amended, so that the Committees hear the staff presentations prior to hearing from delegations; and
- Added a note on Item (c), Ceremonial Activities, to address the format of Ceremonial Activities at electronic meetings:
 - (1) The general Order of Business for the IN-PERSON / VIRTUAL / HYBRID scheduled meetings of Committees, unless amended by the Committee in the course of the meeting, shall be as follows:
 - (a) Ceremonial Activities (**ELE** During a VIRTUAL/HYBRID meeting ceremonial activities shall only be permitted to take place Virtually)
 - (b) Approval of Agenda
 - (c) Declarations of Interest
 - (d) Approval of Minutes of Previous Meeting
 - (e) Communications
 - (f) Delegation Requests
 - (g) Consent Items

- (h) Staff Presentations
- (i) Public Hearings/ Delegations
- (j) Discussion Items
- (k) Motions
- (I) Notice of Motions
- (m) General Information/Other Business
- (n) Private and Confidential
- (o) Adjournment

5.12 Delegations

- Section 5.12(1) has been divided into 2 sections, to address delegation for in IN-PERSON and VIRTUAL / HYBRID meetings:
 - (1) **INP** Persons who wish to appear as a delegation at an IN-PERSON meeting will appear in-person to address a Committee
 - **ELE** Persons who wish to appear as a delegation at a VIRTUAL/ HYBRID meeting will appear virtually to address a Committee
- Sections 5.12(11) and 5.12(12) have been included based on the feedback received from the Committee Chairs:
 - (11) Delegates appearing IN-PERSON or VIRTUALLY will adhere to Section 8, Order and Decorum of this by-law.
 - (12) Members of Committees can ask questions of the delegations for clarification purposes only and not for debate.
- The title of Section 5.13, has been amended to reflect 'at Standing Committees':

5.13 Public Hearings at Standing Committees

- Section 5.13(3)(c) has been divided into 2 sections, to address public hearing delegations for in IN-PERSON and VIRTUAL / HYBRID meetings:
 - (3) Public Hearings shall commence with:
 - (a) an introduction of the subject matter by the Chair or by staff;
 - (b) the staff presentation, if any and if not waived by the members of the Standing Committee; and
 - (c) **INP** The signing in of all delegations on a sheet, provided by staff, with their name, address, including postal code, phone number and e-mail address (optional).

- Presentations by delegations who have registered to attend VIRTUALLY, will be heard.
- Section 5.13(5) has been divided into 2 sections, to address public hearing delegations for in IN-PERSON and VIRTUAL / HYBRID meetings:
 - (5) **INP** Notwithstanding subsection 5.13(4), Standing Committee shall hear oral representations from any person in attendance at a Public Hearing, whether or not a request was made in advance.
 - **ELE** Notwithstanding subsection 5.13(4), Standing Committee shall only hear oral representations from persons who have registered to attend VIRTUALLY.
- Section 5.13(6) has been amended to state a 5 minute limit, as there is nothing within the legislation that provides a delegation with unlimited time to speak at a public hearing:
 - (6) IN-PERSON or VIRTUAL delegations made at a Public Hearing shall be limited to an oral presentation of not more than five minutes.
- Section 5.13(7) has been amended to reflect a concern raised by the Governance Review Sub-Committee respecting the number of copies a delegation is to bring to a meeting:
 - (7) Where a delegation wishes to provide written material to become part of the public record to the members of the Standing Committee at a Public Hearing, the delegation shall:
 - (a) comply with subsection 5.12(8); or
 - (b) if after the deadline, provide a copy of the written material to the Clerk at the IN-PERSON meeting for distribution after the meeting; or
 - (c) if after the deadline, provide a copy of the written material to the Clerk via email at a VIRTUAL / HYBRID meeting for distribution after the meeting.

5.14 Communication Items

• Sections 5.14(1), 5.14(2) and 5.14(5) have been amended to reflect 'Communication Items.....'.

- Added Section 5.14(6) to apprise those who send in communication items for inclusion on Council agendas that reference another person or staff member will be redacted as per the relevant legislation or procedure:
 - (5) Communication Items shall not contain the names of individuals (staff or public) with the exception of the author's name. Any names found in the communication will be redacted pursuant to the obligations of *Municipal Freedom of Information and Protection of Privacy Act* (MFIPPA), this Procedural By-law and other relevant legislation or procedure.
- Section 5.14(10) has been amended to include 'or any member of a body subject to review by the Integrity Commissioner':
 - (10) Communications respecting the conduct of a member(s) of Council or any member of a body subject to review by the Integrity Commissioner will not be placed on a Standing Committee Agenda and the author will be advised of the process for filing a complaint with the Integrity Commissioner.

5.15 Reconsideration of a Matter Decided Within a Standing Committee Meeting

- The original Section 5.15(2) (in italics), has been incorporated into Section 5.15(1):
 - (1) At any Standing Committee Meeting, after a matter has been decided by the Standing Committee, including both successful or failed motions, a Member of the Standing Committee who voted in the majority, may present a Motion to reconsider the matter prior to the adjournment of the meeting. The Chair may ask the Member of the Standing Committee to confirm that they voted with the majority on the issue in question.

5.16 Conflict of Interest Declarations

- Section 5.16(1) has been amended to remove the reference to the Conflict of Interest Act, 1990:
 - (1) Where a member has disclosed a conflict of interest at a meeting, the member, shall file a written statement with the Clerk.

SECTION 6- RULES OF DEBATE, VOTING PROCEDURES AND MOTION PROCEDURES

- Sections 6.1 formerly 3.12 (Rules of Debate); 6.2 formerly 3.13 (Voting Procedures) and 6.3 formerly 3.14 (Motion Procedures) have been moved from Section 3 and have formed a new section, as they apply to Council and all Committees.
- Throughout Sections 6.1, 6.2 and 6.3, the wording has been amended to include 'the Mayor and/or the Chair' and 'Council and/or Committee' where it was omitted.

6.2 Voting Procedures

- Sections 6.2(3) and 6.2(9) have been divided into 2 sections, to address voting procedures for in IN-PERSON and VIRTUAL / HYBRID meetings:
 - (3) **INP** When an electronic voting system is available at an IN-PERSON meeting every Member of Council or Committee present and participating in a vote on a motion, shall vote electronically excluding those motions referred to in (i) and (ii) below which shall be by show of hands:
 - (i) procedural (majority vote: refer, defer, withdraw, recess, call the question and adjourn); and
 - (ii) Consent Items.
 - **ELE** When an electronic voting system is available at a VIRTUAL / HYBRID meeting every Member of Council or Committee present and participating in a vote on a motion, shall vote electronically.
 - (9) INP When a Member of Council at an IN-PERSON meeting of Council or Committee does not disclose a pecuniary interest, and does not vote on a Motion, the member will be deemed as having voted in the negative and when an electronic voting system is in use the recusal will be displayed and noted in the proceedings.
 - **ELE** When a Member of Council at a VIRTUAL / HYBRID meeting of Council or Committee does not disclose a pecuniary interest, and does not vote electronically on a Motion, the Mayor/Chair will call on the member to vote 3 (three) times, if the member doesn't respond the member will be noted as being 'Not Present' during the vote.
- Sections 6.2(6) and 6.2(7), have been amended for better understanding:
 - (6) When an electronic vote is taken and a member advises the Mayor or the Chair of a point of privilege with respect to their vote being omitted or in error immediately following and prior to the taking of another vote, the vote will be considered void and a new electronic vote will be retaken immediately.
 - (7) When an electronic vote is taken and a member advises the Mayor or the Chair of a point of privilege with respect to their vote being omitted or in error following the taking of another vote, the member's request will be considered through a vote to reconsider the matter:
 - (i) If a two-thirds majority vote to reconsider is decided in the affirmative, the electronic vote will be retaken immediately;

- (ii) If a two-thirds majority vote to reconsider is decided in the negative, the original electronic vote stands.
- Section 6.2(17) has been included to address the voting rights of members of Council who are representatives on an Advisory Committee, when attending an Advisory Committee meeting, as members of Council have their opportunity to vote on these matters at Standing Committee and Council:
 - (17) Members of Council who are appointed as representatives of an Advisory Committee, do not have voting rights while attending an Advisory Committee meeting.

SECTION 9 - MEETINGS CLOSED TO THE PUBLIC

- Section 9.1 has been amended to reflect the content of Section 239(2) 'or a local board' where it was omitted:
 - **9.1** No meeting of Council or a Committee or part thereof shall be closed to the public unless the subject matter being considered is:
 - (a) the security of the property of the City or a local board;
 - (b) personal matters about an identifiable individual, including City or a local board employees;
 - (c) a proposed or pending acquisition or disposition of land for City or a local board purposes;
 - (d) labour relations or employee negotiations;
 - (e) litigation or potential litigation, including matters before administrative tribunals, affecting the City or a local board;
 - (f) the receiving of advice that is subject to solicitor-client privilege, including communications necessary for that purpose;
 - (g) a matter in respect of which Council or a Committee may hold a closed meeting under an Act other than the *Municipal Act*, 2001;
 - (h) information explicitly supplied in confidence to the City or a local board by Canada, a province or territory or a Crown agency of any of them;
 - (i) a trade secret or scientific, technical, commercial, financial or labour relations information, supplied in confidence to the municipality or local board, which, if disclosed, could reasonably be expected to prejudice significantly the competitive position or interfere significantly with the contractual or other negotiations of a person, group of persons, or organization;
 - (j) a trade secret or scientific, technical, commercial or financial information that belongs to the City or a local board and has monetary value or potential monetary value; or

(k) a position, plan, procedure, criteria or instruction to be applied to any negotiations carried on or to be carried on by or on behalf of the City or a local board.

SECTION 10 - VIRTUAL PARTICIPATION AT COUNCIL AND/OR COMMITTEE MEETINGS

- Section 10, has been amended to reflect 'virtual' participation at Council and/or Committee meetings rather than 'electronic' participation.
- Added Section 11, to provide for virtual meetings during an emergency:

SECTION 11 – VIRTUAL MEETINGS DURING AN EMERGENCY

- 11.1 All VIRTUAL / HYBRID Council and Committee meetings shall be streamed live.
- 11.2 Upon being apprised that a virtual meeting is interrupted and cannot proceed with its live feed, due to loss of connection, sound or video, the meeting will be recessed for up to 15 minutes.
- **11.3** If the live feed of a Virtual meeting cannot be resumed within 15 minutes, the meeting will be considered adjourned. The Council/ Committee meeting will be rescheduled to another day or the next regularly scheduled meeting date.
- 11.4 If the Virtual meeting is interrupted and its live feed cannot be resumed, the names of the Council/Committee members present and any decisions of Council/Committee up to the point in time of the interruption shall be recorded in the minutes of the meeting.

SECTION 12 - ROLE OF CLERK

- Sections 12.1 and 12.2 have been amended to provide for the electronic distribution of agendas:
- **12.1** The Clerk shall make the agendas of Council and Committee meetings available to members of Council and Senior Leadership Team electronically at least 5 days prior to the scheduled meetings.
- **12.2** The Clerk shall make the agendas of Council and Committee meetings available to the media and general public electronically, simultaneously with the distribution set out in subsection 12.1 on the City's website.

Appendices

Appendix A - GENERAL ISSUES COMMITTEE

Added "Upper Levels of Government Funding Matters" to the Mandate; and

• Added "all matters having definite litigation" to Legal Services to the Mandate.

Appendix B, C, D, E and F - OTHER STANDING COMMITTEES

 Added "Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee", under Specific duties shall include:

Appendix J - HEARING PROCEDURES FOR DEVELOPMENT CHARGE COMPLAINTS - SECTION 20

 Appendix J has been added to address the Hearing Procedures for Development Charge Complaints – Section 20

Page 1 of 81

Authority: Item ____, Audit, Finance &

Administration Committee Report 21-0__ (Governance Review Sub-

Committee Report 21-___)
Date: ______, 2021

Bill No.

CITY OF HAMILTON

BY-LAW NO. 21-___

A BY-LAW TO GOVERN THE PROCEEDINGS OF COUNCIL AND COMMITTEES OF COUNCIL

WHEREAS Section 238 of the *Municipal Act, 2001* requires that Council of the City of Hamilton adopt, by By-law, procedures governing the calling, place, and proceedings of meetings; and

WHEREAS, the Council of the City of Hamilton, in adopting Item _____ of Audit, Finance and Administration Committee Report 21-____ on _____, 2021 recommended that By-laws 18-270, 19-090, 19-212,19-308, 20-042, 20-055, 20-103, 20-129, 20-145, 20-146, 20-151, 20-184 be repealed and replaced;

NOW THEREFORE, BE IT RESOLVED, the Council of the City of Hamilton enacts as follows:

INDEX

TITLE	SECTION	PAGE
Definitions	1	3
Application	2	7
Council Meetings	3	7
First Council Meeting	3.1	7
Regular Council Meeting Times	3.2	7
Location of Meetings	3.3	8
Special Meetings of Council	3.4	8
Meetings of Council for Emergencies	3.5	9
Cancellation of a Council Meeting	3.6	9
Quorum	3.7	10
Remedy for Lack of Quorum	3.8	10
Delegations	3.9	11
Communication Items	3.10	11
Order of Business	3.11	12
Bills and By-laws	3.12	12
Editorial and Other Changes	3.13	13
Shareholder and Board of Director Meetings	3.14	14
Conflict of Interest Declarations	3.15	14
Provisional Governance Committee During a Declared Emergency	3.16	14

		Page 2 of 81
Council Communication Updates	3.17	15
Committee of the Whole	4	15
Council may, by resolution:	4.1	15
Standing Committee Meetings	5	15
Standing Committees	5.1	15
Appointment of Standing Committee Chairs and		
Vice Chairs	5.2	16
Standing Committee Membership	5.3	16
Quorum	5.4	17
Location of Meetings	5.5	18
Standing Committee Meeting Times	5.6	18
Cancellation of a Committee Meeting	5.7	19
Committee Reports	5.8	19
Rules of Procedure	5.9	19
Duties of Standing Committees Order of Business	5.10	20
	5.11	20
Delegations Public Hearings at Standing Committees	5.12 5.13	20 22
Communication Items	5.14	23
Reconsideration of a Matter Decided Within a	5.15	24
Standing Committee Meeting	0.10	21
Conflict of Interest Declarations	5.16	24
Special Meetings of a Standing Committee	5.17	25
Change to a Schedules Committee Meeting	5.18	25
Rules of Debate, Voting Procedures and Motion	6	25
Procedures		
Rules of Debate	6.1	26
Relinquishing the Chair	6.1(9)	27
Voting Procedures	6.2	27
Motion Procedures	6.3	29
Notice of Motion	6.3(1)	29
Dispensing with Notice Motion to Withdraw	6.3(2)	29
Motion to William Motion to Amend	6.3(3)	29
Motion to Ameria Motion to Defer	6.3(4)	29 30
Motion to Refer	6.3(5) 6.3(6)	30
Motion to Adjourn	6.3(7)	30
Motion to Call the Question	6.3(8)	31
Motion to Reconsider a Matter Decided by Council	6.3(9)	31
Order of Procedural Motions	6.3(10)	32
Non-Debatable Motions	6.3(11)	32
Motion to Suspend the Rules of Procedure	6.3(12)	32
Motion Beyond Jurisdiction of Council	6.3(13)	32
Establishment of Advisory Committees or Task	7	32
Forces		
Order and Decorum	8	33
Meetings Closed to the Public	9	34

		Page 3 of 81
Virtual Participation at Council and/or Committee Meetings	10	35
Virtual Participation During an Emergency	10.1	35
Virtual Meetings during an Emergency	11	36
Role of Clerk	12	36
General	13	37
Appendices		
General Issues Committee	Appendix A	38
Board of Health	Appendix B	40
Public Works Committee	Appendix C	41
Planning Committee	Appendix D	42
Emergency & Community Services Committee	Appendix E	43
Audit, Finance & Administration Committee	Appendix F	44
Roles of Council, Mayor and Committee Chair	Appendix G	53
Criteria for the Creation of Advisory Committees or Task Forces	Appendix H	56
Hamilton Advisory Committee/Task Force Code of Conduct	Appendix I	57
Hearing Procedures for Development Charge Complaints – Section 20	Appendix J	59

SECTION 1 - DEFINITIONS

1.1 In this By-law,

[&]quot;Act" means the Municipal Act, 2001.

[&]quot;Amend" means to alter or vary the terms of a main motion without materially changing its purpose, and amendment has a corresponding meaning.

[&]quot;Approve" means to support or endorse the action, report or document.

[&]quot;Bill" means a proposed form of By-law.

[&]quot;By-law" means a Bill, in a form approved by Council, passed for the purpose of giving general effect to a previous decision or proceedings of Council.

[&]quot;Chair" means the member presiding at any meeting of Council or Committee.

[&]quot;Challenge" means to appeal a ruling of the Chair.

[&]quot;City" means the geographical area of the City of Hamilton or the municipal corporation as the context requires.

[&]quot;Clerk" means the City Clerk, or their designate.

[&]quot;Closed Meeting" means a meeting, or part of a meeting, which is closed to the public as permitted by the Act, also referred to as an "in-camera meeting".

Page 4 of 81

- "Committee" means a Standing Committee, Sub-Committee, Selection Committee or an Advisory Committee or Task Force established by Council from time to time.
- "Committee of the Whole" means a Committee comprised of all Members of Council that directly reports to Council.
- "Confirming By-law" means a by-law to confirm and ratify the actions of Council.
- "Consent Items" means those items on an agenda of a Standing Committee; Sub-Committee or Advisory Committee that may be approved by Committee without debate.
- "Council" means the City Council.
- "Defer" means to delay consideration of a matter by Council or a Committee until a specified time or event.
- "Deputy Mayor" means a Member of Council who is appointed for a one month period from amongst members of Council on a rotation basis determined by the Clerk at the beginning of the term of Council:
 - (a) to assist the Mayor; and
 - (b) to act from time to time in the place of the Mayor, with:
 - all the rights, powers and authority of the Mayor when the Mayor is absent from the City or the office of Mayor is vacant; and
 - (ii) subject to paragraph (i), those rights, powers and authority of the Mayor that:
 - the Mayor is unable to act for the time being due to illness, incapacity or other reason, including acting in the place of the Mayor at any Council or Committee meeting when the Mayor is absent; or,
 - 2. the Mayor refuses to act:
- "ELE" means Electronic meeting procedures and applies to Virtual and Hybrid meetings during an emergency and used only when the procedures for Electronic meetings differ from In-Person (INP) procedures.
- "Emergency" means a situation or impending situation that constitutes a danger of major proportions that could result in serious harm to persons or substantial damage to property and that is caused by the forces of nature, a disease or other health risk, an accident or an act whether intentional or not.
- **"Ex-officio"** means by virtue of the office or position and the Mayor is an ex officio member of all Standing Committees who, when attending a Standing Committee meeting, shall have full voting privileges and shall be counted for the purposes of the Standing Committee's quorum as such quorum is set out in subsection 5.4 of this By-law.

Page **5** of **81**

- **"Friendly Amendment"** means an amendment made to the motion under debate with the consent of the mover and seconder, and without the requirement for an amending motion to be made.
- "Hybrid" means a proceeding where some members participate virtually and some members participate in-person at Council and Committee meetings during an emergency.
- "INP" means In-Person meeting procedures and applies only when the procedures for In-Person meetings differs from Electronic (ELE) meeting procedures.
- "In-Person" means a proceeding where members participate In-Person at Council and Committee meetings.
- **"Majority"** for the purposes of voting, means more than half the total of members of Council or Committee present and not prohibited by statute from voting.
- "Mayor" means the Head of the Council and in this By-law means the Deputy Mayor when the Deputy Mayor is acting in place and stead of the Mayor.
- "Meeting" means any scheduled, special or other meeting of a Council, local board or of a committee of either of them, where, a quorum of members is present, and members discuss or otherwise deal with any matter in a way that materially advances the business or decision-making of the Council, local board or committee.
- "Member of Council" is a person who has been elected to City Council.
- **"Motion"** means a proposal by a member for the consideration of Council or a Committee included in the published agenda, that is moved by a member and seconded by another member.
- "Notice of Motion" means advanced, written notice, given by a member and received by the Clerk, advising Council or Committee that a Motion will be brought forward at a future Meeting of Council or Committee that may be included in the published agenda or in the addendum.
- **"Point of Order"** means a question <u>raised</u> by a member with respect to any departure from the Procedural By-law or in the practised conduct of Council or Committee business.
- "Point of Privilege" means a question by a member with respect to the rights matter that affects the credibility, reputation, and dignity of a Member individually or that Council or Committee collectively hold, that ensure Council's or a Committee's ability to function freely; a need for assistance, to be excused for illness or a personal emergency.
- "Public Hearing" means a Meeting or portion of a Meeting held pursuant to the requirements of a statute, regulation or by-law to hold a public meeting to hear representations on a matter from any person who attends the public meeting.

Page 6 of 81

- "Question" means an issue before a meeting on which a decision has to be made. A question cannot be debated, amended or voted on until it has been proposed as a motion.
- "Quorum" means the number of members required to be present at a meeting to validate the transactions of its business.
- "Receive" means to acknowledge the receipt of information and to include it in the official record.
- "Recess" means a short intermission in a meeting's proceedings, which does not close the meeting, and after which business will immediately be resumed at exactly the point where it was stopped.
- "Recorded Vote" means a written record of the name and vote of each member of Council or a Committee who votes on a question and of each member present who does not vote.
- "Refer" means to send a matter currently under consideration to an individual; a department; a Committee or Council for their consideration.
- "Rules of Procedure" means the rules and procedures set out in this By-law, for the calling, place and proceedings of the meetings of Council and its Committees.
- "Selection Committee" means a Committee established by Council, comprised entirely of Members of Council, to interview and report back to Council on the appointment of citizen representatives to agencies, boards and Committees, and reports directly to Council.
- **"Senior Leadership Team"** means the City Manager and General Managers appointed by Council and anyone designated by the City Manager.
- **"Standing Committee"** means a Committee established by Council, comprised entirely of Members of Council, to carry out duties on an ongoing basis, as specified by Council, and that reports directly to Council.
- "Time Sensitive" means useful for a specific amount of time only or has to be used or completed by a specific time.
- "Virtual" means a proceeding where members participate online or on a computer screen, rather than In-Person during Council and Committee meetings during an emergency.
- "Website" means the official City website.

Page **7** of **81**

SECTION 2 - APPLICATION

- 2.1 The rules of procedure shall be observed in all <u>IN-PERSON / VIRTUAL / HYBRID</u> proceedings of Council and shall be the rules for the order and dispatch of business in Council and unless specifically provided, with necessary modifications, apply to all Committees.
- 2.2 All matters relating to the <u>IN-PERSON / VIRTUAL / HYBRID</u> proceedings of Council and Committees for which rules have not been provided for in this By-law and its Appendices shall be decided, as far as is reasonably practicable, with reference to the parliamentary rules as contained in the most recent edition of *Bourinot's Rules of Order*, Geoffrey H. Stanford, Fourth Edition.

SECTION 3 – COUNCIL MEETINGS

3.1 First Council Meeting

- (1) The First Meeting of Council shall be held on the first Monday of December in an election year at 5:00 p.m.
- (2) The IN-PERSON / VIRTUAL / HYBRID Proceedings of the First Meeting of Council shall include:
 - (a) Ceremonial special address by guests;
 - (b) Filing of Declarations of Office in accordance with the provisions of the Act;
 - (c) Council appointments to Committees, boards and agencies; and
 - (d) Confirming By-law.

3.2 Regular Council Meeting Times

- (1) Unless otherwise decided by Council, Regular Council meetings shall be held:
 - in January:
 on the fourth Wednesday of the month, commencing at 9:30 a.m.
 unless such day is a public or civic holiday, in which case Council shall
 set an alternate day and time;
 - (b) in February, April, May, June, September, October and November: every second and fourth Wednesday of each month, commencing at 9:30 a.m., unless such day is a public or civic holiday, in which case Council shall set an alternate day and time;
 - (c) in March:
 on the fourth Wednesday of the month, commencing at 9:30 a.m.
 unless such day is a public or civic holiday, in which case Council shall
 set an alternate day and time;
 - in July and August:
 on a Friday of each month (one meeting per month), commencing at 9:30 a.m., unless such a day is a public or civic holiday, in which case Council shall set an alternate day and time;
 - (e) in December:

Page 8 of 81

on the second Wednesday of the month, commencing a 9:30 a.m., unless such a day is a public or civic holiday, in which case Council shall set an alternative day and time; or

- (f) in accordance with the schedule approved by Council.
- (2) That, from time to time, there may be a need for Special Council Meeting at 5:00 p.m. to accommodate a Ceremonial Activity where the individuals being recognized are unable to attend a meeting during the day; and
- (3) Advance notice of meetings for the public's information may be found by accessing the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

3.3 Location of Meetings

- (1) IN-PERSON / VIRTUAL / HYBRID Meetings of Council shall be held in the Council Chambers at City Hall in the City of Hamilton or at such other place within the City of Hamilton as Council may from time to time determine.
- (2) When necessary, Meetings of Council may be held in an adjacent municipality (Halton Region; Niagara Region or Haldimand County), with the details posted on the City's website.

3.4 Special Meetings of Council

- (1) In addition to scheduled Council meetings, the Mayor may, at any time, summon a special meeting of Council by giving written direction to the Clerk stating the date, time, and purpose of the special meeting.
- (2) The Clerk shall summon a special meeting of Council when requested to do so in writing by a majority of Members of Council.
- (3) The Clerk shall give each Member of Council, or their designated staff, notice of a special meeting of Council at least 2 days before the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (4) The written notice to be given under subsection 3.4(3) shall state the nature of the business to be considered at the special meeting of Council and no business other than that which is stated in the notice shall be considered at such meeting.

Page 9 of 81

- (5) Notwithstanding subsection 3.4(4), other business may be considered at a special meeting with at least two-thirds of the Members of Council present and voting in the affirmative.
- (6) The Clerk shall prepare the Order of Business for a special meeting of Council with the appropriate selection of agenda categories from subsection 3.11.

3.5 Meetings of Council for Emergencies

- (1) Despite any other provisions of this By-law, a meeting of Council for an emergency may be called by the Mayor, without written notice, to deal with an emergency, provided that an attempt has been made by the Clerk to notify Members of Council about the meeting as soon as possible and in the most expedient manner available.
- (2) The only business to be dealt with at a meeting of Council for an emergency shall be business dealing directly with the emergency.
- (3) The lack of receipt of a notice of, or an agenda for, a meeting of Council for an emergency by any Member of Council shall not affect the validity of the meeting or any action taken at such meeting.
- (4) The Clerk shall prepare the Order of Business for an emergency meeting of Council with the appropriate selection of agenda categories from subsection 3.11.
- (5) In the case of an emergency, Council may hold its meetings at any convenient location within or outside the municipality, and such meetings may be held as IN-PERSON / VIRTUAL/ HYBRID meetings, with the details posted on the City's website.

3.6 Cancellation of a Council Meeting

- (1) The Clerk may cancel a Council meeting in consultation with the Mayor or the Deputy Mayor in the Mayor's absence.
- (2) The Clerk shall give each Member of Council, or their designated staff, notice of a cancellation of a meeting of Council at least 2 days prior to the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (3) The written notice to be given under subsection 3.6(2) shall state the reason for the cancellation of the meeting.

Page **10** of **81**

(4) The Clerk shall ensure that the meeting cancellation information is posted to the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

3.7 Quorum

- (1) As soon as there is a quorum after the time set for the meeting, the Mayor shall take the chair and call the meeting to order.
- A majority of Members of Council is necessary to constitute a quorum of Council.
- (3) If the Mayor does not attend within 15 minutes of the time appointed for a Council meeting and a quorum of Council is present, the Deputy Mayor shall take the chair and call the meeting to order and will preside until the arrival of the Mayor.
- (4) If a quorum is not achieved within 30 minutes after the time appointed for a meeting of Council, the Clerk shall record the names of the Members of Council present and the meeting shall stand adjourned until the date of the next scheduled Council meeting or other meeting called in accordance with the provisions of this By-law.
- (5) INP If Quorum cannot be maintained during an IN-PERSON Council meeting, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the decision to continue the meeting will rest with the Mayor or Deputy Mayor. During the absence of quorum, no decisions may be approved.
 - ELE If Quorum cannot be maintained during a VIRTUAL / HYBRID Council meeting, due to a loss of members' electronic connection, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the meeting will be recessed for up to 15 minutes to allow for the member's electronic connection to be restored. If quorum cannot be restored within 15 minutes the meeting will be recessed to the next available business day. The Clerk will ensure that details regarding the time of the continuation of the Council meeting is on the City's website. Where Quorum during a VIRTUAL/HYBRID Council meeting is lost for any other reason the INP procedure shall apply.

3.8 Remedy for Lack of Quorum

Municipal Conflict of Interest Act

Where the number of Members of Council who, by reason of provisions of the *Municipal Conflict of Interest Act*, are disabled from participating in a meeting is such that at that meeting the remaining Members of Council are not of sufficient number to constitute a quorum, then, the remaining number of Members of Council shall be deemed to constitute a quorum, provided such number is not less than 2.

3.9 Delegations

Delegations are not permitted at meetings of Council.

3.10 Communication Items

- (1) <u>Communication Items</u> shall be addressed to the Mayor and Members of Council or the City Clerk and shall be presented to Council for consideration.
- (2) <u>Communication Items</u> shall be forwarded to the Office of the City Clerk no later than 12 Noon on the Wednesday the week prior to the City Council meeting for inclusion in the printed Council Agenda.
- (3) Communication Items not received for inclusion in the Agenda package, shall be introduced under Changes to the Council Agenda, provided that they are received in the Office of the City Clerk no later than 12 Noon on the business day prior to the Council meeting.
- (4) <u>Communication Items</u> shall not be anonymous or contain any defamatory, obscene, offensive, indecent, improper, rude or vulgar language.
- (5) Communication Items shall not contain the names of individuals (staff or public) with the exception of the author's name. Any names found in the communication will be redacted pursuant to the obligations of Municipal Freedom of Information and Protection of Privacy Act (MFIPPA), this Procedural By-law and other relevant legislation or procedure.
- (6) Debate on Council communications shall be limited to the recommended disposition, or limited to reason for or against the referral of the matter to an appropriate Standing Committee.
- (67) Resolutions from other municipalities requesting support will be included on the Council Agenda to be received, unless otherwise directed by a Member of Council.
- (8) Receipt of resolutions from other municipalities by City Council shall not be construed as endorsement by the City of any recommendations or actions contained therein.
- (9) Communications respecting labour relation matters, conduct of a member(s) of staff and matters before another body (i.e. Police Services Board) will not be placed on the Council Agenda and will be forwarded forthwith by the City Clerk to the relevant City Department or body for consideration and appropriate action.
- (10) Communications respecting the conduct of a member(s) of Council or any member of a body subject to review by the Integrity Commissioner will not be placed on a Council Agenda and the author will be advised of the process for filing a complaint with the Integrity Commissioner.

Page 12 of 81

(11) Notwithstanding subsections 3.10(9) and 3.10(10) Council may add any communication item to a Council Agenda.

3.11 Order of Business

The Order of Business for the regularscheduled IN-PERSON / VIRTUAL / HYBRID meetings of Council, unless changedamended by Council in the course of the meeting, shall be as follows:

- (a) Approval of Agenda
- (b) Declarations of Interest
- (c) Ceremonial Activities (ELE During a VIRTUAL/HYBRID meeting ceremonial activities shall only be permitted to take place Virtually)
- (d) Approval of Minutes of Previous Meeting
- (e) Communications
- (f) Committee Reports
- (g) Motions
- (h) Notice of Motions
- (i) Statement by Members (non-debatable)
- (i) Council Communication Package
- (k) Private and Confidential
- (kl) By-laws and Confirming By-law
- (Im) Adjournment

3.12 Rules of Debate

- (1) No Member of Council shall be deemed to have precedence or seniority over any other Member of Council.
- (2) Before speaking to a question or motion, a Member of Council shall be recognized by the Mayor and be placed on the speakers' list.
- (3) When a Member of Council is speaking, no other Member of Council shall pass between that Member of Council and the Mayor or interrupt that Member of Council except to raise a point of order or privilege.
- (4) A Member of Council may require the question or motion under discussion to be read at any time during the debate but not so as to interrupt a Member of Council who is speaking.
- (5) A Member of Council may not speak on any matter other than the matter being debated.
- (6) All Members of Council are restricted to a five minute time limit when asking a question and/or making a statement during Committee and Council meetings.

Page 13 of 81

- (7) Notwithstanding subsection 3.12(6), an extension of a specific allotment of time to the five minute speaking restriction may be granted with the approval of a majority of the members of the Committee/Council present.
- (8) The following matters may be introduced orally without written notice and without leave:
 - (a) a point of order
 - (b) a point of privilege
 - (c) a motion to adjourn
 - (d) a motion to call the question
 - (e) a motion to go into committee of the whole
 - (f) a motion to recess
 - (g) a motion to refer
 - (h) a motion to defer
 - (j) a motion to go into a closed meeting
 - (k) a motion to suspend the rules of procedure
 - (I) a motion to lift

(9) Relinquishing the Chair

The Mayor may designate another Member of Council as acting Chair during any part of a meeting of the Council and while presiding, such Member of Council shall have all the powers of the Mayor with respect to chairing the meeting.

3.13 Voting Procedures (electronic voting is effective January 2019)

- (1) Once a motion is moved and seconded it shall be deemed to be in the possession of Council or Committee.
- (2) Where a Member of Council present at a meeting of Council or Committee is prohibited from voting by statute, the Clerk shall record the name of the Member of Council and the reason that they are prohibited from voting.
- (3) When an electronic voting system is available every Member of Council or Committee present and participating in a vote on a motion, excluding those motions referred to in (i) and (ii) below, shall vote electronically:
 - (i) procedural (majority vote: refer, defer, withdraw, recess, call the question and adjourn); and
 - (ii) Consent Items.
- (4) When an electronic voting system is used for a vote, the results will be displayed and recorded in the proceedings.
- (5) When an electronic voting system is not available every Member of Council or Committee present shall vote by a show of hands.

Page **14** of **81**

- (6) When an electronic vote is taken and a member advises the Chair of a point of privilege immediately following and prior to the taking of another vote, with respect to their vote being omitted or in error, the vote will be considered void and a new electronic vote will be retaken immediately.
- (7) When an electronic vote is taken and a member advises the Chair of a point of privilege following the taking of another vote, with respect to their vote being omitted or in error, the member's request will be considered through a vote to reconsider the matter:
 - (i) If a two-thirds majority vote to reconsider is decided in the affirmative, the electronic vote will be retaken immediately:
 - (ii) If a two-thirds majority vote to reconsider is decided in the negative, the original electronic vote stands.
- (8) When a vote is taken and a tie results, a question is deemed to be lost.
- (9) When a Member of Council at a meeting of Council or Committee does not disclose a pecuniary interest, and does not vote on a Motion, the member will be deemed as having voted in the negative and when an electronic voting system is in use the recusal will be displayed and noted in the proceedings.
- (10) When an electronic voting system is not available and a recorded vote is requested by a Member of Council or Committee immediately before or after a question is put, the Clerk shall record the name and vote of every Member of Council on the question.
- (11) When the Mayor or Chair calls for a vote on a question, each Member of Council or Committee shall occupy their seat and shall remain in their seat until the result of the vote has been declared by the Mayor or Chair, and during such time no Member of Council shall walk across the room or speak to any other Member of Council or make any noise or disturbance.
- (12) After a question is put by the Mayor or Chair, no Member of Council or Committee shall speak to the question nor shall any other motion be made until after the vote is taken and the result has been declared.
- (13) No vote shall be taken by ballot or by any other method of secret voting and every vote so taken is of no effect, except where permitted by statute.
- (14) Despite subsection 3.13(13), a meeting may be closed to the public during a vote if:
 - (a) the vote relates to an issue permitting the meeting to be closed to the public in accordance with section 8 (Meetings Closed to the Public) of this By-law; and
 - (b) the vote is for a procedural matter or for giving directions or instructions to officers, employees or agents of the City or persons retained by or under a contract with the City.
- (15) The Mayor or Chair shall, upon request of a Member of Council or Committee, divide a question, if the question under consideration contains distinct propositions, and the vote upon each proposal shall be taken separately.

(16) If a Member of Council or Committee disagrees with the announcement of the Mayor or Chair that a question is Carried or Lost on a vote by show of hands, they may immediately after the declaration by the Mayor or the Chair, object to the Mayor's or the Chair's declaration and require that a recorded vote be taken.

3.14 Motions Procedures

(1) Notice of Motion

- (a) Notice of all new motions, except motions listed in subsection 3.12(8) shall be given in writing:
 - (i) at a meeting of Council or Committee, but shall not be debated until the next meeting of Council or Committee; or
 - (ii) delivered to the Clerk at any time before noon on the last business day preceding the date of the meeting at which the notice of motion is to be introduced.
- (b) Where a Member of Council's or Committee's motion has been called by the Mayor or Chair at the subsequent meeting and not proceeded with, it shall be dropped from the agenda unless Council or Committee decides otherwise.
- (c) Where Council or Committee has determined not to drop a motion from the agenda, and at the second meeting such motion is called by the Mayor or Chair and not proceeded with, it shall be deemed to have been withdrawn.

(2) Dispensing with Notice

Where a matter is time sensitive or is an emergency, the Motion may be introduced without notice if Council, without debate, dispenses with notice on the affirmative vote of two-thirds of the Members of Council or Committee present and voting.

(3) Motion to Withdraw

After a motion is moved and seconded it shall be deemed to be in possession of Council, but may be withdrawn by the mover at any time before decision or amendment.

(4) Motion to Amend

- (a) A motion to amend:
 - (i) shall be presented in writing, duly moved and seconded;
 - (ii) shall be made only to a previous question or to amend an amendment to the question;

Page **16** of **81**

- (iii) shall be relevant to the question to be decided; shall not be received if in essence it constitutes a rejection of the main question:
- (iv) shall be put to a vote in the reverse order to that which it is moved.
- (b) If the amendment is not considered a "Friendly Amendment", then the motion to amend:
 - If more than one, will be put to a vote in the reverse order to that in which they were moved;
 - (ii) Will be decided upon or withdrawn before the main motion is put to the vote:
 - (iii) Will not be amended more than once, provided that further amendments may be made to the main motion; and,
 - (iv) Will not be directly contrary to the main motion.

(5) Motion to Defer

A motion to defer a question:

- (a) is debatable; and
- (b) shall include the date when the question is to be considered.

(6) Motion to Refer

- (a) A motion to refer the question:
 - (i) shall include the name of the individual or body to whom the question is to be referred;
 - (ii) may include instructions respecting the terms upon which the question is to be referred;
- (b) A motion to refer a question is not debatable except where instructions are included, in which case, only the instructions shall be debatable.
- (c) A motion to refer a question may be amended in accordance with subsection 3.14(4).
- (d) When a motion to refer a question to Council is defeated by a Standing Committee, the Chair of the Standing Committee shall call the question on the matter.
- (e) A member who has spoken to the motion under debate shall not be permitted to move a motion to refer a question.
- (f) A motion to refer a question is only in order after all first time speakers have been heard.

(7) Motion to Adjourn

A motion to adjourn a Council meeting shall always be in order except:

Page 17 of 81

- (a) when a Member of Council is speaking or during the taking of a vote;
- (b) immediately following the affirmative resolution of a motion that a vote on the question now be taken;
- (c) when a Member of Council has already indicated to the Mayor, that they desire to speak on the question;
- (d) when decided in the negative, cannot be made again until Council has conducted further proceedings.

(8) Motion to Call the Question

A motion to call the question shall be in order except when a Member of Council is speaking or before the first-time speakers' list is exhausted.

(9) Motion to Reconsider a Matter Decided by Council

- (a) At any Council Meeting, after a matter has been decided by Council, including both successful or failed motions, a Member of Council who voted in the majority, may present a Notice of Motion to reconsider the matter. The Chair may ask the Member of Council to confirm that they voted with the majority on the issue in question. Such Notice of Motion shall be referred to the next scheduled meeting of Council and shall be included on the agenda under "Motions".
- (b) No decided matter shall be reconsidered more than once during the term of Council.
- (c) A motion to reconsider a decided matter shall require the approval of at least two-thirds of Council present.
- (d) It is not in order to reconsider a decided matter when the motion has been implemented, resulting in a legally binding commitment that is in place on the date that a motion to reconsider is to be debated.
- (e) No motion to reconsider may, itself, be the subject of a motion to reconsider.
- (f) Debate on a motion to reconsider shall be confined to reasons for or against the reconsideration.
- (g) The following motions cannot be reconsidered:
 - (i) to adjourn;
 - (ii) to recess; and
 - (iii) to suspend the rules of procedure.
- (h) Subsections 3.14(9)(a) to 3.14(9)(f) do not apply when:
 - (i) a motion pertains to a decision of a previous Council;
 - (ii) a motion that, while pertaining to a previously decided motion, does not alter the core purpose or intent of the previously decided motion;
 - (iii) a motion presented in respect of a previously decided matter for the purpose of strengthening or clarifying the original decision; or

Page 18 of 81

(iv) a motion presented in respect of a previously decided matter for the purpose of moving a matter through a series of necessary steps or phases.

(10) Order of Procedural Motions

The following motions are procedurally in order in the following sequence:

- (a) to adjourn;
- (b) to recess;
- (c) to request information;
- (d) to call the question;
- (e) to defer;
- (f) to refer;
- (g) to amend.

(11) Non-Debatable Motions

The following motions/actions are not debatable:

- (a) a motion to adjourn;
- (b) a motion to refer (except where instructions are included, in which case, only the instructions shall be debatable);
- (c) a motion to suspend the rules of procedure;
- (d) a point of order; and
- (e) a point of privilege.

(12) Motion to Suspend the Rules of Procedure

The rules governing the procedure of the Council at Council meetings may be suspended and shall require the approval of at least two-thirds of members present.

(13) Motions Beyond Jurisdiction of Council

A motion in respect of a matter which is beyond the jurisdiction of Council shall not be in order.

3.153.12 Bills and By-laws

- (1) All Bills coming before Council shall be in typewritten form and shall contain no blanks except as may be required to conform to accepted procedure or to comply with provisions of any statute or regulation and shall be complete with the exception of the number and date.
- (2) All Bills coming before Council shall receive one reading, unless otherwise requiring more than one reading, pursuant to a Provincial or Federal statute or regulation.

Page 19 of 81

- (3) Every Bill shall be introduced upon motion by a Member of Council, and any number of Bills may be introduced together in one motion, but Council may at the request of a Member of Council, deal separately with any Bill. If Council deals separately with a Bill:
 - (a) all amendments or debate on the content of the Bill shall be held in committee of the whole;
 - (b) Council will move into committee of the whole for discussion on any Bill following a successful verbal motion to go into committee of the whole;
 - (c) all amendments made in committee of the whole shall be reported by the Mayor to Council, which shall receive the same forthwith, and after the report has been received, a vote on the motion to pass the Bill shall be taken:
 - (d) every By-law passed by Council shall be numbered and dated and shall be sealed with the seal of the municipality and signed by the Mayor and Clerk.
- (4) No Bill or By-law may be defeated on the floor of Council without reconsideration of the original motion which gave rise to the consideration of the Bill or By-law.

3.4613 Editorial and Other Changes

- (1) The Clerk may make the following changes to By-laws or resolutions to:
 - (a) Correct spelling, punctuation or grammatical errors, or errors that are of a clerical, typographical or similar nature.
 - (b) Alter the style or presentation of text or graphics to improve electronic or print presentation.
 - (c) Replace a description of a date or time with the actual date or time.
 - (d) If a provision provides that it is contingent on the occurrence of a future event and the event occurs, remove text referring to the contingency and make any other changes that are required as a result.
 - (e) When the name, title, location or address of a body, office, person, place or thing has been altered, change any reference to the name, title, location or address to reflect any alteration in name, title, location or address.
 - (f) Correct errors in the numbering of provisions or other portions of a By-law and make any changes in cross-references that are required as a result.
 - (g) If a provision of a transitional nature is contained in a By-law, make any changes that are required as a result.
 - (h) Make a correction, if it is patent both that an error has been made and what the correction should be.

Page 20 of 81

(2) The Clerk:

- (a) may provide notice of the changes made under subsections 3.4613(1)(a) to 3.4613(1)(c) inclusive in the manner that they consider appropriate.
- (b) shall provide notice of the changes made under subsections 3.4613(1)(d) to 3.4613(1)(h) inclusive in the manner that they consider appropriate.
- (c) in determining whether to provide notice under subsection 3.4613(2)(a), shall consider:
 - (i) the nature of the change; and
 - (ii) the extent to which notice, and the information provided in it, would provide assistance in understanding the relevant legislative history.
- (d) in providing notice under subsection 3.4613(2)(a) or 3.4613(2)(b), shall state the change or the nature of the change.
- (3) No legal significance shall be inferred from the timing of the exercise of a power under this section.
- (4) Regardless of when a change is made to a By-law under this section, the change may be read into the By-law as of the date it was enacted if it is appropriate to do so.

3.4714 Shareholder and Board of Director Meetings

Where Council acts as the Shareholder or the Board of Directors, or both, of a corporation:

- (a) Council shall meet as the Shareholder or the Board of Directors as necessary to complete such business as would normally be completed by the Shareholder or Board of Directors and as required by the corporation's by-laws and resolutions and any laws that govern the corporation; and
- (b) the Mayor and Clerk are authorized, upon approval by Council, to sign any necessary resolutions.

3.1815 Conflict of Interest Declarations (this section is effective March 1, 2019)

- (1) Where a member, has disclosed an a conflict of interest referred to in Section 5 of the Municipal Conflict of Interest Λet, 1990 at a meeting, the member, shall file a written statement with the Clerk.
- (2) Where a member, has disclosed an interest and filed a written statement with the Clerk, as required by subsection 3.4815(1), the Clerk, shall:
 - (a) file the written statement by the member in the Conflict of Interest Registry;
 - (b) record the interest in the minutes of the meeting;
 - (c) file a record of the interest recorded in the minutes of the meeting in the Conflict of Interest Registry; and
 - (d) make the Conflict of Interest Registry available for public inspection.

Page 21 of 81

3.1916 Provisional Governance Committee During a Declared Emergency

- (1) The Provisional Governance Committee may carry out the duties and responsibilities of the Council only at such times when, to the satisfaction of at least 5 Members of Council attending a properly scheduled meeting of Council, at least 8 Members of Council are not able to attend such meeting due to death, illness or circumstances attributable to an emergency which has been declared by the Mayor or their designate in accordance with the City's Emergency Plan.
- (2) The Provisional Governance Committee shall be comprised of eight Members of Council, and shall elect its own Chair from amongst the attending Members.
- (3) The Provisional Governance Committee shall have all the powers and duties that may legally be delegated to the <u>Provisional Governance</u> Committee pursuant to the Act, or any other applicable Provincial or Federal legislation, subject to the following conditions and restrictions:
 - (a) the delegation shall be effective only during the times specified; and
 - (b) the <u>Provisional Governance</u> Committee shall conduct its meetings in accordance with this By-law, providing that the quorum for the constitution of a valid meeting of the <u>Provisional Governance</u> Committee shall be 5 Members of Council.

3.17 Council Communication Updates

- All staff Communication Update documents sent to Council and posted on the website will be published by the Clerk on the Council agenda for receipt into the public record.
- (2) A member may refer any of the items listed within the Council Communication

 Updates to a Standing Committee by contacting the Clerk and any item so
 referred shall be placed on the next available agenda of the respective

 Standing Committee.

SECTION 4 - COMMITTEE OF THE WHOLE

- **4.1** Council may, by resolution:
 - (a) move into Committee of the Whole to consider, including debate, Standing Committee Reports and: Selection Committee Reports and the Reports of Committees who report directly to Council on the Council meeting agenda; and
 - (b) move out of Committee of the Whole to report to Council.
- 4.2 During the consideration of Standing Committee Reports and Selection Committee Reports, a member may lift an item from the Information Section with the approval of a majority of the members present, for the matter:
 - (a) to be voted upon by Council; or
 - (b) to be debated and/or amended by Council.

Commented [PJ2]: Added Standing and Selection to be consistent with the other sections within the by-law.

Page 22 of 81

4.3 An information item that is lifted from a <u>Standing Committee</u> Reports <u>and Selection Committee Reports</u>, and amended, will be given an item number and added to the appropriate section of the Report.

Commented [PJ3]: Added Standing and Selection to be consistent with the other sections within the by-law.

SECTION 5 - STANDING COMMITTEE MEETINGS

5.1 Standing Committees

The Standing Committees that report directly to Council shall be as follows:

- (a) General Issue Committee
- (b) Board of Health
- (c) Public Works Committee
- (d) Planning Committee
- (e) Emergency & Community Services Committee
- (f) Audit, Finance & Administration Committee

5.2 Appointment of Standing Committee Chairs and Vice Chairs

- (1) Each Standing Committee shall recommend to Council the appointment of a Chair, Vice-Chair and, where applicable, 2nd Vice-Chair for the required term.
 - (a) The term of the Chair, Vice-Chair and 2nd Vice-Chair shall be for a calendar year, or part of a calendar year in an election year.
 - (b) No Member of Council may serve as Chair, Vice-Chair or 2nd Vice-Chair of the same Standing Committee for more than one year in a Council term.
 - (c) The role of the Chair of the General Issues Committee shall be rotated amongst the Deputy Mayors.
- (2) Despite subsection 5.2(1)(a) and (b), the Mayor shall stand as Chair for the term of Council for the Board of Health.
- (3)—The Board of Health shall recommend to Council the appointment of a Vice-Chair for the required term.
- (4)(3) in accordance with 5.2(1). Despite subsection 5.2(1)(a) and (b), the Vice-Chair shall stand as Vice-Chair for the term of Council for the Board of Health. (Approved by Council May 9, 2019)
- (5)(4) Despite subsection 5.2 (1)(a)Each Sub-Committee, Advisory Committee and (b)Task Force shall recommend to Council the appointment of a Chair and Vice Chair. The Chair or Vice-Chair of a Sub-Committee-or an, Advisory Committee or Task Force established by Council may serve for more than one year in a Council term.

5.3 Standing Committee Membership

- (1) General Issues Committee and Board of Health shall be comprised of all Members of Council.
- (2) Public Works Committee shall be comprised of a minimum of 8 Members of Council.

- (3) Planning Committee shall be comprised of a minimum of 7 Members of Council.
- (4) Emergency & Community Services Committee shall be comprised of a minimum of 5 Members of Council.
- (5) Audit, Finance & Administration Committee shall be comprised of a minimum of 7 members of Council.
- (6) Each Member of Council shall sit on a minimum of 2 Standing Committees, in addition to the Board of Health and General Issues Committee.
- (7) Membership on Standing Committees shall be for the duration of the term of Council.
- (8) The Mayor is an ex-officio member of all Standing Committees who, when attending a Standing Committee meeting, shall have full voting privileges and shall be counted for the purpose of the Standing Committee's quorum as set out in subsection 5.4.

5.4 Quorum

- (1) The quorum for all Standing—Committees shall be a half of the membership rounded up to the nearest whole number.
- (2) As soon as there is a quorum after the time set for the meeting, the Chair shall take the chair and call the meeting to order.
- (3) If the Chair does not attend within 15 minutes of the time appointed for a Committee meeting and a quorum of Committee is present, the Vice Chair shall take the chair and call the meeting to order and will preside until the arrival of the Chair.
- (4) If a quorum is not achieved within 30 minutes after the time appointed for <u>aan IN-PERSON / VIRTUAL / HYBRID</u> meeting of Committee, the Clerk shall record the names of the Members of Council present and the meeting shall stand adjourned until the date of the next scheduled Committee meeting or <u>otheranother</u> meeting <u>may be called</u> in accordance with the provisions of this By-law_{\(\bar{\pi}\)}, except as provided for in 5.4(6) and 5.4(7).
- (5) INP If Quorum cannot be maintained during a-an IN-PERSON Committee meeting, the Clerk will advise the Chair that quorum is lost and the decision to continue the meeting will rest with the Chair. During the absence of quorum, no decisions may be approved.
 - ELE If Quorum cannot be maintained during a VIRTUAL / HYBRID Standing
 Committee meeting, due to a loss of members' electronic connection,
 the Clerk will advise the Chair that quorum is lost and the meeting will
 be recessed for up to 15 minutes to allow for the member's
 electronic connection to be restored. If quorum cannot be
 restored within 15 minutes the meeting will be recessed to the next
 available business day. The Clerk will ensure that details regarding

the time of the continuation of the Standing Committee meeting is on the City's website. Where Quorum during a VIRTUAL/HYBRID Standing Committee meeting is lost for any other reason the INP procedure shall apply.

- (6) If quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of Committee, the Chair may decide to proceed informally to hear scheduled presentations. Ratification of any decisions will be postponed to a future meeting.
- (7) If quorum is not achieved within 30 minutes after the time appointed for an IN-PERSON / VIRTUAL / HYBRID meeting of a Committee, the Chair may allow the Committee to discuss items of interest. Ratification of any decisions will be postponed to a future meeting.

5.5 Location of Meetings

- (1) IN-PERSON / VIRTUAL / HYBRID Standing Committee Meetings shall be held in the Council Chambers at City Hall in the City of Hamilton or at such other place within the City of Hamilton as Council may from time to time determine.
- (2) When necessary, Standing Committee Meetings may be held in an adjacent municipality (Halton Region; Niagara Region or Haldimand County), with the details posted on the City's website.

5.6 Standing Committee Meeting Times

- (1) Unless otherwise decided by Council, scheduled meetings of the Standing Committees will be as follows:
 - (a) (i) General Issues Committee shall meet at least once per month on a Wednesday commencing at 9:30 a.m.;
 - (ii) Board of Health shall meet at least once per month commencing at 9:30 a.m.;
 - (iii) Public Works Committee shall meet at least once per month commencing at 1:30 p.m.;
 - (iv) Planning Committee shall meet at least once per month on a Tuesday commencing at 9:30 a.m.;
 - (v) Audit, Finance & Administration Committee shall meet at least once per month on a Thursday commencing at 9:30 a.m.;
 - (vi) Emergency & Community Services Committee shall meet at least once per month on a Thursday commencing at 1:30 p.m.;
 - (b) In the event a meeting date, as shown in subsections 5.56(1)(a)(i) to 5.56(1)(a)(vi) falls on a day which is a public or civic holiday, an alternate date and time will be selected;
 - (c) In accordance with the schedule approved by Council.

Page **25** of **81**

- (2) In addition to regularly scheduled Standing Committee meetings, a special meeting of a Standing Committee may be scheduled when required, at the call of the Chair, at locations and times to permit convenient access for members of the public most affected by such a matter with at least 2 days notice, delivering notice in accordance with subsections 3.45.17(3) and 3.45.17(4).
- (3) Advance notice of Committee meetings for the public's information may be found by accessing the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

5.7 Cancellation of a Committee Meeting

- (1) The Clerk may cancel a Committee meeting in consultation with the Committee Chair or the Committee Vice Chair in the Committee Chair's absence.
- (2) The Clerk shall give each Member of the Committee, or their designated staff, notice of a cancellation of a Committee meeting at least 2 days prior to the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (3) The written notice to be given under subsection 5.67(2) shall state the reason for the cancellation of the meeting.
- (4) The Clerk shall ensure that the meeting cancellation information is posted to the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

Standing & Selection 5.8 Committee Reports

- (1) Each Standing Committee, including—the Selection Committee, and other Committees that report directly to Council shall make recommendations to Council.
- (2) A minimum of 2 days shall pass before the Report of a Standing Committee-and/, the Selection Committee, or Selectionother Committee Report to review. Every effort will be made for Standing Committee and/or Selection Committee Reports to be released at an earlier date.
- (3) Despite subsection 5.78(2), this rule may be suspended by the affirmative vote of two-thirds of the Members of Council present and voting.

5.9 Rules of Procedure

- (1) The rules governing the procedure of the Council at Council meetings shall be observed at Committee meetings, so far as they are applicable.
- (1) No member of Council shall speak more than once, until every member of a Standing Committee, followed by every Member of Council who is not a member of the Standing Committee, has spoken, and then only to provide information or seek clarification from the Chair, other Members of Council or staff.
- (2) A Member of Council may place an item on a future agenda of a Standing Committee meeting by delivering a Notice of Motion as set out in subsection 6.3.14(1) to the Chair of the Standing Committee and to the Committee Legislative Coordinator.

44

- (3) A Member of Council who is not a member of a Standing Committee is entitled to attend such Standing Committee meetings and to participate when a subject matter directly affects their Ward, provided that the Member of Council shall:
 - (a) not be counted for quorum purposes, and;
 - (b) not move any motion or vote on any matter.

5.10 Duties of Standing Committees

The duties of the Standing Committees shall be as set out in Appendices A to F and F1, inclusive.

5.4011 Order of Business

- (1) The general Order of Business for the regularIN-PERSON / VIRTUAL / HYBRID scheduled meetings of Standing—Committees, unless changedamended by the Standing—Committee in the course of the meeting, shall be as follows:
 - (a) Ceremonial Activities (ELE- During a VIRTUAL/HYBRID meeting ceremonial activities shall only be permitted to take place Virtually)
 - (b) Approval of Agenda
 - (c) Declarations of Interest
 - (d) Approval of Minutes of Previous Meeting
 - (e) Communications
 - (f) Delegation Requests
 - (g) Consent Items
 - (h) Public Hearings/Delegations

(i)(h) Staff Presentations

- (i) Public Hearings/ Delegations
- (j) Discussion Items
- (k) Motions
- (I) Notice of Motions

Page 27 of 81

- (m) General Information/Other Business
- (n) Private and Confidential
- (o) Adjournment
- (#2) The Clerk may prepare the Order of Business for the <u>regularscheduled</u> meetings of <u>Standing</u> Committees with slight adjustments to the general Order of Business when deemed necessary.

5.4112 Delegations

- (1) INP Persons who wish to appear as a delegation at an IN-PERSON meeting will appear in-person to address a Standing-Committee
 - ELE Persons who wish to appear as a delegation at a VIRTUAL/
 HYBRID meeting will appear virtually to address a Committee
- (a) Persons wishing to delegate on a matter that :
 - (a) is not listed on a Standing-Committee agenda, shall make a request in writing to the Clerk, and the Clerk will shall list the delegation request on the respective Standing-Committee's upcoming agenda. The requester will be notified of the date of the meeting where their delegation will be heard following Council's ratification of the Standing-Committee's Report.
 - (b) Persons wishing to delegate on a matter that is listed on the agenda for a Standing Committee meetingagenda, shall make a request in writing to be listed as a delegation, such request to be received by to the Clerk no later than 12:00 noon the business day before the meeting. If the Standing Committee is meeting on a Monday, the deadline will be 12:00 noon on the Friday before.
 - (c) All delegates (appearing VIRTUALLY or IN-PERSON) shall complete the "Request to Speak to a Committee of Council" form is-{available on the City's website at https://www.hamilton.ca/council-committee/council-committee-meetings/request-speak-committee-council.
- (2) A request to attend <u>IN-PERSON or VIRTUALLY</u> as a delegation shall include the person's name, reason(s) for the delegation, and if applicable the name, address and telephone number of any person, corporations or organizations which they represent.
- (3) Upon receipt of a request to attend <u>IN-PERSON or VIRTUALLY</u> as a delegation, the Clerk shall list the delegation request on the next appropriate agenda for the relevant <u>Standing</u>-Committee and the decision whether or not to entertain the delegation will be made by the <u>Standing</u> Committee.
- (54) When a person is listed as a delegation on a Standing Committee agenda and such person, without notification to the Clerk, does not attend said meeting, that person must make a new request to be considered as a delegation and must reapply in accordance with subsections 5.4412(1) and 5.4112(2).

- (6) A(5) An IN-PERSON or VIRTUAL delegation, which can be made by two of one or more individualspersons, shall be limited to an oral presentation of not more than five minutes.
- (6) Notwithstanding subsection 5.41(612(5)), an extension of a specific allotment of time to the five-minute speaking restriction may be granted with the approval of a majority of the members of the Committee present.
- (7) Delegations are encouraged to provide a written brief of their presentation to the Clerk, at least 5 days in advance, which will be provided to the members of the Standing-Committee and become part of the public record.
- (8) AAn IN-PERSON or VIRTUAL delegation may provide additional written material regarding the subject matter, for the public record, by providing a copy of such written material to the Clerk at least 5 days in advance of the meeting.
- (9) Except as required by by-law, any <u>IN-PERSON or VIRTUAL</u> delegations who have previously appeared before a particular Committee on a subject matter shall be limited to providing only new information in their second and subsequent appearances at that Committee.
- (10) Notwithstanding subsection 5.41(1012(9), a Committee member may add the delegation on a Committee Agenda.
- 5.12(11)Delegates appearing IN-PERSON or VIRTUALLY will adhere to Section 8 Order and Decorum of this by-law.
- (12) Members of Committees can ask questions of the delegations for clarification purposes only and not for debate.

5.13 Public Hearings at Standing Committees

- (1) Public Hearings shall be held at Standing Committee meetings.
- (2) Advertising or notice of a Public Hearing shall be undertaken as required by applicable legislation, by-law, or according to City policy, to advise interested persons.
- (3) Public Hearings shall commence with:
 - (a) an introduction of the subject matter by the Chair or by staff;
 - (b) the staff presentation, if any and if not waived by the members of the Standing Committee; and
 - (c) INP The signing in of all delegations on a sheet, provided by staff, with their name, address, including postal code, phone number and e-mail address (optional).
 - ELE Presentations by delegations who have registered to attend VIRTUALLY, will be heard.

Page 29 of 81

- (4) Persons who wish to appear as a delegation IN-PERSON or VIRTUALLY to address Standing Committee on a matter that is the subject of a Public Hearing may make a delegation in accordance with section 5.4412.
- (5) INP Notwithstanding subsection <u>5.13(4)</u>, Standing Committee shall hear oral representations from any person in attendance at a Public Hearing, whether or not a request was made in advance.
 - There is no limitation to the length of ELE Notwithstanding subsection 5.13(4),

 Standing Committee shall only hear oral representations from persons who have registered to attend VIRTUALLY.
- (6) IN-PERSON or VIRTUAL delegations made at a Public Hearing, except that the Chair may encourage the delegation shall be limited to conclude where the representations are irrelevant or repetitive, an oral presentation of not more than five minutes.
- (7) Where a delegation wishes to provide written material to <u>become part of the public record to</u> the members of the Standing Committee at a Public Hearing, which will become public record, the delegation shall:
 - (a) (a) comply with subsection 5.11(912(8); or
 - (b) <u>if after the deadline</u>, provide <u>25 copiesa copy</u> of the written material to the <u>Clerk at the <u>IN-PERSON</u> meeting for distribution <u>atafter</u> the meeting, <u>with the exception of</u>; <u>or</u></u>
 - (c) if after the deadline, provide a copy of the written material to the Clerk via email at a VIRTUAL / HYBRID meeting for distribution after the General Issues Committee, which requires 45 copies. meeting.

5.1314 Communication Items

- (1) <u>Communication Items</u> shall be addressed to the <u>Chair</u> and Members of <u>Committee</u> or the City Clerk and <u>shall be</u> presented to the Standing Committee for consideration if the matter is being considered at the Standing Committee meeting.
- (2) <u>Communication Items</u> shall be forwarded to the Office of the City Clerk no later than 12 Noon on the Monday two weeks prior to the Standing Committee meeting for inclusion in the printed Standing Committee Agenda.
- (3) When the date described in subsection 5.4314(2) is a public or civic holiday, the communication item shall be forwarded to the Office of the City Clerk no later than 12 Noon on the last business day prior to the date as set out in subsection 5.4314(2), for inclusion in the printed Standing Committee Agenda.
- (4) Communication Items not received for inclusion in the Standing Committee Agenda package, shall be introduced under Changes to the Agenda, provided that they are received in the Office of the City Clerk no later than 12 Noon on the business day prior to the Standing Committee meeting.

Page 30 of 81

- (5) <u>Communication Items</u> shall not be <u>defamatoryanonymous</u> or contain any <u>defamatory</u> obscene, offensive, anonymous, indecent, improper, rude or vulgar language.
- (6) Communication Items shall not contain the names of individuals (staff or public) with the exception of the author's name. Any names found in the communication will be redacted pursuant to the obligations of Municipal Freedom of Information and Protection of Privacy Act (MFIPPA), this Procedural By-law and other relevant legislation or procedure.
- (7) Communication Items that are relevant to a matter that appears on the Standing Committee Agenda, will be received and referred to the consideration of that matter.
- (8) Communication Items received after the prescribed deadline as set out in subsection 5.4314(4) will be included on the Council Agenda, to be received and referred to the consideration of that matter within the Standing Committee Report.
- (9) Communications respecting labour relation matters, conduct of a member(s) of staff and matters before another body (i.e. Police Services Board) will not be placed on the <u>Standing Committee Agenda</u> and will be forwarded forthwith by the City Clerk to the relevant City Department or body for consideration and appropriate action.
- (10) Communications respecting the conduct of a member(s) of Council or any member of a body subject to review by the Integrity Commissioner will not be placed on a <u>Standing</u> Committee Agenda and the author will be advised of the process for filing a complaint with the Integrity Commissioner.
- (11) Notwithstanding subsections 5.4314(9) and 5.4314(10) a <u>Standing Committee</u> member may add any communication item to a <u>Standing Committee</u> Agenda.

5.1415 Reconsideration of a Matter Decided Within a Standing Committee Meeting

(1) At any Standing Committee Meeting, after a matter has been decided by the Standing Committee, including both successful or failed motions, a Member of the <u>Standing</u> Committee who voted in the majority, may present a Motion to reconsider the matter prior to the adjournment of the meeting. <u>The Chair may ask the Member of the Standing Committee to confirm that they voted with the majority on the issue in question.</u>

The Chair may ask the Member of the Committee to confirm that they voted with the majority on the issue in question.

- (2) A motion to reconsider a decided matter shall require the approval of at least two-thirds of Standing Committee members present.
- (3) No motion to reconsider may, itself, be the subject of a motion to reconsider.

Page 31 of 81

- (4) Debate on a motion to reconsider shall be confined to reasons for or against the reconsideration.
- (5) The following motions cannot be reconsidered:
 - (a) to adjourn;
 - (b) to recess; and
 - (c) to suspend the rules of procedure.

5.1516 Conflict of Interest Declarations (this section is effective March 1, 2019)

- (1) Where a member has disclosed an interest referred to in Section 5 of the Municipal Conflict of Interest Act, 1990a conflict of interest at a meeting, the member shall file a written statement with the Clerk.
- (2) The Clerk shall establish and maintain a registry in which shall be kept,
 - (a) a copy of each statement filed under section 5.4516(1); and
 - (b) a copy of each declaration recorded in the minutes of the meeting.
- (3) The Clerk shall make the Conflict of Interest Registry available for public inspection.

5.17 Special Meetings of a Standing Committee

- (1) In addition to scheduled Standing Committee meetings, the Mayor or the <u>Standing</u> Committee Chair may, at any time, summon a special meeting of a Standing Committee by giving written direction to the Clerk stating the date, time, and purpose of the special meeting.
- (2) The Clerk shall summon a special meeting of a Standing Committee when requested to do so in writing by a majority of Members of the Standing Committee.
- (3) The Clerk shall give each Member of the Standing Committee or their designated staff, notice of a special meeting of a Standing Committee at least 2 days before the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (4) The written notice to be given under subsection 5.4617(3) shall state the nature of the business to be considered at the special meeting of a Standing Committee and no business other than that which is stated in the notice shall be considered at such meeting.

Page **32** of **81**

- (5) Notwithstanding subsection 5.4617(4), other business may be considered at a special meeting with at least two-thirds of the <u>Standing</u> Committee members present and voting in the affirmative.
- (6) The Clerk shall prepare the Order of Business for a special meeting of a Standing Committee with the appropriate selection of agenda categories from subsection 5.4011.

5.18 Change to a Scheduled Committee Meeting

- (1) The Clerk may change the time or location of a Committee meeting in consultation with the Committee Chair or the Committee Vice Chair in the Committee Chair's absence.
- (2) The Clerk shall give each Member of the Committee, or their designated staff, notice of the change to the Committee meeting at least 2 days prior to the time appointed for such meeting by:
 - (a) delivering a written notice personally;
 - (b) delivering such notice at their residence or place of business; or
 - (c) facsimile transmission or electronic mail to such residence or place of business.
- (3) The written notice to be given under subsection 5.4718(2) shall state the reason for the change to the meeting.
- (4) The Clerk shall ensure that the change to the Committee meeting is posted to the Committee and Council Meeting Calendar on the City's website at www.hamilton.ca.

SECTION 6-RULES OF DEBATE, VOTING PROCEDURES AND MOTION PROCEDURES

6.1 Rules of Debate

- (1) No Member of Council shall be deemed to have precedence or seniority over any other Member of Council.
- (2) Before speaking to a question or motion, a Member of Council shall be recognized by the Mayor or the Chair and be placed on the speakers' list.
- (3) When a Member of Council is speaking, no other Member of Council shall pass between that Member of Council and the Mayor or the Chair or interrupt that Member of Council except to raise a point of order or privilege.
- (4) A Member of Council may require the question or motion under discussion to be read at any time during the debate but not so as to interrupt a Member of Council who is speaking.
- (5) A Member of Council may not speak on any matter other than the matter being debated.

- (6) All Members of Council are restricted to a five minute time limit when asking a question and/or making a statement during Council and Committee and Council meetings.
- (7) Notwithstanding subsection 6.1(6), an extension of a specific allotment of time to the five minute speaking restriction may be granted with the approval of a majority of the members of the **Council or** Committee Council present.
- (8) The following matters may be introduced orally without written notice and without leave:
 - (a) a point of order
 - (b) a point of privilege
 - (c) a motion to adjourn
 - (d) a motion to call the question
 - (e) a motion to go into committee of the whole
 - (f) a motion to recess
 - (g) a motion to refer
 - (h) a motion to defer
 - (i) a motion to go into a closed meeting
 - (j) a motion to suspend the rules of procedure
 - (k) a motion to lift

(9) Relinquishing the Chair

The Mayor or the Chair may designate another Member of Council as acting Chair during any part of a meeting of the Council and while presiding, such Member of Council shall have all the powers of the Mayor or the Chair with respect to chairing the meeting.

6.2 Voting Procedures

- (1) Once a motion is moved and seconded it shall be deemed to be in the possession of Council or Committee.
- (2) Where a Member of Council present at a meeting of Council or Committee is prohibited from voting by statute, the Clerk shall record the name of the Member of Council and the reason that they are prohibited from voting.
- (3) INP When an electronic voting system is available at an IN-PERSON meeting every Member of Council or Committee present and participating in a vote on a motion, shall vote electronically excluding those motions referred to in (i) and (ii) below which shall be by show of hands:

Page **34** of **81**

- procedural (majority vote: refer, defer, withdraw, recess, call the question and adjourn); and
- (ii) Consent Items.
- When an electronic voting system is available at a VIRTUAL /
 HYBRID meeting every Member of Council or Committee present and
 participating in a vote on a motion, shall vote electronically.
- (4) When an electronic voting system is used available for a vote, the results will be displayed and recorded in the proceedings.
- (5) When an electronic voting system is not available every Member of Council or Committee present shall vote by a show of hands.
- (6) When an electronic vote is taken and a member advises the Mayor or the Chair of a point of privilege with respect to their vote being omitted or in error immediately following and prior to the taking of another vote, with respect to their vote being omitted or in error the vote will be considered void and a new electronic vote will be retaken immediately.
- (7) When an electronic vote is taken and a member advises the Mayor or the Chair of a point of privilege following the taking of another vote with respect to their vote being omitted or in error following the taking of another vote, the member's request will be considered through a vote to reconsider the matter:
 - (i) If a two-thirds majority vote to reconsider is decided in the affirmative, the electronic vote will be retaken immediately:
 - (ii) If a two-thirds majority vote to reconsider is decided in the negative, the original electronic vote stands.
- (8) When a vote is taken and a tie results, a question is deemed to be lost.
- (9) INP When a Member of Council at an IN-PERSON meeting of Council or Committee does not disclose a pecuniary interest, and does not vote on a Motion, the member will be deemed as having voted in the negative and when an electronic voting system is in use the recusal will be displayed and noted in the proceedings.
 - ELE When a Member of Council at a VIRTUAL / HYBRID meeting of Council or Committee does not disclose a pecuniary interest, and does not vote electronically on a Motion, the Mayor/Chair will call on the member to vote 3 (three) times, if the member doesn't respond the member will be noted as being 'Not Present' during the vote.
- (10) When an electronic voting system is not available and a recorded vote is requested by a Member of Council or Committee immediately before or after a question is put, the Clerk shall record the name and vote of every Member of Council or Committee on the question.

Page **35** of **81**

- (11) When the Mayor or the Chair calls for a vote on a question, each Member of Council or Committee shall occupy their seat and shall remain in their seat until the result of the vote has been declared by the Mayor or the Chair, and during such time no Member of Council or Committee shall walk across the room or speak to any other Member of Council or Committee or make any noise or disturbance.
- (12) After a question is put by the Mayor or the Chair, no Member of Council or Committee shall speak to the question nor shall any other motion be made until after the vote is taken and the result has been declared.
- (13) No vote shall be taken by ballot or by any other method of secret voting and every vote so taken is of no effect, except where permitted by statute.
- (14) Despite subsection 6.2(13), a meeting may be closed to the public during a vote if:
 - (a) the vote relates to an issue permitting the meeting to be closed to the public in accordance with section 9 (Meetings Closed to the Public) of this By-law; and
 - (b) the vote is for a procedural matter or for giving directions or instructions to officers, employees or agents of the City or persons retained by or under a contract with the City.
- (15) The Mayor or the Chair shall, upon request of a Member of Council or Committee, divide a question, if the question under consideration contains distinct propositions, and the vote upon each proposal shall be taken separately.
- (16) If a Member of Council or Committee disagrees with the announcement of the Mayor or the Chair that a question is Carried or Lost on a vote by show of hands, they may immediately after the declaration by the Mayor or the Chair, object to the Mayor's or the Chair's declaration and require that a recorded vote be taken.
- (17) Members of Council who are appointed as representatives of an Advisory Committee, do not have voting rights while attending an Advisory Committee meeting.

6.3 Motion Procedures

(1) Notice of Motion

- (a) Notice of all new motions, except motions listed in subsection 6.1(8) shall be given in writing:
 - (i) at a meeting of Council or Committee, but shall not be debated until the next meeting of Council or Committee; or
 - (ii) delivered to the Clerk at any time before noon on the last business day preceding the date of the meeting at which the notice of motion is to be introduced.

Page **36** of **81**

- (b) Where a Member of Council's or Committee's motion has been called by the Mayor or the Chair at the subsequent meeting and not proceeded with, it shall be dropped from the agenda unless Council or Committee decides otherwise.
- (c) Where Council or Committee has determined not to drop a motion from the agenda, and at the second meeting such motion is called by the Mayor or the Chair and not proceeded with, it shall be deemed to have been withdrawn.

(2) Dispensing with Notice

Where a matter is time sensitive or is an emergency, the Motion may be introduced without notice if Council or Committee, without debate, dispenses with notice on the affirmative vote of two-thirds of the Members of Council or Committee present and voting.

(3) Motion to Withdraw

After a motion is moved and seconded it shall be deemed to be in possession of Council or Committee and may be withdrawn by the mover at any time before decision or amendment.

(4) Motion to Amend

- (a) A motion to amend:
 - (i) shall be presented in writing, duly moved and seconded;
 - (ii) shall be made only to a previous question or to amend an amendment to the question;
 - (iii) shall be relevant to the question to be decided; shall not be received if, in essence it constitutes a rejection of the main question;
 - (iv) shall be put to a vote in the reverse order to that which it is moved.
- (b) If the amendment is not considered a "Friendly Amendment", then the motion to amend:
 - (i) If more than one, will be put to a vote in the reverse order to that in which they were moved;
 - (ii) Will be decided upon or withdrawn before the main motion is put to the vote;
 - (iii) Will not be amended more than once, provided that further amendments may be made to the main motion; and,
 - (iv) Will not be directly contrary to the main motion.

(5) Motion to Defer

A motion to defer a question:

Page **37** of **81**

- (a) is debatable; and
- (b) shall include the date when the question is to be considered.

(6) Motion to Refer

- (a) A motion to refer the question:
 - (i) shall include the name of the individual or body to whom the question is to be referred;
 - (ii) may include instructions respecting the terms upon which the question is to be referred;
- (b) A motion to refer a question is not debatable except where instructions are included, in which case, only the instructions shall be debatable.
- (c) A motion to refer a question may be amended in accordance with subsection 6.3(4).
- (d) When a motion to refer a question to Council is defeated by a Standing

 Committee, the Chair of the Standing Committee shall call the question on the matter.
- (e) A member who has spoken to the motion under debate shall not be permitted to move a motion to refer a question.
- (f) A motion to refer a question is only in order after all first time speakers have been heard.

(7) Motion to Adjourn

A motion to adjourn a Council **or Committee** meeting shall always be in order except:

- (a) when a Member of Council or Committee is speaking or during the taking of a vote;
- (b) immediately following the affirmative resolution of a motion that a vote on the question now be taken;
- (c) when a Member of Council or Committee has already indicated to the Mayor or the Chair, that they desire to speak on the question;
- (d) when decided in the negative, cannot be made again until Council or Committee has conducted further proceedings.

(8) Motion to Call the Question

A motion to call the question shall be in order except when a Member of Council **or Committee** is speaking or before the first-time speakers' list is exhausted.

(9) Motion to Reconsider a Matter Decided by Council

(a) At any Council Meeting, after a matter has been decided by Council, including both successful or failed motions, a Member of Council who voted

Page **38** of **81**

in the majority, may present a Notice of Motion to reconsider the matter. The Chair may ask the Member of Council to confirm that they voted with the majority on the issue in question. Such Notice of Motion shall be referred to the next scheduled meeting of Council and shall be included on the agenda under "Motions".

- (b) No decided matter shall be reconsidered more than once during the term of Council.
- (c) A motion to reconsider a decided matter shall require the approval of at least two-thirds of Council present.
- (d) It is not in order to reconsider a decided matter when the motion has been implemented, resulting in a legally binding commitment that is in place on the date that a motion to reconsider is to be debated.
- (e) No motion to reconsider may, itself, be the subject of a motion to reconsider.
- (f) Debate on a motion to reconsider shall be confined to reasons for or against the reconsideration.
- (g) The following motions cannot be reconsidered:
 - (i) to adjourn;
 - (ii) to recess; and
 - (iii) to suspend the rules of procedure.
- (h) Subsections 6.3(9)(a) to (f) do not apply when:
 - (i) a motion pertains to a decision of a previous Council;
 - (ii) a motion that, while pertaining to a previously decided motion, does not alter the core purpose or intent of the previously decided motion;
 - (iii) a motion presented in respect of a previously decided matter for the purpose of strengthening or clarifying the original decision; or
 - (iv) a motion presented in respect of a previously decided matter for the purpose of moving a matter through a series of necessary steps or phases.

(10) Order of Procedural Motions

The following motions are procedurally in order in the following sequence:

- (a) to adjourn;
- (b) to recess;
- (c) to request information;
- (d) to call the question;
- (e) to defer;
- (f) to refer;
- (g) to amend.

(11) Non-Debatable Motions

Page **39** of **81**

The following motions/actions are not debatable:

- (a) a motion to adjourn;
- (b) a motion to refer (except where instructions are included, in which case, only the instructions shall be debatable);
- (c) a motion to suspend the rules of procedure;
- (d) a point of order; and
- (e) a point of privilege.

(12) Motion to Suspend the Rules of Procedure

The rules governing the procedure of the Council **or Committee** at Council **or Committee** meetings may be suspended and shall require the approval of at least two-thirds of members present.

(13) Motions Beyond Jurisdiction of Council

A motion in respect of a matter which is beyond the jurisdiction of Council shall not be in order.

SECTION 7 - ESTABLISHMENT OF ADVISORY COMMITTEES OR TASK FORCES

- **67.1** Council may from time to time, as needed, establish Advisory Committees or Task Forces in response to specific matters requiring immediate or long term attention.
- **67.2** When Council approves of the formation of the Advisory Committee or Task Force the procedures as outlined in Appendix H shall be followed.
- **67.3** When an Advisory Committee or Task Force has completed its mandate, the Standing Committee to which the Advisory Committee or Task Force reports shall dissolve it by motion.

SECTION 78 - ORDER AND DECORUM

- **78.1** No Person in attendance <u>IN-PERSON or VIRTUALLY</u> at a Council or Committee meeting shall:
 - (a) be disrespectful or disrupt the Meeting in any manner;
 - (b) bring food or beverages, water excepted, into the Council Chamber when a<u>an</u> <u>IN-PERSON</u> meeting is being held;
 - (c) address remarks to anyone but the Chair;
 - (d) interrupt a person who has the floor;
 - (e) applaud participants in debate.
- **78.2** Should a person persist in conducting themselves in a manner contrary to the rules set forth in subsection **78**.1 after having been called to order by the Mayor or a Chair of a Committee, the Mayor or a Chair of a Committee may expel that person and may request the assistance of security and/or Police in doing so.

- **78.3** Notwithstanding subsection **78.2**, in the event a member of the public is conducting themselves in a manner which poses a threat to the personal safety of themselves or others in attendance at the meeting, they shall be removed from the meeting, without warning.
- 78.4 Should a Member of Council persist in conducting themselves in a manner contrary to the rules set forth in subsection 78.1 after having been called to order by the Mayor or the Chair of the Committee, the Mayor or the Chair may order them without challenge to vacate the place the meeting is being held. If the Member of Council apologizes, they may, by vote of a majority of the Council or Committee members, be permitted to remain at the meeting.
- **78.5** No person except members of Council or a Committee, the Clerk, and officials authorized by the Clerk, shall:
 - (a) be allowed on the floor while a<u>an IN-PERSON</u> meeting is being held, namely where members of Council or Committee, the Clerk, and officials authorized by the Clerk are seated and from which they speak;
 - (b) before or during a<u>an IN-PERSON</u> meeting of the Council or a Committee, place on the desks of members or otherwise distribute any material whatsoever unless such person is so acting with the approval of the Clerk.
- **78.6** Signs or placards are permitted at Council and Committee Meetings, provided:
 - (a) they do not contain disrespectful or offensive language;
 - (b) they do not disrupt the Meeting;
 - (c) they do not disrupt an attendee's ability to view the proceedings; and
 - (d) they do not contain solid handles or hard backings.

7

- 8.7 Electronic devices at IN-PERSON/VIRTUAL/HYBRID meetings:
 - (a) Every person shall have all electronic devices, including but not limited to phone, computers, and similar electronic devices, etc. switched to a non-audible function during Council and Committee meetings.
 - (b) The use of electronic devices at IN-PERSON / VIRTUAL / HYBRID Closed Meetings is strictly prohibited.
 - (c) Staff from the Office of the City Clerk are exempted from subsection <u>78</u>.7(b) when using electronic devices for record-keeping purposes.

SECTION 89 - MEETINGS CLOSED TO THE PUBLIC

For the purposes of this section, "Committee" means any advisory or other committee, subcommittee or similar entity of which at least 50% of the members are also members of Council.

- **89.1** No meeting of Council or a Committee or part thereof shall be closed to the public unless the subject matter being considered is:
 - (a) the security of the property of the City or a local board;

Page **41** of **81**

- (b) personal matters about an identifiable individual, including City or a local board employees;
- (c) a proposed or pending acquisition or disposition of land for City or a local board purposes;
- (d) labour relations or employee negotiations;
- (e) litigation or potential litigation, including matters before administrative tribunals, affecting the City or a local board;
- (f) the receiving of advice that is subject to solicitor-client privilege, including communications necessary for that purpose;
- (g) a matter in respect of which Council or a Committee may hold a closed meeting under an Act other than the *Municipal Act*, 2001;
- (h) information explicitly supplied in confidence to the City or a local board by Canada, a province or territory or a Crown agency of any of them;
- (i) a trade secret or scientific, technical, commercial, financial or labour relations information, supplied in confidence to the municipality or local board, which, if disclosed, could reasonably be expected to prejudice significantly the competitive position or interfere significantly with the contractual or other negotiations of a person, group of persons, or organization;
- a trade secret or scientific, technical, commercial or financial information that belongs to the City or a local board and has monetary value or potential monetary value; or
- (k) a position, plan, procedure, criteria or instruction to be applied to any negotiations carried on or to be carried on by or on behalf of the City of a local board.
- **89.2** A meeting of Council shall be closed to the public if the subject matter relates to the consideration of:
 - (a) a request under the Municipal Freedom of Information and Protection of Privacy Act if Council is designated as head of the institution for the purposes of that Act; and
 - (b) an ongoing investigation respecting the City or a municipally-controlled corporation by the Ombudsman appointed under the *Ombudsman Act*.
- **89.3** A meeting of Council or a Committee may be closed to the public if:
 - (a) it is held for the purposes of educating or training members; and
 - (b) at the meeting, no member discusses or otherwise deals with any matter in a way that materially advances the business or decision-making of Council or the Committee.
- **89.4** Before holding a meeting or part of a meeting that is to be closed to the public, a Council or Committee shall:
 - (a) state specifically by resolution the fact of the holding of the closed meeting and the general nature of the matter to be considered at the closed meeting;
 - (b) state specifically that members of the public will be invited to return to hear any further deliberations when Council or Committee reconvenes in Open Session,

Page **42** of **81**

- indicated by the removal of the frosting on the windows and the elimination of the white noise; and
- (c) Council or Committee will, in the event they are in Closed Session for more than 30 minutes, wait up to 5 minutes upon reconvening in Open Session before proceeding with the meeting, to provide members of the public and the media time to return to the meeting room.

SECTION 9 - ELECTRONIC10 - VIRTUAL PARTICIPATION AT COUNCIL AND/OR COMMITTEE MEETINGS

910.1 Electronic Virtual participation is prohibited at Council and/or Committee meetings, except where permitted during an Emergency as set out in subsection 910.2.

910.2 Electronic Virtual Participation During an Emergency

- (1) Electronic Virtual participation, in accordance with the Municipal Act, 2001 and/or any other applicable legislation, is permitted by members of Council at Council and/or Committee meetings during an emergency when attending inperson is not possible. (Refer to Appendix 'J' for UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED as they would apply to Electronic participation by members of Council at ELECTRONIC Council and/or Committee meetings during an emergency when attending in person is not possible)
- (2) Electronic Virtual participation, in accordance with the Municipal Act, 2001 and/or any other applicable legislation, is permitted by Citizen Committee members at the Mayor's Task Force on Economic Recovery, Planning Committee, Hamilton Municipal Heritage Committee and associated Working Groups (Education and Communications Working Group, Policy and Design Working Group, Inventory and Research Working Group), Cross-Melville District Heritage Committee and Heritage Permit Review Committee meetings during an emergency when attending in-person is not possible. (Refer to Appendix 'K' for UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED as they would apply to Electronic participation by Citizen Committee members at ELECTRONIC Council and/or Committee meetings during an emergency when attending in-person is not possible) Committee meetings during an emergency when attending in-person is not possible.
- (3) Electronic Virtual participation, that is consistent with the obligation to hold public meetings in accordance with the Municipal Act, 2001 and/or any other applicable legislation, or at a Standing Committee of Council is permitted by members of public at the Planning Committee, Hamilton Municipal Heritage Committee, Cross-Melville District Heritage Committee, Heritage Permit Review Committee, General Issues Committee, Public Works Committee, Audit, Finance and Administration Committee Board of Health and Emergency and Community Services Committee. (Refer to Revised Appendix 'K' for UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED as they would apply to Electronic participation by members of Council at ELECTRONIC Council and/or Committee meetings during an emergency when attending in-person is not possible)the public at Committee meetings.

Page **43** of **81**

Electronic participation, in accordance with the Municipal Act, 2001 and/or any other applicable legislation, is permitted by Citizen Committee members at the Advisory Committee for Immigrants and Refugees, Committee Against Racism, Hamilton Aboriginal Advisory Committee, Hamilton Status of Women Committee, Lesbian, Gay, Bisexual, Transgender and Queer Advisory Committee, Mundialization Committee, Food Advisory Committee, Hamilton Veterans Committee, Housing and Homelessness Advisory Committee, Seniors Advisory Committee, Arts Advisory Committee, Hamilton Cycling Committee, Keep Hamilton Clean and Green Committee, Agriculture and Rural Affairs Advisory Committee, Hamilton Future Fund, Business Improvement Area Advisory Committee, Cleanliness & Security in the Downtown Core, Advisory Committee for Persons with Disabilities and Waste Management Advisory Committee meetings during an emergency when attending in-person is not possible. (Refer to Appendix 'L' for UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED as they would apply to Electronic participation by Citizen Advisory Committee Members at ELECTRONIC Citizen Advisory meetings during an emergency when attending in-person is not possible)

SECTION 10 SECTION 11 - VIRTUAL MEETINGS DURING AN EMERGENCY

- 11.1 All VIRTUAL/ HYBRID Council and Committee meetings shall be streamed live.
- 11.2 Upon being apprised that a virtual meeting is interrupted and cannot proceed with its live feed, due to loss of connection, sound or video, the meeting will be recessed for up to 15 minutes.
- 11.3 If the live feed of a Virtual meeting cannot be resumed within 15 minutes, the meeting will be considered adjourned. The Council/ Committee meeting will be rescheduled to another day or the next regularly scheduled meeting date.
- 11.4 If the Virtual meeting is interrupted and its live feed cannot be resumed, the names of the Council/Committee members present and any decisions of Council/Committee up to the point in time of the interruption shall be recorded in the minutes of the meeting.

SECTION 12 - ROLE OF CLERK

- 4012.1 The Clerk shall <u>distributemake</u> the agendas of Council and <u>Standing</u> Committee meetings <u>available</u> to members of Council and Senior Leadership Team <u>electronically</u> at least 5 days prior to the scheduled meetings.
- **1012.2** The Clerk shall make the agendas of Council and Standing—Committee meetings_τ available to the media and general public electronically, simultaneously with the distribution set out in subsection 4012.1-
- **10.3** The Clerk shall make the agendas of Council and Standing Committee meetings available on the CityCity's website.
- **10.412.3** The Clerk shall advise the Mayor or Chair when items are required to be added to or removed from an agenda.

Page 44 of 8 ²	Pac	е	44	of	8
---	-----	---	----	----	---

- 10.5 12.4 The Clerk shall submit for confirmation the minutes of the previous Council or Standing Committee meeting. Upon approval of the majority of the members of Council or Standing-Committee, the Mayor or Chair and Clerk shall sign the minutes as confirmed or as amended.
- 10.612.5 The Clerk shall advise the Mayor or Chair, if in their opinion, a matter or portion of a matter being discussed in a meeting that is closed to the public is not procedurally appropriate in accordance with section 239 of the Act and section 89 of this By-law.
- 10.712.6 The Mayor or Chair may call upon the Clerk to provide advice regarding procedure. The Clerk shall provide this advice following which the Mayor or Chair will announce their ruling.
- 10.812.7 The Clerk shall assume the role of the Mayor or Chair as the chair of Council or a Standing-Committee, when a ruling of the Mayor or Chair is challenged by a Member of Council. or Committee. The Clerk shall take a vote of the challenge for Members of Council or Committee in favour or opposed to the ruling. The ruling fails when there is a majority vote in favour of the challenge.

SECTION 4413 - GENERAL

- This By-law shall not be amended or repealed except by a majority vote of all Members of Council.
- 13.2 General guidelines describing the respective roles and responsibilities of Council, the Mayor and the Chair of a Committee are set out in Appendix G to this By-law.
- 13.3 This By-law comes into force on the date on which it is passed, except that subsection 3.17 shall come into force on March 1, 2019.
- 44<u>13</u>.4 The short title of this By-law is the Procedural By-law or the Council Procedural By-law.
- 44<u>13.5</u> By-laws <u>18-270, 19-090, 19-212,19-308, 20-042, 20-055, 20-103, 20-129, 20-145, 20-146, 20-151, 20-184, governing and amending the rules and procedures of Council of the City of Hamilton <u>are</u> repealed.</u>

PASSED this day of	, 2019 2021.	
F. Eisenberger	A. Holland	
Mayor	City Clerk	

Appendix A

GENERAL ISSUES COMMITTEE

COMPOSITION

Page **45** of **81**

The General Issues Committee shall be comprised of all 16 members of Council.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Council Strategic Plan
- Corporate Strategic Plan
- Municipal Benchmarking Network Canada (MBNC)
- Annual Operating and Capital Budgets
- Economic Development matters
- Portfolio Management Strategy Real Estate Acquisitions/Disposals
- Hamilton International Airport matters
- Growth Related Integrated Development Strategy (GRIDS)
- Vision 20/20
- Upper Levels of Government Funding Matters
- Department Work Programs: Planning and Economic Development
- Legal Services all matters having definite litigation matters
- Human Resources labour negotiations
- Departmental Organizational Structure Changes
- · Boards and Agencies
- Hamilton Utilities Corporation
- Climate Change
- any and all other matters which Council chooses to refer to the General Issues Committee for consideration

Specific duties shall include:

- To review and monitor the City's and Council's Strategic Plan
- To review corporate and program objectives and performance measures and make recommendations to Council
- To consider and recommend to Council on matters relating to budgets, budget monitoring, re-assessment and related tax policies
- To consider and recommend to Council on matters relating to Business Development, the Small Business Enterprise Centre, Incentive Loans/Grants programs and approvals, BIA initiatives, the Hamilton Incubator of Technology and Tourism
- To consider and recommend to Council on matters relating to GRIDS
- To receive briefings on legal matters involving the City and give direction to the City Solicitor on litigation matters
- To receive information on labour negotiations and provide direction to the Director of Labour Relations

Appendix A

 To consider and make recommendations to Council on matters regarding Boards and Agencies

Page **46** of **81**

- To Meet as Shareholders and/or Board of Directors of a corporation when required.
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee

Page 47 of 81

Appendix B

BOARD OF HEALTH

COMPOSITION

The Board of Health shall be comprised of all 16 members of Council.

MANDATE

General:

To ensure the City of Hamilton meets or exceeds its obligations as a Board of Health under the *Health Protection and Promotion Act*, and in all matters relating to Public Health in order to promote well-being and create opportunities to enhance the quality of life in our community.

Specific duties shall include:

- To consider and recommend to Council on policy matters and emerging issues related to Public Health
- To provide advice and guidance to the Public Health Services including input to Strategic Planning initiatives to ensure alignment with Council's Mission, Vision, Values and Goals
- To receive delegations from the public and conduct public hearings as required by statute and Council
- To consider and recommend to Council the overall service levels in relation to the delivery of Public Health programs
- To consider and recommend to Council policies governing service delivery of the Public Health Services
- To consider and monitor program implementation and performance through staff reports
- To consider the public health needs of the community and recommend advocacy by Council to address these needs
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this <u>Standing Committee</u>; with all matters having definite litigation going to the <u>General Issues Committee</u>.

Page 48 of 81

Appendix C

PUBLIC WORKS COMMITTEE

COMPOSITION

The Public Works Committee shall be comprised of a minimum of 8 Members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Water & Wastewater (Collection/Distribution, Treatment, and Compliance)
- · Waste Management (Solid Waste Planning, Collection, and Disposal)
- Operations & Maintenance (Roads, Traffic, Forestry & Beautification, Parks & Cemeteries, and Trails)
- Transit (Operations and Transit Fleet, Accessible Transportation Services (ATS))
- Fleet and Facilities (Central Fleet, Corporate Buildings, Energy Office)
- Capital Planning and Implementation (Strategic and Environmental Planning, Asset Management, Design and Construction, Open Space Planning)
- Major road construction projects

Specific duties shall include:

- To consider and recommend to Council, Service Programs and Service Levels for all direct and indirect services provided by the Department (i.e. business plans, management plans, operating plans, and other relevant plans)
- To consider and recommend to Council, Policies, By-laws, and procedures governing service delivery implemented by the Department
- To consider and recommend to Council, Asset Management Plans and Forecasts for all infrastructure managed by the Department
- To consider and monitor program implementation and performance through staff reports and make recommendations to Council respecting program management initiatives (i.e. continuous improvement, budget performance, revenue generation and other relevant issues)
- To consider and recommend to Council, procurement in accordance with corporate policies
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Public Works Committee
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this
 Standing Committee; with all matters having definite litigation going to the
 General Issues Committee.

Page 49 of 81

Appendix D

PLANNING COMMITTEE COMPOSITION

The Planning Committee shall be comprised of a minimum of 9 Members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Long Range Planning, land use management, development planning and engineering, transportation planning and downtown planning and implementation
- Parking Operations and Enforcement, School Crossing Guards
- By-law Enforcement, Municipal Licensing, Lottery Licensing, Building Code issues

Specific duties shall include:

To consider and recommend to Council on:

- the administration and enforcement of the Ontario Building Code, Zoning Bylaws, the Property Standards By-laws, Licensing By-law, Animal Control Bylaw, Sign By-law and other relevant By-laws
- all matters related to the Planning Act, the Ontario Heritage Act, the Municipal Act, 2001, the Niagara Escarpment Act and other applicable legislation regarding planning, development engineering, by-law enforcement and licensing and downtown planning issues
- the City's Official Plan and Zoning By-laws and amendments thereto, pursuant to the *Planning Act*, and to conduct such related public meetings as may be required
- applications for subdivision and condominium approval pursuant to the Planning Act and Condominium Act, as applicable
- matters such as community planning, urban design guidelines, heritage policy and related housing policy and programs in accordance with the directions contained in the Official Plan
- · matters relating to Parking Operations and Enforcement
- To advise and assist Council in any matters arising from the work of the Committee of Adjustment in considering applications for consent and minor variance under the *Planning Act*, including possible City participation at any Local Planning Appeal Tribunal (LPAT) Hearings to consider the appeal of Committee of Adjustment decisions
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Planning Committee
- To receive delegations for the public and conduct public hearings as required by statute and Council, specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this
 Standing Committee; with all matters having definite litigation going to the
 General Issues Committee.

Page 50 of 81

Appendix E

EMERGENCY & COMMUNITY SERVICES COMMITTEE

COMPOSITION

The Emergency & Community Services Committee shall be comprised of a minimum of 5 members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Children's Services
- Ontario Works
- Neighbourhood Development
- Housing Services
- Lodges: Macassa Lodge and Wentworth Lodge
- Recreation
- Fire Services
- Paramedic Services
- Advisory Committees that report to the Emergency & Community Services Committee

Specific duties shall include:

To consider and recommend to Council on:

- service levels for all direct and indirect services provided by the Department
- · policies governing service delivery implemented by the Department
- procurement in accordance with corporate policies
- plans for community facility infrastructure
- the delivery of protective services including fire operations, fire prevention, corporate emergency management, and corporate radio communications
- the delivery of emergency medical (paramedic) services
- To consider and monitor program implementation and performance through staff reports
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Emergency & Community Services Committee
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Page **51** of **81**

Appendix F

AUDIT, FINANCE & ADMINISTRATION COMMITTEE

COMPOSITION

The Audit, Finance & Administration Committee shall be comprised of a minimum of 7 members of Council, plus the Mayor as ex-officio.

MANDATE

General:

To report and make recommendations to Council on matters relating to:

- Internal/external audits;
- Oversight of internal control, financial reporting;
- Human Resources programs and services:
- Programs related to Financial Planning and Policy, Treasury Services, Customer Service, Information Technology, Legislative Services, Records Management, Legal Services, Procurement and Risk Management Access & Equity; and
- To hold hearing on complaints made pursuant to Section 20 of the Development Charges Act, 1997

Specific duties shall include:

To consider and make recommendations to Council on:

- matters of policy respecting human resources, including health and safety, union/management relations, organizational planning and development and compensation administration
- matters of policy involving communications issues
- matters of policy and oversight involving financial management, investment, reserves, debt, procurement and risk management
- matters of policy involving general policies and procedures and administrative By-laws
- all audit matters, including promoting an appropriate environment for the management of public funds and the economy, efficiency and effectiveness of operations and a high level of accountability. Ensure compliance with laws, regulations, policies and support high standard of ethical conduct
- matters of policy and direction related to Access and Equity, including the annual corporate accessibility plan
- To act as liaison to the Access & Equity volunteer advisory committees (with the exception of the Advisory Committee for Persons with Disabilities reporting to the General Issues Committee)
- To consider recommendations of any Sub-Committees and/or Task Forces which report directly to the Audit, Finance and Administration Committee
- To receive delegations from the public and conduct public hearings as required by statute and Council specific to the mandate of this committee on matters under the Standing Committee
- Schedule F1 Office of the City Auditor Charter
- Matters that have the potential of litigation and are under the mandate of this Standing Committee; with all matters having definite litigation going to the General Issues Committee.

Page 52 of 81

Schedule F1 to Appendix F

CITY OF HAMILTON OFFICE OF THE CITY AUDITOR CHARTER

INTRODUCTION

The Office of the City Auditor provides independent, objective audit assurance and consulting services designed to add value and improve the City of Hamilton's operations. The Office of the City Auditor (OCA) brings a systematic, disciplined approach to evaluating and improving the effectiveness of risk management, control and governance processes.

In the capacity of Auditor General, the City Auditor assists City Council in holding itself and its administrators accountable for the quality of stewardship over public funds and for the achievement of value for money in City operations.

SCOPE

The Office of the City Auditor completes the following types of work:

- Compliance Auditing
- Value for Money Auditing
- · Special Investigations (Fraud and Waste, Whistleblower)
- Risk assessments
- Consulting

The first three activities are conducted by the Office in its capacity as Auditor General.

Audit Assurance and Special Investigations

Audit and Assurance work includes the examination and evaluation of the adequacy and effectiveness of the City's governance, risk management process, systems of internal control and the quality of performance in carrying out assigned responsibilities to achieve the organization's goals and objectives. This includes both compliance and value for money audits. The City's processes should function in a manner that ensures:

- · Risks are appropriately identified and managed.
- Significant financial, managerial, and operating information is accurate, reliable and timely.
- Actions are in compliance with policies, standards, procedures, and applicable laws and regulations.
- Resources are acquired economically, used efficiently and are adequately protected.
- · Programs, plans and objectives are achieved.
- Significant legislative and regulatory issues impacting the City are recognized and addressed properly.

Schedule F1 to Appendix F

Page 53 of 81

 Quality and continuous improvement are fostered in the organization's internal control processes.

In addition to audits, special investigations may be undertaken by the OCA pursuant to Council having given that authority under the Whistleblower By-law (19-181) and the City of Hamilton's Fraud Policy and Protocol. These include reviews or investigations of matters of alleged or suspected wrongdoing, per the Whistleblower By-law or fraud, waste or other similar activities reported by employees or residents through the Fraud and Waste Hotline or other mechanisms.

The City Auditor has been appointed by By-law 19-180 (City Auditor By-law) as an Auditor General under Section 223.19 of the *Municipal Act*, 2001, with the responsibilities, including the powers, duties and protections, under Sections 223.19 to 223.23 of the *Municipal Act*, 2001. These responsibilities apply to the extent authorized by these sections of the *Municipal Act*, 2001. The powers and protections provided to the City Auditor (by appointment as an Auditor General) strengthens the position of independence and institutional authority of the City Auditor while providing information to Council for its oversight role.

Consulting and Risk Assessment

In addition to audit engagements, the Office of the City Auditor provides advisory or other consulting services, as planned, or at the request of Council or senior management. These types of services may include:

- · Risk assessments and related workshops;
- Performing research;
- Providing education/training on audit related topics such as risk management, fraud awareness, performance measurement and internal controls;
- Performing Lessons Learned/Opportunities Assessments on project/program implementations that have experienced difficulties or challenges; and
- Providing counsel or advice (e.g. on the adequacy of draft procedures).

AUTHORITY

The Office of the City Auditor is granted full, free and unrestricted access to any and all records, property and personnel relevant to any function under review. Access to personal information is provided for under the *Municipal Freedom of Information and Protection of Privacy Act* (in particular, Subsections 31(c) and 32(d)).

The OCA has the authority to conduct audits and reviews of all City departments, Members of Council, agencies, boards and commissions, as well as other entities the City is related to or has an interest in.

Schedule F1 to Appendix F

Page 54 of 81

All employees will assist the OCA in fulfilling its objectives.

As an Auditor General, the OCA has the responsibilities, including the powers, duties and protections, under Sections 223.19 to 223.23 of the *Municipal Act*, 2001 for:

- · City Departments;
- Members of Council;
- Local boards (not including the Board of Health, the Hamilton Public Library Board, the Police Services Board or other local boards in accordance with the current definition of "local board" under Section 223.1 of the Municipal Act, 2001);
- Municipally-controlled corporations (a corporation that has 50 per cent or more of its
 issued and outstanding shares vested in the City or that has the appointment of a
 majority of its board of directors made or approved by the City, not including a
 corporation established in accordance with Section 203 of the Municipal Act, 2001);
 and
- Grant recipients (as currently defined under Section 223.1 of the Municipal Act, 2001
 as "a person or entity that receives a grant directly or indirectly from the municipality,
 a local board or a municipally-controlled corporation");

These responsibilities under Section 223.19 to 223.23 of the *Municipal Act, 2001* include the powers to access information and to examine persons under Section 33 of the *Public Inquiries Act, 2009*; the duty to preserve secrecy with respect to all matters that come to its knowledge in the course of performing its functions; and the protection of not being a competent or compellable witness in a civil proceeding.

The relevant excerpts from the *Municipal Act* (accessed on July 16, 2019 at https://www.ontario.ca/laws/statute/01m25#BK272) are included below for reference:

Auditor General

223.19 (1) Without limiting sections 9, 10 and 11, those sections authorize the municipality to appoint an Auditor General who reports to council and is responsible for assisting the council in holding itself and its administrators accountable for the quality of stewardship over public funds and for achievement of value for money in municipal operations. 2006, c. 32, Sched. A, s. 98.

Same

(1.1) The Auditor General shall perform his or her responsibilities under this Part in an independent manner. 2009, c. 33, Sched. 21, s. 6 (11).

Exceptions

(2) Despite subsection (1), the responsibilities of the Auditor General shall not include the matters described in clauses 296 (1) (a) and (b) for which the municipal auditor is responsible. 2006, c. 32, Sched. A, s. 98.

Schedule F1 to Appendix F

Powers and duties

(3) Subject to this Part, in carrying out his or her responsibilities, the Auditor General may exercise the powers and shall perform the duties as may be

Page **55** of **81**

assigned to him or her by the municipality in respect of the municipality, its local boards and such municipally-controlled corporations and grant recipients as the municipality may specify. 2006, c. 32, Sched. A, s. 98.

Grant recipients

(4) The authority of the Auditor General to exercise powers and perform duties under this Part in relation to a grant recipient applies only in respect of grants received by the grant recipient directly or indirectly from the municipality, a local board or a municipally-controlled corporation after the date on which this section comes into force. 2006, c. 32, Sched. A, s. 98.

Delegation

(5) The Auditor General may delegate in writing to any person, other than a member of council, any of the Auditor General's powers and duties under this Part. 2006, c. 32, Sched. A, s. 98.

Same

(6) The Auditor General may continue to exercise the delegated powers and duties, despite the delegation. 2006, c. 32, Sched. A, s. 98.

Status

(7) The Auditor General is not required to be a municipal employee. 2006, c. 32, Sched. A, s. 98.

Duty to furnish information

223.20 (1) The municipality, its local boards and the municipally-controlled corporations and grant recipients referred to in subsection 223.19 (3) shall give the Auditor General such information regarding their powers, duties, activities, organization, financial transactions and methods of business as the Auditor General believes to be necessary to perform his or her duties under this Part. 2006, c. 32, Sched. A, s. 98.

Access to records

(2) The Auditor General is entitled to have free access to all books, accounts, financial records, electronic data processing records, reports, files and all other papers, things or property belonging to or used by the municipality, the local board, the municipally-controlled corporation or the grant recipient, as the case may be, that the Auditor General believes to be necessary to perform his or her duties under this Part. 2006, c. 32, Sched. A, s. 98.

No waiver of privilege

(3) A disclosure to the Auditor General under subsection (1) or (2) does not constitute a waiver of solicitor-client privilege, litigation privilege or settlement privilege. 2006, c. 32, Sched. A, s. 98.

Schedule F1 to Appendix F

Powers re examination

223.21 (1) The Auditor General may examine any person on oath on any matter pertinent to an audit or examination under this Part. 2006, c. 32, Sched. A, s. 98.

Page **56** of **81**

Application of Public Inquiries Act, 2009

(2) Section 33 of the Public Inquiries Act, 2009 applies to an examination by the Auditor General. 2009, c. 33, Sched. 6, s. 72 (3).

Duty of confidentiality

223.22 (1) The Auditor General and every person acting under the instructions of the Auditor General shall preserve secrecy with respect to all matters that come to his or her knowledge in the course of his or her duties under this Part. 2006, c. 32, Sched. A, s. 98.

Same

- (2) Subject to subsection (3), the persons required to preserve secrecy under subsection (1) shall not communicate information to another person in respect of any matter described in subsection (1) except as may be required,
- (a) in connection with the administration of this Part, including reports made by the Auditor General, or with any proceedings under this Part; or
- (b) under the Criminal Code (Canada). 2006, c. 32, Sched. A, s. 98.

Same

(3) A person required to preserve secrecy under subsection (1) shall not disclose any information or document disclosed to the Auditor General under section 223.20 that is subject to solicitor-client privilege, litigation privilege or settlement privilege unless the person has the consent of each holder of the privilege. 2006, c. 32, Sched. A, s. 98.

Section prevails

(4) This section prevails over the Municipal Freedom of Information and Protection of Privacy Act. 2006, c. 32, Sched. A, s. 98.

Testimony

223.23 Neither the Auditor General nor any person acting under the instructions of the Auditor General is a competent or compellable witness in a civil proceeding in connection with anything done under this Part. 2006, c. 32, Sched. A, s. 98.

INDEPENDENCE

Independence is an essential component to maintaining public trust and preserving objectivity and integrity associated with the audit function.

Schedule F1 to Appendix F

To provide for the independence of Office of the City Auditor, its personnel report to the City Auditor, who reports administratively to the City Manager and functionally to the Audit, Finance and Administration Committee of Council. Audit and other reports are sent directly to the Audit, Finance and Administration Committee for discussion and approval and then

Page **57** of **81**

to Council. These reporting relationships help ensure independence, promote comprehensive audit objectivity and coverage and assure adequate consideration of audit recommendations.

All OCA activities shall remain free of influence by any element in the organization, including matters of audit selection, scope, procedures, frequency, timing or report content to permit maintenance of an independent and objective attitude necessary in rendering reports. The City Auditor has the authority to revise and extend the scope of any audit or investigation in the course of their examination.

The OCA shall have no direct operational responsibility or authority over any of the activities it reviews. Accordingly, it shall not develop nor install systems or procedures, prepare records or engage in any other activity which would normally be audited.

RESPONSIBILITIES

The City Auditor and the staff of the OCA have the responsibility to:

- Review operations within the City at appropriate intervals to determine whether
 planning, organizing, directing and controlling are in accordance with management
 instructions, policies and procedures and in a manner that is consistent with both
 City objectives and high standards of administrative practice.
- Determine the adequacy and effectiveness of the systems of internal accounting, financial and operating controls.
- Review the reliability, utility and integrity of financial information and the means used to identify, measure, classify and report such information.
- Review the established systems to ensure compliance with those policies, plans, procedures, laws and regulations which would have a significant impact on operations and reports and determine whether the organization is in compliance.
- Review the means of safeguarding assets and, as appropriate, verify the existence
 of such assets.
- Carry out value for money (VFM) / performance audits to determine the efficiency and effectiveness of services and evaluate attainment of corporate objectives and value to residents.
- Report to those members of management who should be informed or who should take corrective action, the results of audit examinations, the audit opinions formed, and the recommendations made.

Schedule F1 to Appendix F

 Evaluate any plans or actions taken to correct reported conditions and provide timely follow-up to ensure satisfactory disposition of audit findings in the manner and timeframe committed to by management in the original audit report. If the corrective

Page **58** of **81**

action is considered unsatisfactory, hold further discussions to achieve acceptable disposition.

- Develop a flexible multi-year work plan, including any risks or control concerns identified by management or other audits as well as appropriate special tasks or projects requested by management.
- Undertake assessments, investigations, or refer issues to other appropriate parties as a result of disclosures under a Whistleblower By-law or the Fraud and Waste Hotline.
- Maintain a professional audit staff with sufficient knowledge, skills and experience.

AUDIT PLANNING

Each term of Council, the City Auditor shall prepare a multi-year work plan, setting out the proposed work for the term.

For the projects, the following sources are considered:

- Prioritization of the audit universe using a risk-based methodology;
- Requests from Members of Council, senior management and staff;
- Any audits planned for the past term but delayed or not completed;
- Any conditions or concerns discovered or communicated throughout past years; and
- Input from members of the public, via the Fraud and Waste Hotline and resident complaints or suggestions.
- Areas likely to provide significant payback in terms of increased revenues, reduced costs, operational efficiencies and quality of services.

The multi-year work plan will be presented to the Audit, Finance and Administration Committee for approval. Any changes to the work plan requested by Council or individual Members of Council will require a majority of at least two-thirds the total members of Council present and not prohibited by statute from voting for the City Auditor to consider.

Schedule F1 to Appendix F

REPORTING

A written report is prepared and issued by the City Auditor following the conclusion of each audit assurance project. The report will include management's responses to the report findings and recommendations. Management's response will include a statement of

Page **59** of **81**

general agreement or disagreement with the stated findings and recommendations as well as a timeframe for anticipated completion of action to be taken and an explanation for any recommendation not addressed.

The OCA is responsible for appropriate follow up on audit findings and recommendations. All significant findings will remain open until the City Auditor has determined management has appropriately taken action to resolve the finding.

By-law 19-181 (Whistleblower By-law), Section 19 — Responsibility of the City Auditor requires a quarterly report for, in the aggregate, on the number, nature and outcome of disclosures of serious wrongdoing made under this By-law. Quarterly reports known as "Whistleblower, Fraud and Waste Information Updates" are submitted to the Audit, Finance and Administration Committee and capture at a high level the special investigations activity undertaken by the OCA.

Consulting and risk assessment project reports are issued at the discretion of the City Auditor. Considerations for reports include project size and type, topic, risk profile and organizational impact.

All reports are generally presented to the Audit, Finance and Administration Committee. Reports may be presented to the General Issues Committee, if appropriate.

PROFESSIONAL STANDARDS

Under the City Auditor's authority granted in this Office of the City Auditor Charter, all OCA activity will be guided by the following professional standards and/or legislative requirements:

- Municipal Act, Sections 223.19-223.23 (Auditor General), as applicable.
- City of Hamilton Whistleblower By-law (19-181).
- International Standards for the Professional Practice of Internal Auditing issued by the Institute of Internal Auditors, for research and general consulting work.
- Certified Fraud Examiners Code of Professional Standards issued by the Association
 of Certified Fraud Examiners for Fraud, Waste and Whistleblower assessments and
 investigations.

When working as an Auditor General performing audit assurance engagements, work activities will be guided by professional standards and/or legislative requirements, as applicable.

Audit Services professional staff are also bound by the standards and ethics of their respective professional organizations, which include the Chartered Professional Accountants of Canada (CPA), Institute of Internal Auditors (IIA) and Association of Certified Fraud Examiners (ACFE).

(Approved by Council – September 11, 2019 – Report AUD19005)

Page **60** of **81**

Appendix G

ROLES OF COUNCIL, MAYOR AND COMMITTEE CHAIR

PREAMBLE

The general requirements for holding office for elected officials for the City of Hamilton are contained in the *Municipal Act*, 2001.

GENERAL ROLE/GUIDELINES (COUNCIL)

- (a) Within the authority of enabling legislation, to develop policies for the purpose of guiding the administration of municipal government in the City of Hamilton.
- (b) Subject to legislative restrictions, develop regulations to be adopted in By-laws and resolutions for the overall benefit of the citizens of the community.
- (c) Appoint statutory officers and senior officials to ensure that a functional management system is in place and which will administer the City within the adopted policies of Council.
- (d) To collectively oversee the administrative functions as carried out by appointed officials within the delegated authority and the policies adopted by Council.
- (e) To monitor the administrative process and ensure that the will and direction of the City is placed into effect.
- (f) To be prepared to attend regularly scheduled meetings and special meetings of Council and Committees as required and to participate in the debate for the purpose of developing and adopting policies and directions for the City of Hamilton.
- (g) Absences from the meetings of Council for three successive months results in a member's seat being declared vacant unless authorized by resolution of Council.
- (h) To act as liaison between the citizens they represent and the City, to ensure that the intention of established policies and regulations are applied in a manner that is conducive to the interests of the citizens as a whole.
- (i) To oversee the financial affairs and delivery of City services through the adoption of policies and budget control guidelines and to ensure that appropriate audit procedures and monitoring programmes are in effect.

Page **61** of **81**

Appendix G

ROLE OF THE MAYOR

(including the Deputy Mayor while fulfilling the duties of Mayor)

The Mayor is responsible to act as the Head of Council, as detailed in the *Municipal Act, 2001* providing leadership to other Members of Council.

- (a) To act as the Council's corporate representative when dealing with other government agencies and the private sector consistent with the vision and direction expressed by the Council of the day.
- (b) The Mayor and the City Manager must work in close liaison as the pivotal link between the policy-making body of Council and the administrative organization of the City.
- (c) The role of Mayor is considered as statutory and policy-related, to act as the Head of Council and to co-ordinate political representation on behalf of the City when required at meetings, receptions, functions, and community activities, and to direct administrative functions to the attention of the City Manager.

ROLE OF COMMITTEE CHAIR

- (a) To ensure that the general functions noted in the roles of Council and in the policies and procedures established by the City are maintained.
- (b) To ensure that the rules of procedure with respect to conduct are followed.
- (c) The role of Chair, in co-ordinating the meetings of a Committee, will be considered statutory and policy-related, with administrative functions to be delegated to appointed staff and officials through the appropriate General Manager or Medical Officer of Health, following appropriate reporting procedures of the City.
- (d) It is recognized that the role of Chair, as outlined above, does not provide any administrative authority over staff, and that any administrative requests should be brought to the attention of the City Manager, appropriate General Manager or Medical Officer of Health or appropriate delegated staff.

DUTIES OF COMMITTEE CHAIR

- (a) Open the meetings of the Committee by taking the chair and calling the members to order.
- (b) Announce the business before the Committee and the order in which it is to be acted upon.
- (c) Receive and submit, in the proper manner, all motions presented by the members of the Committee.
- (d) Put to a vote all motions, which are moved, or necessarily arise in the course of the proceedings, and to announce the result.

Page **62** of **81**

Appendix G

- (e) Enforce the rules of procedure and rule upon all procedural matters.
- (f) Guide the members when engaged in debate in accordance with the rules of procedure.
- (g) Enforce on all occasions the observance of order and decorum among the members and the attending public.
- (h) Order any member persisting in a breach of the rules of procedure to vacate the meeting room.
- (i) Permit questions to be asked through the Chair of any staff in order to provide information to assist any debate when the Chair deems it proper.
- (j) Rule on any points of order or points of privilege without debate or comment.
- (k) Rule whether a motion or proposed amendment is in order.
- (I) Determine which member has the right to speak.
- (m) Ascertain that all members who wish to speak on a motion have spoken and that the members are ready to vote, and shall then put the vote.
- (n) May call a member to order.
- (o) Where it is not possible to maintain order, the Chair may, without any motion being put, adjourn the meeting to a time to be named by the Chair.
- (p) Adjourn the meeting when the business is concluded.
- (q) The Chair may state relevant facts and the Chair's position on any matter before the Committee without leaving the chair, but it shall not be permissible for the Chair to move a motion or debate a question without first leaving the chair.
- (r) If the Chair desires to leave the chair to move a motion, or to take part in the debate, the Chair shall call on the Vice-Chair, or in the absence of the Vice-Chair, on another member to preside until the Chair resumes the chair.
- (s) The Committee Chair shall vote on any questions before the Committee and in the event of an equality of votes (tie vote) the Committee Chair will not have an extra casting vote and the question being voted upon is deemed lost.

Page **63** of **81**

Appendix H

CRITERIA FOR THE CREATION OF ADVISORY COMMITTEES OR TASK FORCES

- (a) That all requests for the formation of an Advisory Committee or Task Force be presented to a Standing Committee for consideration.
- (b) That all requests, upon approved motion by the Standing Committee, be referred to the applicable General Manager for a report back on the following;
 - (i) inventory of previous and existing activities related to the issue
 - (ii) inclusion of a sunset clause, which outlines the time needed to complete the mandate
 - (iii) reporting structure, membership composition, mandate, objectives, legislative requirements, work plan and timelines for the Special Purpose Body
 - (iv) responsibilities of the members
 - (v) membership expertise requirements for the Special Purpose Body
 - (vi) proposed budget allocation requirement and source of funding
 - (vii) staffing requirements
 - (viii) other necessary resources
- (c) The Clerk may invite citizens to serve as members and such invitations shall include the Advisory Committee or Task Force's mandate including any membership expertise requirements and the time needed to complete the mandate;
- (d) Applications received for membership shall be forwarded by the Clerk to the Selection Committee:
- (e) The Selection Committee may consider balanced geographical representation in selecting the members, where appropriate;
- (f) The rules of procedure shall be observed so far as they are applicable;
- (g) A Code of Conduct, setting out general standards for acceptable conduct by Members of Advisory Committees and Task Forces in the performance of their duties, is set out in Appendix I to this By-law.
- (h) All minutes and reports shall be submitted to the Standing Committee to which the Advisory Committee or Task Force reports; and
- (i) Staff may act as resource persons in a non-voting capacity.

Page **64** of **81**

Appendix I

HAMILTON ADVISORY COMMITTEE/TASK FORCE CODE OF CONDUCT

Council has adopted this Code of Conduct for the guidance of Appointees to Advisory Committees and Task Forces providing recommendations to Standing Committees and to assist Appointees in performing their duties in a manner which will promote the public's confidence in these Advisory Committees and Task Forces operating with integrity, transparency and courtesy.

It is recognized that the Code of Conduct cannot anticipate all possible fact situations in which Appointees may be called upon to exercise judgement as to the appropriate standard of conduct. When this occurs, Appointees are to ensure that their decisions maintain the Advisory Committee or Task Force's integrity, transparency and courtesy.

This Code of Conduct does not apply to Members of Council who are subject to the Council Code of Conduct.

Failure to comply with this Code of Conduct may result in the Advisory Committee or Task Force:

- (1) requesting an apology from the Appointee; and/or
- (2) removing the Appointee from the Advisory Committee or Task Force for a portion or all of their term.

1. GOOD CONDUCT

Appointees shall act with honesty and integrity including:

- acting in a manner that contributes to the public's confidence in the Advisory Committee or Task Force; and,
- not engaging in conduct that may, or may appear to, constitute an abuse of their position as an Appointees.

2. MEETINGS

Appointees shall maintain proper control over meetings demonstrating respect for everyone who is involved in a proceeding.

Appointees are expected to attend all meetings of the Advisory Committee or Task Force. If an Appointee misses more than three meetings during their term, the Chair, after hearing and considering any explanation provided by the Appointee, may remove the Appointee from the Advisory Committee or Task Force for the remainder of their term.

3. COLLEGIALITY

Appointees shall respect and co-operate with other Appointees and the Advisory Committee or Task Force staff.

4. GIFTS OR BENEFITS

Appointees shall not accept a gift or benefit that may appear as being offered because they are Appointees.

5. CONFIDENTIAL INFORMATION

Appointees shall not disclose to any member of the public any confidential information acquired by virtue of their position.

Page **65** of **81**

Appendix I

6. COMMUNICATION

Appointees should accurately communicate a recommendation or direction.

Appointees may provide their own personal opinion on a matter, provided that it is made clear to the party they are speaking to that the comments are their own and are not being made on behalf of the Committee providing it does not relate to In Camera discussions.

Appointees may refer the media or others making inquiries to the Chair, or, in the absence of the Chair, to the Vice-Chair.

Page **66** of **81**

Appendix J

CONDUCTING AN ELECTRONIC MEETING

The following UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED applies to Electronic participation by members of Council at ELECTRONIC Council and/or Committee meetings during an emergency when attending in-person is not possible:

- 2.1 The rules of procedure shall be observed in all ELECTRONIC proceedings of Council and shall be the rules for the order and dispatch of business in Council and unless specifically provided, with necessary modifications, apply to all Committees.
- 2.2 All matters relating to the ELECTRONIC proceedings of Council and Committees for which rules have not been provided for in this By-law and its Appendices shall be decided, as far as is reasonably practicable, with reference to the parliamentary rules as contained in the most recent edition of Bourinot's Rules of Order, Geoffrey H. Stanford, Fourth Edition.

3.2 Regular Council Meeting Times

- (1) Unless otherwise decided by Council, ELECTRONIC Council meetings shall be held:
 - (a) in January:
 - on the fourth Wednesday of the month, commencing at 9:30 a.m. unless such day is a public or civic holiday, in which case Council shall set an alternate day and time:
 - (b) in February, April, May, June, September, October and November:

 every second and fourth Wednesday of each month, commencing at 9:30

 a.m., unless such day is a public or civic holiday, in which case Council shall
 - (c) in March:
 - on the fourth Wednesday of the month, commencing at 9:30 a.m. unless such day is a public or civic holiday, in which case Council shall set an alternate day and time;
 - (d) in July and August:

set an alternate day and time;

- on a Friday of each month (one meeting per month), commencing at 9:30 a.m., unless such a day is a public or civic holiday, in which case Council shall set an alternate day and time;
- (e) in December:
 - on the second Wednesday of the month, commencing a 9:30 a.m., unless such a day is a public or civic holiday, in which case Council shall set an alternative day and time; or
- (f) in accordance with the schedule approved by Council.

Page 67 of 81

3.7 Quorum

(5) If Quorum cannot be maintained during an ELECTRONIC meeting, DUE TO A LOSS OF ELECTRONIC CONNECTION, the Clerk will advise the Mayor or Deputy Mayor that quorum is lost and the MEETING WILL BE RECESSED UNTIL THE ELECTRONIC CONNECTION IS RESUMED, WHICH COULD RESULT IN A DELAY TO THE NEXT AVAILABLE BUSINESS DAY. THE CLERK WILL ENSURE THAT THE PROPER MESSAGING RESPECTING THE CONTINUATION OF A COUNCIL MEETING TO THE FOLLOWING DAY OR NEXT AVAILABLE DAY IS ON THE CITY'S WEBSITE.

3.11 Order of Business

	Order of Business for ELECTRONIC meetings of Council, unless changed by cil in the course of the meeting, shall be as follows:
(a)	Approval of Agenda
(b)	Declarations of Interest
(c)	Approval of Minutes of Previous Meeting
(d)	Communications
(e)	Committee of the Whole
	(i) Consent Items
	(ii) Public Hearings / Written Delegations
	(iii) Staff Presentations

- (f) Motions
- (g) Notice of Motions
- (h) Statement by Members (non-debatable)
- (j) Private and Confidential
- (k) By-laws and Confirming By-law

(iv) Discussion Items

(I) Adjournment

3.13 Voting Procedures (electronic voting is effective January 2019)

- (3) (i) When an electronic voting system is available AT AN

 ELECTRONIC MEETING every Member of Council or Committee
 present and participating in a vote on a motion, excluding those
 motions referred to in (i) and (ii) below, shall vote electronically:
 - (a) procedural (majority vote: refer, defer, withdraw, recess, call the question and adjourn); and
 - (b) Consent Items.

Page **68** of **81**

(ii) IF A MEMBER OF COUNCIL DOES NOT VOTE ELECTRONICALLY WHEN A VOTE IS CALLED BY THE MAYOR/CHAIR AT AN ELECTRONIC MEETING, THE MAYOR/CHAIR WILL CALL ON THE MEMBER TO VOTE 3 (THREE) TIMES, IF THE MEMBER DOESN'T RESPOND THE MEMBER WILL BE NOTED AS BEING 'NOT PRESENT' DURING THE VOTE.

4.1 Council may, by resolution:

- (a) move into Committee of the Whole AT AN ELECTRONIC MEETING to consider, including debate, CONSENT ITEMS, PUBLIC HEARINGS/WRITTEN DELEGATIONS, STAFF PRESENTATIONS AND DISCUSSION ITEMS on the Council meeting agenda; and
- (b) move out of Committee of the Whole to report to Council.
- **4.2** During the consideration of CONSENT ITEMS AT AN ELECTRONIC MEETING, a member may lift an item from the LIST with the approval of a majority of the members present, for the matter:
 - (a) to be voted upon SEPARATELY; or
 - (b) to be debated and/or amended.

5.5 Quorum

(5) If Quorum cannot be maintained during an ELECTRONIC meeting, DUE TO A LOSS OF ELECTRONIC CONNECTION, the Clerk will advise the Chair that quorum is lost and the MEETING WILL BE RECESSED UNTIL THE ELECTRONIC CONNECTION IS RESUMED, WHICH COULD RESULT IN A DELAY TO THE NEXT AVAILABLE BUSINESS DAY. THE CLERK WILL ENSURE THAT THE PROPER MESSAGING RESPECTING THE CONTINUATION OF A COMMITTEE MEETING TO THE FOLLOWING OR NEXT AVAILABLE DAY IS ON THE CITY'S WEBSITE.

5.10 Order of Business

- (i) The general Order of Business for the ELECTRONIC meetings of Standing Committees, unless changed by the Standing Committee in the course of the meeting, shall be as follows:
 - (a) Approval of Agenda
 - (b) Declarations of Interest
 - (c) Approval of Minutes of Previous Meeting
 - (d) Communications
 - (e) Consent Items
 - (f) Public Hearings/Written Delegations
 - (a) Staff Presentations
 - (h) Discussion Items
 - (i) Motions

Page **69** of **81**

- (i) Notice of Motions
- (k) General Information/Other Business
- (I) Private and Confidential
- (m)Adjournment

5.11 WRITTEN Delegations

- (1) Persons who wish to delegate to COUNCIL OR COMMITTEE on a matter that:
 - (b) is listed OR NOT LISTED on the agenda for a COMMITTEE OF THE WHOLE OR COMMITTEE, shall SUBMIT THEIR DELEGATION in writing TO THE CLERK, the Clerk will list the DELEGATION on the COMMITTEE OF THE WHOLE'S OR THE COMMITTEE'S upcoming agenda. Such WRITTEN DELEGATIONS MUST be received by the Clerk no later than 12:00 noon the business day before the meeting.
- 10.1 The Clerk shall MAKE the agendas of ELECTRONIC Council and Committee meetings AVAILABLE to members of Council and Senior Leadership Team at least 5 days prior to the scheduled meetings ON THE CITY'S WEBSITE.
- **10.2** The Clerk shall make the agendas of ELECTRONIC Council and Committee meetings, available to the media and general public, simultaneously with the distribution set out in subsection 10.1.

Page **70** of **81**

Appendix J

<u>HEARING PROCEDURES FOR</u> DEVELOPMENT CHARGE COMPLAINTS – SECTION 20

COMMITTEE COMPOSITION

The Audit, Finance & Administration Committee has delegated authority to hold hearings for Section 20 Development Charges complaints. The Audit, Finance & Administration (AF&A) shall be comprised of a minimum of 7 members of Council, plus the Mayor as exofficio.

SECTION 20 COMPLAINT

As Per the *Development Charges Act, 1997*, a person may submit a formal complain to Council respecting one of three issues:

- 20 (1) A person required to pay a development charge, or the person's agent, may complain to the council of the municipality imposing the development charge that,
 - (a) the amount of the development charge was incorrectly determined;
 - (b) whether a credit is available to be used against the development charge, or the amount of the credit or the service with respect to which the credit was given, was incorrectly determined; or
 - (c) there was an error in the application of the development charge bylaw. 1997, c. 27, s. 20 (1).

The formal complaint must be submitted within 90 days of the charge being payable (Development Charges Act, 1997) and must be in accordance with the grounds permitted under the Development Charges Act:

(2) A complaint may not be made under subsection (1) later than 90 days after the day the development charge, or any part of it, is payable. 1997, c. 27, s. 20 (2).

The formal complaint must be in writing, stating the complainant's name, address where notice can be given, and the reason for the complaint.

DEVELOPMENT CHARGES COMPLAINT HEARING - SECTION 20 COMPLAINT

A hearing (or tribunal) must be held within 60 days of receiving the complaint (Section 22(2)). A Hearing Notice will be mailed to the complainant at least 14 days before the hearing.

The Complaint Hearing is governed by the procedures set out in the Council Procedure By-law and the Statutory Powers Procedure Act ("SPPA").

Page **71** of **81**

Appendix J

The Audit, Finance & Administration Committee, in conducting the hearing of a Development Charges Complaint, is required to only consider the grounds permitted under subsection 20(1) of the *Development Charges Act*.

The grounds set out in the *Development Charges Act* are very specific and quite focused. They do not include a request to be exempt from a Development Charge otherwise applicable pursuant to a Development Charge By-law nor do they include a request to create a new category of development not found in a Development Charge By-law. A Development Charge Complaint cannot be used to amend the Development Charge By-law to alter the Development Charge rate otherwise validly applicable or to add a credit or exemption not already within the Development Charge By-law.

1. Party Representatives

There are two parties to the hearing of a Development Charge Complaint, the City and the Complainant. The City's Finance Division is responsible for administering the Development Charge By-law and a solicitor from the Legal Division will represent the City at the hearing. The Complainant may represent themselves or have legal representation as well.

In addition, the Committee will have a solicitor from the Legal Division to provide legal advice to the Committee during a hearing and deliberations. This practice is permissible provided that the Solicitor does not take part in making findings of fact or in making the ultimate decision on the matter.

2. Live Stream & Electronic Voting

The Development Charge Hearing is not live-streamed when meeting IN-PERSON, unless specifically requested by someone in attendance and then agreed upon by both parties.

Electronic votes are not taken during a Development Charge Hearing.

Subject to Section 4 of this Appendix J, the Development Charge Hearing is live-streamed when meeting VIRTUALLY, in order to have all proceedings open to the public and the parties.

3. Conflict of Interest

If any Committee member feels that they should not be hearing a particular matter, the member may inform the Committee that they will not participate, giving a general or specific reason why not, e.g. one of the witnesses is a relation.

Page **72** of **81**

Appendix J

4. Open and Closed Proceedings / Deliberations

All proceedings are to be open to the public and the parties unless one of the exceptions under the SPPA or the *Municipal Act, 2001* applies. Section 9(1) of the SPPA provides that a hearing may be closed to the public if:

- (a) a matter involving public security may be disclosed; or
- (b) intimate financial or personal matters or other matters may be disclosed at the hearing of such a nature, having regard to the circumstances, that the desirability of avoiding disclosure thereof in the interest of any person affected or in the public interest outweighs the desirability of adhering to the principle that hearings be open to the public.

If the Committee believes that such matters could be disclosed, it should ask each of the parties if this is the case. If the Committee is convinced that the harm outweighs the desirability of the hearing being open to the public, the hearing may be closed to the public.

When a hearing is closed to the public under the SPPA, only the parties and their representatives remain in attendance.

The Committee may also rely on the authority under section 239 of the *Municipal Act*, 2001 in closing proceedings to the public or the public and the parties - for example, if it wishes to receive advice subject to solicitor client privilege.

The Committee may, but is not required to, retire to deliberate in the absence of the public and the parties. Deliberations occur when the Committee considers the evidence and submissions in arriving at a decision. The decision itself is announced in the presence of the public and the parties. The authority for retiring to deliberate is found at common law and is referred to as the "confidentiality principle" or the "rule on deliberative secrecy".

5. Adjournments

Adjournments may be requested by either party at the start of or during a hearing. The Committee may grant or refuse an adjournment request in light of a number of considerations including: the legitimate inability of the complainant or a witness to attend or, within reason, the counsel of their choice; or, the necessity for time to prepare before a hearing or to respond to new and unexpected issues or allegations arising in the course of a hearing.

6. Agreed Upon Statements of Fact and Joint Submissions

The parties may submit an agreed upon statement of facts which means that they will inform the Committee that some or all of the facts relevant to the hearing are not in dispute. Agreed upon facts need not be proven and should be accepted by the Committee.

Page **73** of **81**

Appendix J

The parties may go beyond an agreed upon statement of facts to make a joint submission, asking for a final decision that is acceptable to both. The Committee must give serious consideration to a joint submission and must not reject it without good cause. While the Committee may reject all or part of a joint submission, if this is being considered, both parties must be given the opportunity to make representations before the final decision is made.

7. Witnesses

The parties may call witnesses during the Hearing. (Section 10.1, SPPA)

At the request of a party or on its own initiative, the Committee may require the attendance of a witness to give evidence by issuing a summons.

A witness who is summoned to testify before the Committee cannot refuse to answer a proper and relevant question and Section 13 of the *Canadian Charter of Rights and Freedoms* and Section 14 of the SPPA protect witnesses when providing answers.

The Committee should be hearing only the relevant evidence of the parties' witnesses.

Witnesses should be sworn or affirmed by the Chair of the Committee prior to commencing their testimony.

Lawyers or representatives acting as an advocate for a complainant cannot be a witness.

8. Evidence

Administrative tribunals are given much more latitude than courts with respect to the evidence which they may receive and consider in arriving at a decision. Accordingly, the Committee may receive hearsay evidence and unsworn evidence. (Section 15, SPPA)

The fundamental test with respect to the admissibility of evidence is that it must be relevant to the issues which are involved in the hearing. Relevance for a Development Charge hearing is determined by reference to the permitted grounds set out in subsection 20(1) of the Development Charges Act and by the grounds set out in the complaint, but only if those grounds are permitted grounds.

When the Committee is confronted with an objection to the admissibility of a relevant piece of evidence, the evidence should generally be admitted unless it is clearly irrelevant. The Committee should consider the objection with respect to the weight it gives to that particular evidence when arriving at its decision. The general principle is that indirect evidence (hearsay) should be given less weight than direct evidence such as a witness' own observations, unless there is a valid reason to conclude that the direct evidence is not credible.

Page **74** of **81**

Appendix J

The Committee may make a finding of credibility in considering the testimony of a witness — giving little or no weight to testimony it does not find credible. An administrative tribunal may find it very difficult to indicate in a decision that a witness was not credible. It is advisable for the administrative tribunal to fully and clearly explain itself, for example, by stating X's testimony was not relied upon because they admitted to a direct financial interest in the outcome of the hearing and because their answers were influenced by this.

Although unaffirmed or unsworn evidence is admissible, testimony to the Committee should be given under affirmation or oath. Each witness should be affirmed or sworn immediately before giving their testimony. A witness should be asked whether they prefer to be sworn or affirmed.

An unrepresented complainant is acting in two capacities, both as their own representative and as a witness. When they are acting as a witness – for example, telling the Committee what did or did not happen – they should be under affirmation.

The parties may examine their own witnesses and cross-examine other witnesses. (Section 10.1, SPPA) The Committee may also question witnesses. Generally, this should be done after the parties have finished questioning the witness.

The Chair or Vice Chair presiding should allow each party to ask any further questions of the witness they may have arising from questions posed by a member of the Committee.

The onus is on the complainant to satisfy the Committee that the Development Charge amount was incorrectly determined, that a credit should have been issued against the Development Charge, that a credit was incorrectly applied, or there was an error in the application of the Development Charge by-law.

9. Note-taking

Notes taken by Committee members during the Hearing should be kept for a reasonable period of time, enough time for whatever appeal right there may be to be exercised. A problem arises when protection of privacy and freedom of information legislation applies. In the case of the *Municipal Freedom of Information and Protection of Privacy Act*, there is no exception made for the notes of members sitting on a municipal tribunal such as a Development Charge Complaint Hearing. As a consequence, if the City, in this case Clerks, has custody or control of the notes, these notes may be subject to disclosure under MFIPPA. In addition, the notes arguably would be subject to the Records Retention By-law and could only be destroyed in accordance with that By-law.

Consequently, to ensure that adjudicative independence can be maintained, it is recommended that if a Committee member chooses to take notes, they:

- (1) Do so for the purposes of helping:
 - (a) to remember and understand what occurred during a hearing; and (b) to make a decision in respect of a hearing;
- (2) Keep their notes confidential, not allowing any other person to see, read or use the notes for any purpose;
- (3) maintain responsibility for the care and safe-keeping of their notes;
- (4) store their notes at their office or home; and
- (5) destroy their notes after some reasonable period of time such as one year.

Page **75** of **81**

Appendix J

10. Decisions

In the context of a Development Charge Complaint hearing, the Committee makes two types of decisions:

- (1) procedural (decisions such as adjourning the hearing to another date; and,(2) final decisions concerning whether:
 - (a) the amount of the development charge was incorrectly determined;
 - (b) whether a credit is available to be used against the development charge, or the amount of the credit or the service with respect to which the credit was given, was incorrectly determined; or
 - (c) there was an error in the application of the development charge by-law.

The procedural decision or final decision of the majority of the members of the Committee is the procedural decision or final decision of the Hearing, allowing for a vote with dissent. However, it is recommended that the Committee operate on a consensual basis in respect of Development Charge Complaint decisions.

The Committee is not required to give reasons for a procedural decision that is made with the consent of the parties. The Committee must give oral and/or written reasons for a procedural decision that does not have the consent of both parties.

For a final decision, the Committee must announce it orally in the presence of the parties and the public. Additionally, it must make its decision in writing to the complainant and/or their legal counsel or agent within a reasonable time subsequent to the completion of the hearing.

Subsection 17(1) of the SPPA requires a written decision to include reasons for the decision only if a party to the complaint requests reasons. If reasons are requested the decision should summarize the facts and arguments presented by the parties, the findings of fact made by the Committee. The decision must explain the relationship between the evidence and its conclusions, including why evidence was rejected or given little credibility.

Committee members should refrain from expressing opinions on the merits or strength of a case until after all of the evidence and submissions have been heard and they are giving their decision.

It is important that the Committee provides clear instructions to the Legislative Coordinator about the decisions it makes, however, it is acceptable for the Legislative Coordinator to prepare a draft decision for approval by the Committee.

11. Appeal

The Development Charges Act requires notice of the decision of the Hearing to be mailed within 20 days after the day the decision is made.

Page 76 of 81

Appendix K

CONDUCTING AN ELECTRONIC MEETING - PUBLIC

MAYOR'S TASK FORCE ON ECONOMIC RECOVERY, PLANNING COMMITTEE, HAMILTON MUNICIPAL HERITAGE COMMITTEE AND ASSOCIATED WORKING GROUPS (EDUCATION AND COMMUNICATIONS WORKING GROUP, POLICY AND DESIGN WORKING GROUP, INVENTORY AND RESEARCH WORKING GROUP), CROSS-MELVILLE DISTRICT HERITAGE COMMITTEE, HERITAGE PERMIT REVIEW COMMITTEE, GENERAL ISSUES COMMITTEE, PUBLIC WORKS COMMITTEE, AUDIT, FINANCE AND ADMINISTRATION COMMITTEE BOARD OF HEALTH AND EMERGENCY AND COMMUNITY SERVICES COMMITTEE

The following UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED applies to Electronic participation by citizen Committee members and members of the public at ELECTRONIC Committee meetings duringappeal is an emergency when attending inpersonappeal de nove which means LPAT is not possible with respect to the Mayor's Task Force on Economic Recovery, Planning Committee, Hamilton Municipal Heritage Committee and associated Working Groups (Education and Communications Working Group, Policy and Design Working Group, Inventory and Research Working Group), Cross-Melville District Heritage Committee, Heritage Permit Review Committee, General Issues Committee, Public Works Committee, required to consider the Audit, Finance and Administration Committee Board of Health and Emergency and Community Services Committee only:

- 21 The rules of procedure shall be observed in all ELECTRONIC proceedings of Council and shall be the rules for the order and dispatch of business in Council and unless specifically provided, with necessary modifications, apply to all Committees.
- All matters relating to the ELECTRONIC proceedings of Council and Committees for which rules have not been provided for in this By-law and its Appendices shall be decided, as far as is reasonably practicable, with reference to the parliamentary rules as contained in the most recent edition of Bourinot's Rules of Order, Geoffrey H. Stanford, Fourth Edition.

5.4 Quorum

(5) If Quorum cannot be maintained during an ELECTRONIC meeting, DUE TO A LOSS OF ELECTRONIC CONNECTION, the Clerk will advise the Chair that quorum is lost and the MEETING WILL BE RECESSED UNTIL THE ELECTRONIC CONNECTION IS RESUMED, WHICH COULD RESULT IN A DELAY TO THE NEXT AVAILABLE BUSINESS DAY. THE CLERK WILL ENSURE THAT THE PROPER MESSAGING RESPECTING THE

Page **77** of **81**

CONTINUATION OF A COMMITTEE MEETING TO THE FOLLOWING OR NEXT AVAILABLE DAY IS ON THE CITY'S WEBSITE.

5.10 Order of Business

- The general Order of Business for the ELECTRONIC meetings of Standing Committees, unless changed by the Standing Committee in the course of the meeting, shall be as follows:
 - (a) Approval of Agenda
 - (b) Declarations of Interest
 - (c) Approval of Minutes of Previous Meeting
 - (d) Communications
 - (e) Consent Items
 - (f) Public Hearings/Written Delegations/Virtual Delegations
 - (g) Staff Presentations
 - (h) Discussion Items
 - (i) Motions
 - (i) Notice of Motions
 - (k) General Information/Other Business
 - (I) Private and Confidential
 - (m) Adjournment

5.11 WRITTEN Delegations/VIRTUAL Delegations

(1) Persons who wish to APPEAR VIRTUALLY as a delegation at a Standing Committee of Council or at a public meeting held in accordance with applicable legislation, in a manner that is consistent with the obligation to hold public meetings at electronic Committee meetings during an emergency when attending in person is not possible with respect to the Planning Committee, Hamilton Municipal Heritage Committee, Cross-Melville District Heritage Committee, Heritage Permit Review Committee, General Issues Committee, Public Works Committee, Audit, Finance and Administration Committee Board of Health and Emergency and Community Services Committee, shall make a request in writing to be listed as a delegation, such request to be received by the Clerk no later than 12:00 noon the business day before the meeting. If the Committee is meeting on a Monday, the deadline will be 12:00 noon on the Friday.

The "Request to Speak to a Committee of Council VIRTUALLY" form is available on the City's website at

Page **78** of **81**

- (2) Persons who wish to delegate to Committee on matters that don't require the holding of a public meeting in accordance with applicable legislation, may submit their delegation in WRITING to the Clerk, the Clerk will list the delegation on the Committee's upcoming agenda. Such WRITTEN delegations must be received by the Clerk no later than 12:00 noon the business day before the meeting.
- (3) A request to APPEAR VIRTUALLY as a delegation at a Standing Committee of Council or at a public meeting held in accordance with applicable legislation, shall include the person's name, reason(s) for the delegation, and if applicable the name, address and telephone number of any person, corporations or organizations which they represent. Requests to appear virtually will only be accepted for items listed on the agenda.
- (4) Upon receipt of a request to APPEAR VIRTUALLY as a delegation or at a Standing Committee of Council or at a public meeting held in accordance with applicable legislation, the Clerk shall list the delegation request on the next appropriate agenda for the relevant Committee.
- (5) A VIRTUAL delegation of one or more persons, being made at a public meeting held in accordance with applicable legislation, will be limited to five (5) minutes.
- (6) Notwithstanding subsection 5.11(5), an extension of a specific allotment of time to the five-minute speaking restriction may be granted with the approval of a majority of the members of the Committee present.
- (7) Delegates appearing Virtually will adhere to Section 7 Order and Decorum of the Procedural By-law 18-270 as amended.
- (8) A VIRTUAL delegation may provide additional written material<u>reasons in its</u> <u>decision regarding</u> the subject matter, for the public record, by providing a copy of such written material to the Clork at least 5 days in advance of the meeting.
- (9) Except as required by by law, any VIRTUAL Delegations who have previously appeared before a particular Committee on a subject matter shall be limited to providing only new information in their second and subsequent appearances at that Committee.

5.12 Public Hearings

- (1) Public Hearings shall be held at Standing Committee meetings.
- (2) Advertising or notice of a Public Hearing shall be undertaken as required by applicable legislation, by-law, or according to City policy, to advise interested persons.

Appendix A to Report FCS21004/LS21001

- (3) Public Hearings shall follow the following order:
 - (a) an introduction of the subject matter by the Chair or by staff;
 - (b) the staff presentation, if any and if not waived by the members of the Standing Committee; and
 - (c) presentations by delegations who have registered to APPEAR VIRTUALLY, will be heard.
- (4) Persons who wish to appear as a delegation VIRTUALLY to address
 Committee on a matter that is the subject of a Public Hearing may make a delegation in accordance with section 5.11 of Appendix K.
- (5) VIRTUAL delegations made at a Public Hearing will be limited to 5 minutes.
- (6) Where a delegation wishes to provide written material to the members of the Standing Committee at a Public Hearing, which will become public record, the delegation shall:
 - (a) comply with subsection 5.11(8) of Appendix K.
- 10.1 The Clerk shall MAKE the agendas of ELECTRONIC Council and Committee meetings AVAILABLE to members of Council and Senior Leadership Team at least 5 days prior to the scheduled meetings ON THE CITY'S WEBSITE.
- 10.2 The Clerk shall make the agendas of ELECTRONIC Council and Committee meetings, available to the media and general public, simultaneously with the distribution set out in subsection 10.1.

Procedural By-law

Page 2 of 81

Appendix L

CONDUCTING AN ELECTRONIC MEETING – Citizen Advisory Committees

ADVISORY COMMITTEE FOR IMMIGRANTS AND REFUGEES, COMMITTEE
AGAINST RACISM, HAMILTON ABORIGINAL ADVISORY COMMITTEE, HAMILTON
STATUS OF WOMEN COMMITTEE, LESBIAN, GAY, BISEXUAL, TRANSGENDER
AND QUEER ADVISORY COMMITTEE, MUNDIALIZATION COMMITTEE, FOOD
ADVISORY COMMITTEE, HAMILTON VETERANS COMMITTEE, HOUSING AND
HOMELESSNESS ADVISORY COMMITTEE, SENIORS ADVISORY COMMITTEE,
ARTS ADVISORY COMMITTEE, HAMILTON CYCLING COMMITTEE, KEEP
HAMILTON CLEAN AND GREEN COMMITTEE, AGRICULTURE AND RURAL
AFFAIRS ADVISORY COMMITTEE, HAMILTON FUTURE FUND, BUSINESS
IMPROVEMENT AREA ADVISORY COMMITTEE, CLEANLINESS & SECURITY IN THE
DOWNTOWN CORE, ADVISORY COMMITTEE FOR PERSONS WITH DISABILITIES
AND WASTE MANAGEMENT ADVISORY COMMITTEE

The following UPDATED SECTIONS OF BY-LAW 18-270, AS AMENDED applies to Electronic participation by Citizen Advisory Committee Members at ELECTRONIC Citizen Advisory Committee meetings during an emergency when attending in-person is not possible with respect to the Advisory Committee for Immigrants and Refugees, Committee Against Racism, Hamilton Aboriginal Advisory Committee, Hamilton Status of Women Committee, Lesbian, Gay, Bisexual, Transgender and Queer Advisory Committee, Mundialization Committee, Food Advisory Committee, Hamilton Veterans Committee, Housing and Homelessness Advisory Committee, Seniors Advisory Committee, Arts Advisory Committee, Hamilton Cycling Committee, Keep Hamilton Clean and Green Committee, Agriculture and Rural Affairs Advisory Committee, Hamilton Future Fund, Business Improvement Area Advisory Committee, Cleanliness & Security in the Downtown Core, Advisory Committee for Persons with Disabilities and Waste Management Advisory Committee only:

- 2.1 The rules of procedure shall be observed in all ELECTRONIC proceedings of Council and shall be the rules for the order and dispatch of business in Council and unless specifically provided, with necessary modifications, apply to all Committees.
- 2.2 All matters relating to the ELECTRONIC proceedings of Council and Committees for which rules have not been provided for in this By-law and its Appendices shall be decided, as far as is reasonably practicable, with reference to the parliamentary rules as contained in the most recent edition of Bourinot's Rules of Order, Geoffrey H. Stanford, Fourth Edition.

5.6 Quorum

(5) if quorum cannot be met or maintained during an ELECTRONIC meeting, for whatever reason, the Clerk/Staff Liaison will advise the Chair that quorum has not been reached or has been lost and the meeting will be recessed for up to 15 minutes, or until quorum is reached. If quorum cannot be reached within 15

Procedural By-law

Page 3 of 81

minutes, the meeting will be considered adjourned. The Committee will meet at the next regularly scheduled meeting date.

(6) If Quorum cannot be obtained, the Committee Secretary shall record the names of the Committee members present in the minutes of the meeting.

5.10 Order of Business

- (iii) The general Order of Business for the ELECTRONIC meetings of Citizen Advisory Committees, unless changed by the Committee in the course of the meeting and shall be as follows:
 - (a) Changes to the Agenda
 - (b) Declarations of Interest
 - (c) Approval of Minutes of Previous Meeting
 - (d) Communications
 - (e) Delegations
 - (f) Consent Items
 - (g) Presentations
 - (h) Discussion Items
 - (i) Notices of Motion
 - (i) Motions
 - (k) Other Business
 - (I) Adjournment

5.11 WRITTEN Delegations

- (1) Persons who wish to delegate to ELECTRONIC Citizen Advisory Committee shall submit their delegation in WRITING to the Clerk/Staff Liaison, the Clerk/Staff Liaison will list the delegation on the Committee's upcoming agenda. Such WRITTEN delegations must be received by the Clerk/Staff Liaison no later than 12:00 noon the business day before the meeting. If the Committee is meeting on a Monday, the deadline will be 12:00 noon on the Friday.
- 10.1 The Clerk/Staff Liaison shall MAKE the agendas of ELECTRONIC Citizen Advisory Committee meetings available to members of the committee 3 days prior to the scheduled meetings ON THE CITY'S WEBSITE.
- **10.2** The Clerk/Staff Liaison shall make the agendas of ELECTRONIC Citizen Advisory Committee meetings, available to the media and general public, simultaneously with the distribution set out in subsection 10.1 an appeal.